

Trade Liberalization and Child Labor in Selected SAARC and ASEAN Countries

Bilal Tariq

Doctor of Philosophy 2016

Trade Liberalization and Child Labor in Selected SAARC and ASEAN Countries

Bilal Tariq

A thesis submitted

In fulfillment of the requirements for the degree of

Doctor of Philosophy (Economics)

Faculty of Economics and Business

UNIVERSITI MALAYSIA SARAWAK

2016

DECLARATION

Name: Bilal Tariq

Matric Number: 14-0100-58

I hereby declare that this research is the result of my own investigation, except where otherwise stated. Other sources are acknowledged by giving explicit references and a bibliography is appended. The thesis has not been accepted for any degree and is not concurrently submitted in candidature of any other degree.

Signature -----

Date -----

ACKNOWLEDGEMENTS

I am thankful to almighty Allah who granted me courage, determination and potential to accomplish my research work. Numerous salutations are on the Holy Prophet Muhammad (PBUH), the city of knowledge, who has guided all the humanity to seek knowledge from cradle to grave.

My parents are worthy of special thanks for their extraordinary efforts they made in bringing me up. They encountered all the troubles in this regard with the smiling face and went on praying for me. I pray almighty Allah to mercy them and sustain their shadow over my head forever. A special thanks to my sisters who offered invaluable support and humor over the years. Finally, thanks must go to my elder brother (Dr. Usman Tariq) for the excellent example he sets (as well as having shown me the importance of getting a good PhD supervisor), and my younger brother (Nouman Tariq) for daring to follow his dreams.

I would like to express my deepest gratitude to my advisor, Associate Professor Dr. Rossazana Bt Ab Rahim, for her tremendous supervision, caring, tolerance and providing me with an excellent environment for doing research. She provided me the courage and constant guidance to carry out this great task. She gave me her scholarly advice for improving the stuff contained in this work. It is due to her kind supervision that this study took the present form. Once again, I express my heartfelt gratitude for her and pray for her health. I am very much thankful to Dr Dayang Affizah Awang Marikan (Universiti Malaysia Sarawak), Dr. Raushan Koizhaiganova (Korea University) and Associate Professor Dr Evan Lau Poh Hock (Universiti Malaysia Sarawak) for their valuable suggestions and encouragement. I am also indebted to The Centre for Academic Information Services (CAIS) staff that helped me by providing the required books, research journals and official publications. A very special thanks to CALM staff members (Sh Norizan Bt Wan Zain and Zurina binti Naihi) for providing help and assistance whenever I required.

Moreover, very special thanks go to the Faculty of Economics and Business for their guidance and assistance throughout my thesis. I also take this opportunity to thank the wonderful and extremely helpful administrative staff of the university who made it possible for me to come, stay and complete my work at the university.

ABSTRACT

Child labor is a stubborn problem for South Asian Association for Regional Cooperation (SAARC) and Association of South East Asian Nations (ASEAN), which seems to be declining. In recent years, the trade openness has raised the response towards the problem of child labor in these adjacent regions, because of an additional influx of trade of differentiated products. Trade of differentiated products is a significant source of trade between developing countries and needed to be examined with regard to child labor. Existing studies tended to investigate the nexus between the trade and child labor in traditional trade theory setting and the results show mixed evidence across countries. Hence, it would be interesting for this research to embark upon the issue of new trade theory by demonstrating the trade effects of differentiated products on the child labor. Apart from that, it would be appealing to investigate the impact of free trade on the child labor in the major SAARC countries, namely, Bangladesh, India, Pakistan, Nepal, and Sri Lanka and selected ASEAN countries, namely, Cambodia, Indonesia, Philippine, and Thailand, which are known to have a relatively high number of child labor. Empirically, this research assesses the theoretical treatment of child labor and trade related effects by employing selection, scale and technique variables. The panel data method is employed to justify the spatial and temporal dimensions of the research. The estimation procedure of this dissertation consists of three main steps. In the first step, an exposure assessment of the simple model by capturing selection, scale and technique effects obtained for the child labor in closed economy case. In the second step, a variable of trade combined with selection, scale and technique effects variables to find the effect of a change in trade on child labor. Subsequently, the third step presents trade interactions to check the trade-induced effects across the selected SAARC and ASEAN countries. The findings in this research confirm the importance of selection, scale and technique effects in the estimation of the full impact of international trade on child labor. Finally, this research proposes that the trade induced child labor effects can be worthwhile to address the underlying economies that gives a rise to offend child labor practices. The findings of this study recommends better policies to overcome the problem of child labor in the context of open economy and closed economy.

Keywords: Trade liberalization, Child labor, Differentiated products

Liberalisasi Perdagangan dan Buruh Kanak-kanak di Negara-negara SAARC dan ASEAN Terpilih

ABSTRAK

Buruh kanak-kanak ialah masalah yang sukar ditangani oleh Pertubuhan Kerjasama Serantau Asia Selatan (SAARC) dan pertubuhan Negara-negara Asia Tenggara (ASEAN), dengan kadar yang semakin menurun. Sejak kebelakangan ini, perdagangan bebas telah menimbulkan masalah buruh kanak-kanak di antara kawasan-kawasan yang berdekatan ekoran daripada peningkatan akitiviti perdagangan bagi produk-produk yang berbeza. Perdagangan produk yang berbeza adalah sumber penting dalam perdagangan antara negara-negara membangun; maka usaha pemeriksaan teliti terhadap penggunaan buruh kanak-kanak perlu dilakukan. Kajian terdahulu mengkaji hubungan antara buruh kanak-kanak dan perdagangan berdasarkan teori perdagangan tradisional telah menunjukkan hasil kajian yang bercampuran buruh kanak-kanak. Oleh itu, kajian ini menyiasat isu yang berkait rapat dengan teori perdagangan baru, dengan menunjukkan kesan perdagangan produk-produk berbeza terhadap buruh kanak-kanak. Selain itu, kajian ini juga akan memberikan maklumat yang penting berkaitan dengan kesan perdagangan bebas terhadap buruh kanak-kanak di negara-negara SAARC utama iaitu, Bangladesh, India, Pakistan, Nepal dan Sri Lanka dan negara-negara ASEAN yang terpilih iaitu, Kemboja, Indonesia, Filipina dan Thailand, yang terkenal dengan mempunyai bilangan buruh kanak-kanak yang tinggi. Secara empirikal, kajian ini menyiasat buruh kanak-kanak dan kesan teori perdagangan baru dengan mengambilkira kesan pembolehubah berkaitan pemilihan buruh, skala dan teknik. Kaedah data panel telah diguna untuk menjustifikasikan dimensi ruang dan masa kajian. Prosedur kajian ini terdiri daripada tiga langkah utama. Pada langkah pertama, hubungan kesan pemilihan, skala dan teknik oleh buruh kanak-kanak dalam model ekonomi tertutup. Pada langkah kedua, pembolehubah perdagangan digabungkan dengan kesan pilihan, skala dan teknik untuk mencari kesan perubahan dalam perdagangan buruh kanak-kanak. Selepas itu, langkah ketiga membentangkan interaksi perdagangan bagi memeriksa kesan perdagangan yang sedang berlaku di seluruh negara SAARC dan ASEAN yang terpilih. Hasil kajian ini membuktikan kepentingan kesan pemilihan, skala dan teknik dalam pengkajian kesan perdangangan antarabangsa terhadap buruh kanak-kanak. Implikasi ini kajian ini menyarankan polisi yang lebih baik bagi menangani masalah buruh kanak-kanak dalam konteks ekonomi terbuka dan ekonomi tertutup.

Kata kunci: Liberalisasi perdagangan, Buruh kanak-kanak, Produk-produk berbeza

TABLE OF CONTENTS

DECLARATION	i
ACKNOWLEDGEMENTS	ii
ABSTRACT	iv
ABSTRAK	vi
TABLE OF CONTENTS	viii
LIST OF TABLES	xiii
LIST OF FIGURES	xiv
LIST OF ABBREVIATIONS / ACRONYMS	XV
CHAPTER 1: INTRODUCTION	1
1.1 Chapter Overview	1
1.2 Introduction	1
1.3 Research Background	4
1.4 Definitions of Child Labor	8
1.5 Child Labor in SAARC and ASEAN	9
1.6 Trade and Child Labor	
1.7 New Trade Theory and Child Labor	
1.8 Problem Statement	
1.9 Objective of the Study	

1.10 Significance of the Study	
1.11 Organization of the Study	
CHAPTER 2: AN OVERVIEW OF SAARC AND ASEAN	
2.1 Chapter Overview	
2.2 Introduction	
2.3 Background of the ASEAN Countries	
2.3.1 Overview of ASEAN Economies	
2.3.2 Child Labor in ASEAN	
2.4 Background of the SAARC Countries	
2.4.1 Overview of SAARC Economies	
2.4.2 Child Labor in SAARC	
2.5 Major Developments in Child Labor Reduction	
CHAPTER 3: LITERATURE REVIEW	54
3.1 Chapter Overview	54
3.2 Introduction	
3.3 Theoretical Studies	
3.4 Empirical Studies	
3.5 A Concluding Remark	

CH	APTER 4: THEORETICAL APPROACH AND METHODOLOGY	83
4.1	Chapter Overview	83
4.2	Theoretical Framework	83
4.3	Conceptual Framework	89
	4.3.1 Hypothesis 1	91
	4.3.2 Hypothesis 2	92
	4.3.3 Hypothesis 3	93
	4.3.4 Hypothesis 4	94
4.4	The Model	95
4.5	Monopolistic Competition and the Child Labor	101
4.6	Trade Induced Child Labor Effects	102
	4.6.1 Effect of Trade Openness on Child Labor	102
	4.6.2 Selection Effect	104
	4.6.3 Scale Effect	105
	4.6.4 Technique Effect	106
4.7	The Estimation Model of Trade of Differentiated Products	108
	4.7.1 Explanation of Variables	109
	4.7.2 Child Labor Effects in the Closed Economy	115
	4.7.3 Child Labor Effects in the Open Economy	115

4.7.4 Trade-Induced Child Labor Effects	117
4.8 Measurement and Data Sources	119
4.9 Method	
4.10 Unobservable Variables	
CHAPTER 5: EMPIRICAL RESULTS	127
5.1 Chapter Overview	
5.2 Descriptive Statistics	127
5.3 Empirical Results	
5.3.1 Estimation Results for Model A and Model B	
5.3.2 Discussion of the Results	
5.3.3 Estimation Results for Interacted and Non-interacted Models	136
5.3.4 Discussion of the Results for Interacted and Non-interacted Mod	dels138
5.4 Alternate Empirical Specifications	
5.4.1 Alternative Specification	
5.4.2 Robustness of Results	147
CHAPTER 6: CONCLUSION AND RECOMMENDATIONS	157
6.1 Chapter Overview	157
6.2 Summary of the Results	157
6.3 Recommendations of the Study	

6.4	Limitations of the Study	166
6.5	Directions for Future Research	167
6.6	Policy Implications	168
	6.6.1 Policy Implication of Scale and Technique Effects	168
	6.6.2 Policy Implications of Selection Effect	169
	6.6.3 Policy Implication of Trade Openness	170
RE	FERENCES	171
AP	PENDIX A	200
AP	PENDIX B	209

LIST OF TABLES

Table 1: Establishment and Systems of Governance (ASEAN)	
Table 2: Socio Economic Indicators of ASEAN Countries	
Table 3: Socio Economic Indicators of SAARC Countries	40
Table 4: Establishment and Systems of Governance (SAARC)	42
Table 5: Key Achievements of the ILO to Combat Child Labor	49
Table 6: ILO Convention Ratification Status of Member Countries	
Table 7: Empirical Evidence of Child Labor	76
Table 8: Explanation of Important Symbols	
Table 9: Trade Induced Child Labor Effects	
Table 10: Variable Explanation and Measurement	
Table 11: Variable Description	
Table 12: Panel Data Models with a Small Number of Countries.	
Table 13: Descriptive Statistics of Variables	
Table 14: Estimation Results for Model A and Model B	
Table 15: Estimation Results for Interacted and Non-interacted Models	
Table 16: Alternative Specification	145
Table 17: Robustness of the Results	

LIST OF FIGURES

Figure 1: Percentage of Children Aged 5 to 14 Years Engaged in Child Labor	10
Figure 2: Percentage of Child Labor in SAARC and ASEAN	11
Figure 3: Economic Benefits of Ending Child Labor via Trade Liberalization	14
Figure 4: Worldwide Child Labor	15
Figure 5: The Mechanism of Child Labor	71
Figure 6: Theoretical Framework	84
Figure 7: Trade Induced Child Labor Effects	90
Figure 8: Child Labor and Firms Density	110
Figure 9: Child Labor and Scale of Production	114
Figure 10: Child Labor and Trade Openness	116

LIST OF ABBREVIATIONS / ACRONYMS

ASEAN	Association of South East Asian Nations
SAARC	South Asian Association for Regional Cooperation
IPEC	International Program on the Elimination of Child Labor
ILO	International Labor Organization
UNICEF	United Nations Children's Fund
UNCRC	United Nations Convention on the Rights of the Child
UCW	Understanding Children's Work
FDI	Foreign Direct Investment
C 138	ILO Convention No 138 (Minimum Age Convention)
C 182	ILO Convention No 182 (Worst Forms of Child Labor)
C 29	ILO Convention No 29 (Forced Labor)
C 105	ILO Convention No 105 (Abolition of Forced Labor)
UN ESCAP	The United Nations Economic and Social Commission for Asia and the Pacific
UN	The United Nations
LDCs	Less Developed Nations
GDP	Gross Domestic Product

CHAPTER 1

INTRODUCTION

1.1 Chapter Overview

The chapter introduces the thesis. It provides the background of the study followed by the problem statement, general and specific study objectives. Further, it discusses the significance and scope of the study. Finally, it describes the outline of this thesis.

1.2 Introduction

The child labor is the use of children in industry or business that harms or keeps them away from school (ILO, 2015). A robustness check on trade and child labor literature shows that demand for child labor is associated with demand for trading goods. One possible reason for the link between trade and child labor apparent in Romer and Frankel (1999) study shows that there is a strong association between trade and income as documented by Edmonds and Pavcnik (2005). The growing literature on the modern aspects of international trade sheds the new light on the effects of trade of differentiated products, its market size and the international division of labor. Whereas, the issue of child labor along with the trade of differentiated products is noteworthy and need to be debated in the age of new trade and globalization. The emergence of monopolistic competitive firms in developing countries has created the demand for educated and skilled workers in sufficient numbers to make it possible to create integrated production systems spanning more than one country. Economies that are taking part in international trade are expanding their potential beyond domestic borders to reach a global market. Trade liberalization is now becoming an important aspect of globalization, economic development and sustainability; nevertheless, the countries with accelerated growth in the labor intensive sector are viewed as contributing to the child labor violations (Chan, 2003).

The theoretical work on labor and trade is relatively rich as compare to child labor and trade (Estevez, 2010). The discussions on trade impacts on labor are segregated into different categories i.e., the effects of income deviations, effects of variation in the relative returns to labor suitability, interaction effects of factor endowments, and effects of labor policy. Generally, it is viewed that higher level of trade is associated with greater levels of income, and these are not necessarily homogeneously distributed, comfort from trade do not accrue to the rich alone (Collier and Dollar, 2002).

In the light of previous studies, this study discovers that research on child labor is rooted in one side of the story, namely the household's decision, which is related to the supply side of child labor, specifically based on income level of families. There are only a few studies such as Edmonds and Pavcnik (2005), Dinopolous and Zho (2007), Kis-Katos (2007) and Estevez (2011 and 2014) which examine the issue of child labor from the demand side. The mentioned studies assumed that children are employed in those sectors which are producing homogeneous goods and trade of homogeneous products affects the demand of child labor in exporting countries.

While, this study is focusing on production and trade of differentiated products by utilizing selection effect, this effect demonstrates that how changes in a number of firms parameter affect the relative demand of child workers. Particularly, trade induced child labor selection effect will determine whether trade liberalization is a remedy or exacerbate for child workers. Akram and Mahmood (2012) argue that trade of differentiated products is not as prevalent in developing countries as in developed countries, yet it is an important source of trade between the countries. Edmonds and Pavcnik (2005) confirm that such type of trade indirectly affects the child labor through overall market impacts and it can be observed with regard to child labor. Firms that are engaged in trade of differentiated products in developing countries usually related to those industries where the presence of child labor cannot be rebutted.

In the light of traditional interpretation of trade and child labor literature, Estevez and Levy (2014) find out that impacts of trade on the demand of child labor vary across sectors and the demand of child labor is a function of child worker productivity. Previously, Kurokawa (2011) examines that highly skilled workers are complement with low skilled workers in production of symmetric intermediate varieties. The findings of this study validate that the trade has an impact on the demand of both skilled and unskilled labor, the same is true in case of trade and child labor (Estevez and Levy, 2014). The trade literature postulates that the trade and

child labor are interconnected and the increase in the demand of labor can affect the demand of child workers (Edmonds and Pavcnik, 2005).

Estevez (2010) finds out that firms usually use both unskilled and skilled labor in the production of differentiated goods. The complimentary effect increases the skill premium and changes the demand of unskilled labor directly and indirectly affects the demand of child labor. In a similar style, Estevez and Levy (2014) examine the effect of low and high skilled workers' wages, and choices made by firms to upgrade their techniques of production. Since, the advancement in the technique of production is biased toward skilled workers, and less chances of unskilled labor or child labor in the production of differentiated goods.

1.3 Research Background

The issue of child labor has caught a great attention in the modern economics throughout the history of the world due to globalization (Hafner-Burton, 2013). The 19th and 20th century period, see the worst scenarios on screen; the industrialized nations, especially the Western countries have experienced the worst form of child labor. The child labor issue is worthwhile to investigate the existing studies and at what level counterweight to entrench the child labor with the help of economic tools. As a result of sincere and continuing efforts, the world has now recognized the importance of human capital, innovation and knowledge to counter the child labor. The world increasingly realizes that child labor is a blot for the

economy and it has adverse effects on the sustainable development (Swinnerton and Rogers, 1999).

Trade embargoes such as trade sanctions, outright bans and boycotts are popular remedies for child labor reduction. However, such instrument may have some unintended consequences for the poor countries. The ramification of traditional scholarly perception is that these sensations make the poor countries more unproductive, the only cure of child labor is prosperity via economic tools i.e. trade liberalization (Jafarey and Lahiri, 2002). Bhagwati (2001) states that the complex problems of an economy cannot be solved by sanctions; global issues such as child labor can only be solved by globalization. A trade sanction is a tool of antiglobalization; as a matter of fact, anti-globalization sentiments are most prevalent in the rich countries and poor countries are affected by global elites policies. The sentiments are directly destroying the industries of poor countries (Wood, 1995) and making the poor countries a destitute place (Hameed and Nazir, 2009). As a result, the poor countries become an unsafe place for children (Emerson and Souza, 2011).

The determinants of child labor are strongly related to macroeconomic variables such as performance of an economy, structural changes, behavior variations, and decision-making of economies. The demand and the supply side of the economy, both independently affect the child labor at the macro level. Poverty and inappropriate education system explain the child labor supply side, while the existence of a low paying informal economy is amongst the one of the major causes of the demand side of child labor. The other macroeconomic factors of an economy which determine the demand for child labor are the size of the informal economy, inflexible labor markets, instability of the firm's structure, the failure of industries to scale up the production and lack capacity to adopt the modern manufacturing technologies (Basu et al., 2010; Edmonds, 2010; Basu and Zarghamee, 2009; Edmonds and Schady, 2009).

Traditional economist usually sees this issue solely through the lens of labor demand. The great Smith (1776) talks about the value of children in labor shortage societies, later Engels (1892) emphasizes on the working conditions of children during the initial stage of the industrial revolution. Marx (1867) also has a very serious concern about children, because the replacement of machines with muscular power is now allowing child labor formally. Marx points out that labor supply is not transparent, capitalist and parents are major exploiters because in 18th century families are unable to meet their basic needs (Edmonds, 2007).

Even in the current century, the situation of child labor is a global reality bites, especially in the developing countries. Children are treated as an adult labor, they engage in dangerous work and often has to work very long time in hazardous activities for a very low pay, their earnings are only 10 to 20 percent of adult male labor wages (Abebe and Bessell, 2011 and Webbink et al., 2012). Edmond and Pavcnik (2005), Kis-Katos (2007) and Estevez (2011) have examined the child labor by using global factors such as trade liberalization that affects the demand of child labor. These studies assume that child labor is common in those sectors which produce homogeneous goods for trade. In the latter study, Estevez and Levy (2014) examine the

child labor in intra industry trade setting, they assume that children are nearly as productive as their adult counter partners and wage of child labor reflect the productivity of child workers.

Labor force can be categorized into skilled and unskilled labor (Bharadwaj, 2014), male and female labor, adult and child labor (Azmat and Petrongolo, 2014). Adult labor and child labor are close substitutes (Basu and Van, 1998) and child labor are economical and can be employed in any field of work at cheapest price as compared to adult labor (Hindman, 2011), but this justification for hiring the child labor is not tolerable in civilized and developed countries because child labor is deprived of happiness and enjoyment of childhood life (Nieboer, 2011). An historical overview of these western nations tells us that the developed nations were also victimized the children at the initial stage of development. This holds true in sample of developing countries of Far-East Asia which have low child labor incidences (Lin, 2011).

A global picture of child labor shows that out of total child workers 45% are in Asia, 40% are in Africa and only 6% in Latin America (ILO, 2012 and UNICEF, 2014). This figure imply that the child labor is still a colossal problem in Asia and Africa. In this context, the number of economically active children is highly concentrated in the poorest nations of Africa and Asia. The most serious concern is that many of these working children have to work in unsafe and sometime hazardous environment (UNICEF, 2014).

1.4 Definitions of Child Labor

Child labor is defined as all economic activities that involve the children under the age of 12 years (Islam and Choe, 2013). It also refers to a weekly economic exercise lasting enough to affect the education and health of those aged 12-14 (Dorji and Gyeltshen, 2013). In other words, it is defined as all hazardous work which could affect the health of children under the age of 18 (Aktar and Abdullah, 2013).

UNICEF (2008) interprets the child labor as work that violates a minimum number of hours, depending on the age of an adolescent and on the type of work, such work is considered dangerous to the child and should therefore be eradicated. Ages 5-11, at least one hour of commercial activity or 28 hours of family work per week. Ages 12-14, at least 14 hours of commercial activity or 28 hours of family work per week. Ages 15-17, at least 43 hours of domestic or economic work per week (Johnson and Bahemuka, 2013).

Ventevogel et al. (2013) define that child labor on the patterns of children's activities. Such as expressed it on the basis of the adverse impact of work on children's mental, physical, moral and social development as well as deprivation of educational opportunities. Different countries enacted and introduced various legislations, for combating the child labor, which depend upon the socioeconomic constraints of these countries. These legislations have proposed for defining child labor to find a solution to the problem. These definitions provide the base to make a law on child labor in SAARC region. On the basis of child labor definitions South Asian countries are applying minimum age acts according to their economic, political and social structures.

According to the international definition of employment, the child labor can be identified as economically active children if they work for at least one hour in a reference week¹. Thus, child labor in this research is a substantial as promulgated by the ILO and world development indicators, which is based on the agency's Minimum Age Convention No 138 (1973). Child labor is surprisingly difficult to define, this research establishes a normal definition and considered children under the age of 14, engage in economic activity.

1.5 Child Labor in SAARC and ASEAN

Figure 1 shows that child labor relatively looks small but if we compare in absolute term, it is intolerable in the South and South East Asian region, the population between 5-14 year old is under serious threats of hazardous work in industry and local family businesses. The countries like Nepal and Cambodia are facing a worse child labor situation due to limited trade facilities (Luong, 2011). India has relatively less child labor percentage, but the total volume of child labor victims is at an alarming level because of a huge population share in the SAARC region.

¹ The calendar week containing the 12th day of the month, which is used as the time period for documenting the employment and labor force status of respondents.



Figure 1: Percentage of Children Aged 5 to 14 Years Engaged in Child Labor Source: UNICEF Global Databases (2014)

India, Bangladesh and the Philippines are the leading countries in the Asia, which has a large number of children who are engaged in different kinds of production (Rahman and Khanam, 2012). In Bangladesh, there are more than 14 kinds of goods made by children, many of them are of an industrial nature, such as footwear, leather and textile including jute (Hindman, 2011). In the Philippines, most children are pushed into the labor force, and forcefully bounded to work on the family farm, but the most miserable, facts are to be, they found in mining pyrotechnics and pornography activities (Aldaba et al., 2003). India is the largest single producer of the juvenile workforce in the world due to the large population and significant poverty (Phillips et al., 2014).



Figure 2: Percentage of Child Labor in SAARC and ASEAN Source: UNICEF Global Databases (2014)

Figure 2 shows the phenomenon of child labor in SAARC² and ASEAN countries. The situation of child labor in Nepal is worst among all the countries of the region due to the intriguing relationship between adult minimum wage and child labor which is associated with conventional trade practices. Whereas, Cambodia is the worst country in South-East Asia due to limited trade facilities and war situations. In Bangladesh 13% children have to work in the labor market and most of the children are under age 15, while in Pakistan and Thailand 8 % children are engaged in child labor. Moreover, in India 12% children are engaged in child labor, but in absolute term the situation in India is the worst in the SAARC and ASEAN region.

 $^{^{2}}$ It is worth noting, that South Asia has 21.6 million children who are working, out of 246 million children between 5-14 years (ILO-IPEC, 2013).

The main cause of child labor in Indonesia, Philippines and Cambodia usually associated with the poverty and illiteracy level of their families especially their parents. Other exploratory factors of child labor are economic and social circumstances, high rates of unemployment, lack of awareness and lack of access to quality education (Kampan and Tanielian, 2014). Generally, child labor is condoned by the community (UNICEF, 2011) in numerous cases, adolescent³ expects to work as adults, especially in the agronomic setting (Hall et al., 2011).

1.6 Trade and Child Labor

During the period of 1990 to 2010, the world observes the dramatic changes in the technologies, labor markets and the structure of trade due to trade advancement in the world (Meschi et al., 2011). The West is adopting more efficient and capital intensive green technologies and shifting its older and less labor intensive technologies to developing countries (Kucera and Roncolato, 2011). These structural changes are now becoming the source of change in the mode of trade (Kumar, 2013), now the developing regions like SAARC and ASEAN are moving towards the trade of differentiated products instead of trade of homogeneous goods (Akram and Mahmood, 2012). This transformation is changing the form of child labor in developing regions. Firms that are engaged in the production of differentiating goods may benefit from technology in which unskilled humans are not highly valued, whereas

³ Adolescent, are considered as adults too early, in life often carrying out physical labor equivalent to adults (Lieten, 2011).

other firms would benefit from more skilled and trained workforce (Baland and Robinson, 2000).

The opponents of trade liberalization advocate that growing labor demand due to market integration, increases the earning opportunities of children and surely leads to more child labor. Some human right specialists recommend the practice of trade sanctions tools, especially on the exports of those countries which have high levels of child labor, as a means to eliminate the child labor (Kis-Katos and Sparrow, 2011).

Banerjee and Nag (2013) argue that child labor reduction can be achieved by trade liberalization and this reduction positively affects the labor productivity in a more than offsetting fashion. Past studies had focused on the child labor issue such as their education and poverty levels. However, studies on the nexus between child labor and trade appears to be limited (Kis-Katos and Sparrow, 2011). During the twentieth century child labor focus on the conditions of child workers (Hindman, 2011), however, during the 21st century, existing studies shift their focus to the determinants, demand and supply side of the child labor (Blau, 2012).

For instance, earlier studies centered mostly on the social cost of child labor (Nardinelli, 1980), the widespread existence of this issue have now fallen into the economic benefit of ending child labor (Glut, 1995). Figure 3 shows that the global benefits of ending child labor can be 4300 billion US dollars in one fiscal year, the South Asia only region which can attain maximum benefit from ending the child labor by trade liberalization.



Figure 3: Economic Benefits of Ending Child Labor via Trade Liberalization (In Billion US\$) Source: International Labor Organization (2002)

Firms in transition economies, Malaysia and Singapore, entered the market economy with a bundle of resources and classified partially industrialized, these countries have relatively less child labor as compared to other countries that's why the economic benefits looks small, but these countries have already enjoyed the economic benefits of ending the child labor in the last two decades (Kenayathulla, 2012). On the other hand, some countries of the ASEAN region like Cambodia, Indonesia, Philippine, and Thailand are still facing child labor and trying to draw a line between acceptable and unacceptable child labor (Kampan and Tanielian, 2014).

The political environment of reducing or eliminating child labor in the last decade of current century got the tremendous momentum due to awareness (Hindman, 2011) and trade

libralizaton at global level (Edmonds and Pavcnik, 2005). If the child labor situation is examined at the beginning of the 21st century, it was worse (ILO–IPEC, 2013), but the continuous struggles of non government organizations (NGOs), global collaboration and modernization compel the working children to come out the economic activity (Mohapatra and Dash, (2011).



Figure 4: Worldwide Child Labor

Source: ILO–IPEC (2013)

Figure 4 shows that, at the beginning of the 21st century, at global level, the conditions of factory work were needlessly unhealthy and oppressive for working children that were 246 million in 2012, but in 2012 this trend decline to 168 million, this is a great achievement in the child labor reduction activities. Similarly the working children in hazardous activities also

decline from 171 to 85 million (ILO-IPEC, 2013). It is interesting to note that the work on elimination of child labor is in the right direction, but the target of international child labor organizations is to eliminate the worst form of child labor⁴ at the end of 2016, it seems difficult but achievements are hopeful (van de Glind, 2010).

The tendency of the decline in child labor is mainly because of trade liberalization. Fors (2012) illustrate that the trends in participation rates, for the children, as child labor 10-15 years between 2000 and 2012 are decreasing, from the year 2000 and on words the declining trend of child labor in the world seems to slow down surprisingly. In these years public officials and the media have shown a great sympathy in child labor, along with social awareness, the effect of economic growth on child labor starts to be left distinctly by increasing trade trends in developing countries.

1.7 New Trade Theory and Child Labor

Krugman's (1979) framework of international trade states that the industrial sector having features of increasing returns to scale⁵ and total number of firms in the industry shrinks although each one of the remaining firms produces more than before (Feenstra, 2003). Trade expands the consumption basket and enables the consumers to use both local and imported

⁴ Harmful to the child either psychologically or physically (Dessy and Pallage, 2005).

⁵ Reduction in cost per unit resulting from increased production (Balassa, 2013).

varieties of goods. At the same time, when income level of a country increases due to trade openness; the country implements stricter child labor approach, which causes amplified child labor reduction activity (Edmonds and Pavcnik, 2005; Edmonds, 2015). The increasing returns to scale in Krugman's framework triggers the producer to hire more efficient and skilled labor to raise the production level. This framework has the potential to support the child labor elimination policies and discourage the child labor, because the producer wants to produce more and additional unit of adult labor is not expensive for him due to increasing returns to scale. In this framework, there is a selection on the production side and the consumption side. The producer selects best available labor in the production process for making differentiated products (Miles et al., 1978) and the consumer has an option to select a variety of differentiated products (Krugman and Venables, 1996).

Krugman's framework (1979) on international trade find out that, only more competent firms have a potential to upgrade their scale of production by using more skilled workers due to increasing returns to scale (Krugman, 1979). Estevez (2011) concludes that child labor is an inappropriate option for producers than unskilled adult labor even after accounting for productivity differences. Estevez and Levy (2014) assume that, in the production of differentiated product, child labor is not a better option and productivity differences make it an expensive option for the producers.

1.8 Problem Statement

Child labor has become an issue that invokes passion because it brings together people concerned about the exploitation of children on moral and ethical grounds, some of child labor issues are managed at national level while others are internationally workable (Edmonds and Pavcnik, 2005). Trade liberalization is combating the problems that are connected with the child labor and converting local constraints into an aggregated global constraint (Emerson and Souza, 2011).

The effects of trade liberalization on child labor are universally debated and aroused great sympathy among reformers (Abebe and Bessell, 2011). From a theoretical point of view the effects are ambiguous (e.g., Basu et al. 2010) as trade liberalization works probably through various routes (such as Casson, 2012). There is a conventional view among economists that trade and child labor issues should be treated independently and separately (Basu and Zarghamee, 2009). Nevertheless, this approach need to be supported theoretically in the process of trade of differentiated product models as low wages generate child labor and upset the effects of trade liberalization on child labor, national welfare and income distribution (Doran, 2013).

It is widely recognized that international trade raises the economic welfare, but, if a production is labor intensive then international trade leads to some detrimental effects on child labor. The existing literature on trade and child labor provides mixed evidence due to the

structure of developing economies, sometimes it is beneficial in the short run and harmful in the long run or vice versa (Edmonds and Pavcnik, 2006).

The incidence of child labor in the presence of monopolistic competitive firms and trade of differentiated products between structurally identical countries is remained ambiguous in the existing literature (Estevez, 2010). Furthermore, the effects of production of differentiated goods are mostly ignored in a stream of studies about the impact of trade on child labor (Davies, 2005 and Cigno et al., 2002). The effect of strict regulation by using different types of penalties is still questioned in case of production of differentiated goods because these effects depends upon the firm's production process which relatively affects the productivity of child labor and adult labor (Estevez, 2011).

The trade literature suggests that if a country engaged in the trade of homogeneous goods, then unskilled labor force is demanded especially in the traditional agricultural sector. As a result, the demand for the child labor in the economy increases. On the contrary, another stream of studies deliberately deviates from the conventional trade theory to new trade theory and postulate that highly skilled and well-educated workers are required to produce differentiated goods. Therefore the trade of differentiated goods has resulted in a reduction of demand for child labor (Estevez and Levy, 2014).

The trade of differentiated goods is not free from child labor effects. They certainly enhance the child labor, the nature of trade induced child labor effects need to be analyzed in
the trading nations, and whose responsibility is to evaluate child labor effects of trade. These concerns are frequently discussed in trade and child labor literature (Diller and Levy, 1997). The majority of studies investigates the child labor effects of trade by using traditional trade theories and find mixed evidence of the research. Despite the substantial number of studies that have investigated the child labor effects of trade, to date, only a few studies which investigate the impact of new trade on child labor. A large body of research, primarily in the SAARC region, narrates that trade appears to be a source of reduction in child labor (Topalova, 2010 and Ray, 2000), but recent developments in new trade theory open the new era for research (Iqbal et al., 2012).

On this note of course, it has raised the concerns regarding the structures and processes generated by trade liberalization, which can affect the child labor in a monopolistic competitive environment. Furthermore, a paucity of research that investigates the effects of trade of differentiated products on child labor offers another motivation for this dissertation to embark upon the issue of child labor and trade. So, It will be interesting to investigate the impact of trade expansion and child labor, along with the nature of trade induced child labor effects.

In the recent years, the impact of globalization on the child labor has started to spark both public and academic debates. The role of multinationals in world trade and globalization is increasing and the concerns regarding the child labor and development policies in SAARC and ASEAN regions are also mushrooming. Under these circumstances, this dissertation tries to find out the impact of trade liberalization and stringency in child labor policy which can affect the firm's labor amelioration⁶ process, its scale of production, consumption of goods and the number of firms in the economy.

A review on a bulk of literature reveals that the current models of trade and child labor collectively does not address the following comprehension queries. The most prominent concern is the trade of differentiated products that can influence the child labor in the economy (Sapir, 1995). Estevez (2010) examines that there is scarcity in the trade and child labor literature regarding the trade liberalization that can affect the firm's labor amelioration process, its scale of production, technique of productions and the number of firms in the economy. This research is an attempt to fill this theoretical gap by reviewing the research literature and articulate an expansionist theory of new trade and child labor by using trade liberalization affects on the demand for child labor from domestic and foreign firms.

Several studies have attempted to correlate the trade and child labor in traditional trade theory, to the best of this study knowledge, there is a paucity of empirical studies that tries to systematically demonstrate the effects of trade on child labor under the new trade theory. Thus, this thesis is an attempt to sort out the effects of trade on the child labor in the South Asian Association for Regional Cooperation (SAARC) and the Association of Southeast Asian Nations (ASEAN) countries by applying the new trade theory. Now the question arises why

⁶ The act of making something better, in this study the allocation of resources from output towards hiring more skilled workers and terminate the less skilled workers (or child labor) engaged in the production process is used to express the act of amelioration.

SAARC and ASEAN? There are many reasons for taking theses regions, but the most important reason is that the SAARC and ASEAN region are two pivotal regional platforms for economic cooperation in Asia. Multiple stress of accelerating population growth rates, rural to urban migrations and globalization is major contributing factors of child labor in SAARC and ASEAN regions and sensitivity of child rights issue making it an issue of priority demanding immediate attention.

1.9 Objective of the Study

The general objective of this research is to propose a trade and child labor framework under the new trade theory. The purpose of this framework is to give a new idea of increasing returns to scale and monopolistic competitive environment which allows this research to depart from the traditional trade and child labor models that follow the constant returns to scale and perfect completion.

To achieve the general objectives, this research aims at addressing the following specific objectives:

- To investigate the empirical relationship between the trade and child labor in the major SAARC and selected ASEAN countries.
- To assess the incidence of child labor in the presence of production and trade of differentiated products in the selected countries.

3. To investigate the nexus between child labor and the trade induced scale, technique and selection effects for SAARC and ASEAN countries.

1.10 Significance of the Study

As far as this research concerns, this is the first step to empirically address the child labor effects of trade of differentiated products and is the first to provide the evidence of the significance of controlling the child labor by selection effect. The important aspect of the foregoing analysis is that it focuses on the implication of child labor penalty in a theoretical framework and the impact of trade liberalization on child labor by using empirical analysis in the markets where increasing returns and monopolistic competition prevails, policy effect yield selection, technique and scale effects.

This thesis gives a direction to fill the gap in the literature by developing the new theoretical and empirical evidence of trade and child labor. Besides the empirical evidence of trade and child labor, this research shows the trade induced child labor effects in distinguishing products that can be disintegrated into a selection effect, a scale effect, and a technique effect. In addition, the upcoming theoretical and mathematical formulation developed in this thesis assumes that firms are taking part in withdrawing the child workers from the industry due to child labor penalty. This approach gives a new line for the researchers by incorporating the heavy child labor penalties in the model as a policy variable. On the other hand, this dissertation

explores the trade induced child labor selection, scale and technique effects by using empirical strategy.

It is hoped that the present thesis on trade and child labor will guide and stimulate the further work in the sphere of new trade theory. It will be a valuable addition to the trade literature to control the child labor by using market forces and economics tools. Not only this study will give the new direction to trade policymakers in promoting the trade in the SAARC and ASEAN regions, but also it will support the arguments of human rights activists who want to eliminate the child labor.

1.11 Organization of the Study

This study is divided into six chapters. Chapter 1 focuses on the introduction of the study, covering statement of the problem, significance of this thesis and description of the study area objectives. Chapter 2 gives a general overview of child labor in SAARC and ASEAN countries. A comprehensive review of the literature relevant to the issue under consideration is given in Chapter 3. The data used and methodology developed in this dissertation are explained in Chapter 4. Chapter 5 presents the main findings of newest estimates and trends of child labor in SAARC and ASEAN countries. Finally, Chapter 6 presents the conclusions, limitations and recommendations of the research.

CHAPTER 2

AN OVERVIEW OF SAARC AND ASEAN

2.1 Chapter Overview

In this chapter, a brief overview of the Association of South East Asian Nations (ASEAN) and South Asian Association for Regional Cooperation (SAARC) economies is discussed. The child labor situation in selected ASEAN countries and major SAARC countries is also part of this chapter. Moreover, different definition of child labor and major global and selected regional development is also supplied in this chapter.

2.2 Introduction

Two important regional associations of Asia, ASEAN and SAARC, consist of more than seventeen different economies. These economies vary from each other in term of their political structure, degree of independence, size and performance and other socioeconomic indicators. However, some associations can be drawn among these regional economies. This section will briefly examine the overall economic structure, trade flows and patterns in the SAARC (South Asian Association for Regional Cooperation) and ASEAN (Association of South East Asian Nations) region.

2.3 Background of the ASEAN Countries

South East Asia is a rapidly growing region as a tourist destination and global hub for economic cooperation (Van Soolingen et al., 1995). It is the integration of two heterogeneous segments; insular South Asia, a string of archipelagoes to the south and east of mainland and mainland South Asia, a continent projection (Emmerson, 1984). Extending more 1,100 kilometers southward from the mainland into Southeast Asia there came a Malay Peninsula; this is a geographic part of the mainland South Asia, and it shares many primitive and derived features with the surrounding islands along with ecological and cultural affinities (Dick and Rimmer, 1998) and thus works as an economic junction between the two regions (Terwiel, 2013).

2.3.1 Overview of ASEAN Economies

Association of South East Asian Nations is a set of heterogeneous associates with respect to their origin, liberalization, economic performance and system of governance. These countries were colonies of different powers, i.e. French, Spanish and British. These countries adopt their legal framework of French civil law and English common law with some amendments according to their socio-political and cultural suitability. The performance of these regional economies differs markedly in term of human capital, population growth, and employment in relation to each other.

After the World War II, European oppression on the South East Asia, Indian Subcontinent and the Middle East, now, was ungovernable. Especially after the Japanese occupation of the British, Dutch and French territories shattered the image of European preeminence during wartime. The destabilization of European rule led to the rapid growth of nationalist movements in Asia especially in Indonesia and Malaya. So, these rulers have to liberalize the occupied areas. In the aftermath of independence the nations of Southeast Asia establish the Association of South East Asian Nations in 1967.

Mainland South Asia is split into the countries of Cambodia, Laos, Myanmar, Thailand, and Vietnam; the small city state of Singapore located off the southern tip of the Malay Peninsula (Miettinen et al., 2011). Cambodia, Laos and Vietnam located at the eastern chunk of the mainland and collectively these countries are known as Indochinese Peninsula (Emmerson, 1984). Malaysia occupies the both mainland (Peninsular) and insular (Borneo) part of island Borneo island, along with the small sultanate of Brunei, the rest of insular South East Asia contains the archipelago nations of Indonesia and Philippines (Miettinen et al., 2011).

The confederation of Southeast Asia has a major impact on the dissemination of the region's economic development, and it formulates uneven patterns of population progression and economic expansion (Dufhues et al., 2011). It anathematizes a strong sense of class distinction and transpires a larger discrepancy between the wealthy and poor nations (Cook and Pincus, 2014). The great depression of the late 1920's and 1930's puts a stop to the economic development of the region (Evers, 1987). Unemployment rose sharply, and the period shaped

the seeds of political revolution that ended with the independence of most of the region's countries after World War II (Aspinall et al., 2011). However, the rise of development policies in the 1940's and 1950's made the economic development strategies of all the capitalist Southeast Asian States. The policies have drawn attention to the urban industrial planning (Berger, 2003), while agricultural development has been observed as complementary to industrial growth (Hort and Kuhnle, 2000). So far, these initiatives have met with mixed success (Berger, 2003). Indeed, international differences grew progressively (Cook and Pincus, 2014) and the trading patterns of the East Asian countries, however, have shifted over the time and has continued to be exporting raw materials and importing manufactured goods (Sally and Sen, 2011). Singapore has advanced the level of industrialization and became one of the world's great centers of industry and commerce (Chongvilaivan, 2012).

There is a great disproportion in the development of the region (Siah et al., 2014), particularly between the member and nonmember of the Association of Southeast Asian Nations (ASEAN) countries, i.e. Brunei, Indonesia, Malaysia, Philippines, Singapore and Thailand have experienced significant economic development (Sharma and Chua, 2000). However, the economy of the Philippines has grown at a much slower rate compared to other ASEAN countries (Gill et al., 2007). The countries, i.e. Cambodia, Laos, Myanmar and Vietnam could not follow the miracle path of sustainable development (Noland, 2013), that's why these countries are among the poorest nations in the world (Salazar and Das, 2007).

Since 1970, there are large differences in the patterns of structural change across the region (Serrano and Pinilla, 2014), with the cumulative growth projections in Malaysia and Thailand and collateral filters of labor laws allow only limited progress in the Cambodia, Laos, and Vietnam (Leung and Viseth, 2005). Hunger, food insecurity and severe malnutrition are fundamental challenges in the pockets of even the most developed countries (Timmer, 2014), but, they have been especially vulnerable in Cambodia because of internal conflicts and crop failure and the disorder of agronomy everywhere is plainly associated with restrained land, functionally the landless remains poor and ultimately experiencing the great poverty and poor health (Sedara et al., 2002). Landlessness peasants are perhaps most serious in the Philippines (Melgar-Quinonez et al., 2006).

Agriculture is a key source of income and an important trigger of economic growth in the region except Brunei and Singapore (Qiu et al., 2007). Agricultural employment, however, has been declining steadily due to rise in employment in the services (Park and Shin, 2012). More than two thirds of the workforce of Cambodia and Laos gain their livelihood directly from agriculture (World Bank Group, 2012), which is one of the most fundamental practices for the survival and wellbeing on a subsistence basis (ILO, 2006). Evidence from micro-studies of the ASEAN countries suggests that there is a considerable restructuring in the region toward the growth in industry and services (Hutton, 2003), however, these countries have still huge potential for further economic consolidation (Terada, 2011).

	Independe	ence		Government		
	Date	From	Name	Nature	Legal System	
Brunei Darussalam	1-Jan- 1984	United Kingdom	Negara Brunei Darussalam	Constitutional Sultanate	English Common Law. For Muslims, Islamic Sharia Law	
Cambodia	9-Nov- 1953	France	Kingdom of Cambodia	Multiparty Democracy under a Constitutional Monarchy Established in September 1993	Primarily a Civil Law Mixture of French-influenced	
Indonesia	17-Aug- 1945	Japan	Republic of Indonesia	Republic	Roman-Dutch Law	
Laos	19-Jul- 1949	France	Lao People's Democratic Republic	Communist State	Traditional Customs, French Legal Norms, and Procedures	
Malaysia	31-Aug- 1957	United Kingdom	Malaysia	Constitutional Monarchy	English Common Law	
Philippines	12-Jun- 1898	Spain	Republic of the Philippines	Republic	Spanish and Anglo- American Law	
Singapore	9-Aug- 1965	Malaysian Federation	Republic of Singapore	Parliamentary Republic	English Common Law	
Thailand	1238 ^a		Kingdom of Thailand	Constitutional Monarchy	Civil Law System	
Vietnam	2-Sep- 1945	France	Socialist Republic of Vietnam	Communist State	Communist Legal Theory and French Civil Law System	

Table 1: Establishment and Systems of Governance (ASEAN)

(a) Traditional founding date, never colonized

Source: Central Intelligence Agency (2010)

Industrialization in the Southeast Asia is a primarily an indigenous phenomenon (Terwiel, 2013), much of the development policies have critical goals in the economies of the ASEAN countries (Dick and Rimmer, 1998); and, in all of them except Brunei, industry's productivity has grown considerably over the period (Hutton, 2003). The most apparent increases have acknowledged in Singapore, Thailand and the Philippines (Vallance, 1999). Small-scale corporation dominates in the region, both in term of the number of workers employed and the number of companies and agriculture product processing industries has a potent role in increasing concentrations of consumer goods industries in all nations (Phadungkiati and Connell, 2014). The manufacture of a variety of products, headed by electronic, electrical and transportation equipment industry is dominant, the conspicuous exception is Singapore (Athukorala, 2011). Textiles and clothing are considered as major sectors in Thailand, Myanmar and Philippines (Ofreneo, 2012), the growth of the chemical industry in Thailand increases due to robustness of domestic demand (Jayanthakumaran et al., 2013).

The value of major ASEAN countries' trade is about more than one third of the United States of America (Péridy, 2005). The most astonishing thing is that these countries are now influencing the global trade by using the market powers (Aizenman et al., 2011). On the other hand minor ASEAN countries are struggling for getting some share of trade from the ASEAN region. Exports, as a percentage of the GDP, are small in Cambodia, Myanmar, Vietnam and Laos and moderately in Thailand, the Philippines and Indonesia, extremely large proportion in Singapore, Malaysia and Brunei (WDI, 2014). The commodity composition of export portfolios

has got a paramount importance (Siah et al., 2014). In this aspect, Indonesian trade structure, dominated by oil, has been comparatively effective in expanding its exports toward, coffee, rubber, textiles, rattan and plywood (Basri and Hill, 2008).

Characteristics	Brunei	Cambodia	Indonesia	Lao PDR	Malaysia	Philippines	Singapore	Thailand
GDP Growth Rate	0.95	7.31	6.26	8.02	5.64	6.80	2.50	7.67
Per Capita Income, (Current US\$)	41126.6	945.5	3551.4	1408.3	10440.0	2587.6	54007.3	5479.8
Exports (% of GDP)	81.36	62.79	24.29	38.84	85.25	30.78	195.07	74.97
Inflation	0.46	-	4.28	4.26	1.66	3.17	4.53	3.01
Unemployment Rate	3.80	0.20	6.10	-	3.00	7.00	2.80	0.70
Poverty Headcount Ratio	-	17.70	12.00	23.20	1.70	25.20	-	13.20
Population (Million)	0.41	14.86	246.86	6.65	29.24	96.71	5.31	66.79

Table 2: Socio Economic Indicators of ASEAN Countries

Source: World Bank (2012)

In order to observe the major socioeconomic indicators like GDP growth, GDP per capita income, export percentage of GDP, inflation, unemployment rate and population in million are shown in Table 2 for the purpose of quick analysis of South East Asian countries.

The table depicts that the trend of per capita income in Brunei Darussalam and Singapore is admirable as compare to the other partner, while the GDP growth rates of Cambodia and Lao PDR are satisfactory. Unemployment rate of Indonesia and Philippines is high compared to other developing economies of ASEAN.

On the other hand, Malaysia, with a trade pattern of exporting palm oil, tropical hardwoods, and tin, now derives the substantial part of their export income from petroleum products (Ramasamy and Yeung, 2007). These earnings are stimulating the country's industrial base (Basri and Hill, 2008; Ramasamy and Yeung, 2007). Thailand attainments are much less from Petro-export structure, where food and differentiated goods and services account for almost all of its trade (Sawyer et al., 2010). On the contrary, Brunei relies entirely on its petrol-exports. Singapore is the only economy, which has utilized its unique geographic importance and the most competent human capital is available here to attract multinational corporations (Feridun and Sissoko, 2011).

Cambodians per capita income is expeditiously increasing, but it is very low as compared to other regional countries (Murakami, 2013). In 2010, Cambodian real per capita income was \$2,470 (purchasing power parity) and nominal per capita income was \$1,040 (UN ESCAP, 2010). Agriculture sector required substantial restructuring due to vast potential; most of rural households are dependent on agriculture and complementary sub-sectors (Sedara et al., 2002). The major exports, which are highly concentrated, are fish, rice, garments, timber, and

rubber (Sotharith, 2011). Tourism is considered as a fastest growing industry in Cambodia, with arrivals increasing from 0.2 million to over 2 million in recent years (Kelsall and Seiha, 2014).

Indonesia has an assorted economy in which both the private and public sectors are playing a growing and dynamic role in strategic adaptation of policies to control the feasible growth path. The country is an unprecedented economy of scale and dramatically increasing throughput and reducing the cost of the product as compared to other countries of Southeast Asia (Basri and Hill, 2008). Indonesia is the 27th biggest exporting country in the world (Durotoye, 2014) and the leading export commodities are oil and gas, plywood, rubber, electrical appliances, and textiles (Yusoff and Febrina, 2014). The gross domestic product is about \$1 trillion due to this huge share of the industrial sector, Indonesia is now a significant member of G-20 major economies (Felman et al., 2014), it is the world's 10 largest by real GDP, and its share in global economic output is 2.3 per cent (Laksmana, 2011). The industry sector is the economy's leading sector contributing 46% of annual GDP, the second substantial sector is services sector, having a share of 39% in the economy's total income and another significant segment of the economy is an agricultural sector, adding 14% income to economy annually (WDI, 2012). However, the service sector has engaged 48.87% of the total labor force more people than other sectors, traditional agriculture sector is giving chance to 38.3% of the entire labor force and industry sector occupying 12.79 % of the overall labor force (Wahid and Furuholt, 2012). The population of Indonesia is more than 237 million, with a 1.9% population growth. Java is the world most populous island and 58% of the population lives in Java. The population of Indonesia is expected to grow to around 269 million by 2020 (Akbar et al., 2012).

The economy of the Philippines is one of the leading economy in the world and its rank is 75th, with an estimated gross domestic product \$272.02 billion in 2013 (Nalla and Mamayek, 2013). The major exported items are semiconductors and electronic products, copper products, garments, petroleum products, transport equipment, fruit, and coconut oil. Multinationals in newly industrialized country are facing the dilemma of whether and when they can engage their capital. That's why the Philippines' economy is in evolutionary stage and has transitioned from agriculture to services and manufacturing (Athukorala, 2011). Labor force is around 37 million and mostly engaged in agriculture sector. The agriculture sector is giving the employment to 34.84% of the entire labor force, while contributing to GDP only about 16.12%. The industrial sector is facilitating 14% of the workforce and accounts for 29.8 % of GDP. At the same time, the 46.7% of the workers participating in the services sector are adding 55.9% of GDP (Bayangos and Jansen, 2011).

After Indonesia, Thailand is the second major socioeconomic player of the Southeast Asia; Thailand is a transformative industrialized country (Nguyen and Pham, 2011), and export dependent economy, export to GDP ratio is virtually increasing from 1960 (Sachdev, 2013). The share of industrial and service sector is more than 40 per cent (Nguyen and Pham, 2011), which highest in the entire GDP (Sachdev, 2013). Thailand's agricultural sector is a traditional sector (Rigg et al., 2012), which produces 8 per cent of GDP (Tongzon, 2002). The trade of logistics and communication sectors account for 13% and 10 % of gross domestic product respectively. The construction and mining sector contribute only 4% of the country's GDP.

Other sectors i.e. the education, financial, hospitality, telecommunication sector accounts for about one fourth of the country's gross domestic product (Gödecke and Waibel, 2011).

2.3.2 Child Labor in ASEAN

The ASEAN countries deserve rebuke for its working children during the past few decades, Cambodia, Thailand, Indonesia and Philippines are laced with child labor problem in this region. The IPEC finds out that Indonesia and Thailand involve extreme risks of child labor due to uncertainties and poverty. While Cambodia and Philippines pose extreme child labor complicity risks due to low scale industries (ILO, 2003).

It is generally believed that Cambodia's economic growth and development are a causal factor to the increase in the child labor. The dramatic increase in the demands of the construction is one example where construction industry has pushed children to work in brick kilns, foreclosing the opportunity of schooling for most of them (Ang et al., 2012). ILO (2003) reported that one in every ten children in the Phnom Pehn, the capital and largest city of Cambodia, above the age of seven years is involved in child domestic labor; young children have to work in the field work as scavengers and spend their time for rummaging in dumps for items that can be sold (Kim, 2011). Some children spend their time in the streets peddling. The tourist industry is also playing a key role in this form of child labor as many tourists are willing to buy from children, out of good intentions and social responsibility, mounting the demands. This strengthens the beliefs that teenagers are more appreciated in the streets than at school

(Maeve, 2011). World Bank (2008) confirms that 48% children from aged 5 to 14 are not attending the school and they are economically active (Kim, 2011). Many of these children have to work long hours and approximately 79,000 children between the ages of 6 to 17 worked 25 hours a week and did not attend the school (Despotis, 2005). When children are out of school, they are denying the skills and knowledge needed for state development. Without vital life skills and training, they are vulnerable to abuse and misuse, which may complicate and exacerbate the cycle of poverty in their families. Consequently, this lack of education will hold back economic growth in Cambodia. There is a need to re-estimate the child labor in Cambodia because child labor could be missing out on education (Kim, 2011).

There are more than three million children involved in hazardous work in the Philippines. Out of 29.8 million children aged 5 to 17, 45% are doing taught work and mostly boys are engaged in hazardous work (Gamlin et al., 2013). Sometime, these boys have jobs that place them in instant physical danger. These treats contain the experience of injurious chemical and sharp tools, that can be less recognizable but not less hazardous. Teenagers are obligatory to work very long time with a few breaks. Some of them are abused by their bosses, both substantially and psychologically (Hindman, 2009). Children in risky work are prevalent in the Bicol, Central, Luzon, Mindanao, Northern and Western Visayan island regions (Reyes et al., 2014).

Indonesia, the 4th most populous (Durotoye, 2014), and the largest Muslim population in the world (Abduh and Omar, 2012), about 30 per cent of the entire population are under age 15

(Foldvari et al., 2013). A significant effort has been made in expanding the access to school education, still, more than 4 million school age children do not attend school (Kis-Katos and Sparrow, 2011). Most of teenagers enter the workforce at a very early phase, and are helpless for exploitation in the worst form of child labor (Sen, 2013). Recent estimates indicate that there are more than 2.7 million children engaged in repeated antisocial acts with some involved in the worst form of child labor (Sim et al., 2013).

Thailand has made considerable efforts to eliminate the worst forms of child labor, therefore, set a minimum age for domestic work at the age of 15. For this purpose government passed the parliamentary regulations for domestic workers, which offers protections for domestic child workers, also updated the list of dangerous work (Hepburn and Simon, 2013) and working circumstances banned to teenagers and change the minimum age in sea vessels from 15 to 17 (Ratner et al., 2014). Children in the agriculture sector are exposed to unsafe tools and pesticides. Most of children face such problems in the sugar cane fields, but, only a few children face such situations in the production of roses, rubber, and oranges (Siddiqi and Patrinos, 1995). Children are taking part in shrimp and seafood processing industry and are subject to long, late hours in hard working situations and involve in heavy lifting (Ratner et al., 2014). Teenagers, mainly girls, sometime have to work in local services, and endure physical and sexual abuse from their employers (Sychareun et al., 2011). Young children are rewarded to fight in daring boxing called Muay Thai (Li and Wang, 2011). Teenagers are found in money making sexual abuse, including pornography. Children from Cambodia and Indonesia are traded

in Thailand for commercial sexual exploitation (Sychareun et al., 2011). Children are also trafficked into urban areas to sell and beg on the streets (Sankharat, 2013).

2.4 Background of the SAARC Countries

SAARC includes eight countries: Afghanistan, Bangladesh, Bhutan, India, Maldives, Nepal, Pakistan and Sri Lanka. Its entire population is approximately 1.64 billion, which represents 23% of the world population. SAARC has many distinct qualities; all ideologies, faiths and religions of the world are gathered here. It has an imaginative irrigated farmland in the world with over 66 million hectares. It has one of the oldest living civilization in the world. It acquires the second largest rail network in the world after the United States of America. This region holds 450 million people in the workforce, which is the largest of any other economic zone. Consumer base surpasses 750 million reporting the largest number of consumers in a single economic zone in the world (Herath, 2012).

2.4.1 Overview of SAARC Economies

As far as, the SAARC is concerned, it is considered 3rd largest economy of the world in term of GDP (derived at purchasing power parity) and contributing US\$9. 05 trillion in global GDP, on the other hand, if it is observed for SAARC countries, it is at 8th number having share of US\$2. 5999 trillion (on the base of nominal GDP). SAARC nations represent 23% of the

world population and comprise 3.9 % of the world's area (Knox and McCarthy, 2014 and Chand, 2010).

Characteristics	Afghanistan	Bangladesh	Bhutan	India	Maldives	Nepal	Pakistan	Sri Lanka
GDP Growth Rate	14.43	6.52	5.07	4.73	1.33	4.85	3.50	6.34
Per Capita Income, (Current US\$)	690	950	2320	1550	5430	700	1250	2910
Exports (% of GDP)	5.51	20.16	38.72	23.99	111.31	10.70	12.39	22.83
Inflation	7.21	6.21	10.91	9.31	12.13	9.45	9.68	7.54
Unemployment Rate	8.7	4.5	2.1	3.6	11.1	2.7	5	4
Poverty Headcount Ratio	35.8	31.51	12	21.9	-	25.2	12.4	8.9
Population (Million)	29.82	154.69	0.74	1,237	0.33	27.47	179.16	20.32
Population (Share %)	1.81	9.38	0.04	74.98	0.02	1.67	10.86	1.23
Population (Growth Rate)	3.04	1.6	2.3	1.4	1.9	2.0	2.3	0.9

Table 3: Socio Economic Indicators of SAARC Countries

Source: World Bank (2012)

Table 3 shows that, SAARC economies are experiencing the economic crunches in the shapes of high inflation i.e. India, Maldives, Nepal and Pakistan, by comparing with per capita

income of these countries. High population growth rates and poverty indexes in Bangladesh, India, Pakistan and Nepal are changing the consumption pattern and labor composition in the region. Export to GDP ratios in the major SAARC countries like Bangladesh, India, and Pakistan are unsatisfactory for globalization. Under these circumstances there are high chances of child labor in these countries.

All SAARC countries stock market capitalization is 3.78% of the global stock market capitalization. The SAARC contribution of FDI in world FDI is 3.26%. Given tranquility and peace, FDI can register massive expansion as enabling environment scan enable SAARC to capture a big share of mutual benefits for SAARC and global world (Kanungo, 2012). Table 3 shows a short summary of socioeconomic indicators, like GDP growth, GDP per capita income across, export percentage of GDP, inflation, unemployment rate and population, for the purpose of quick analysis of South Asian countries. The table depicts that the trend of per capita income in small countries i.e. Bhutan and the Maldives, is admirable as compared to major SAARC countries, on the other hand the inflation rates of these countries are very high. Unemployment rate of Afghanistan and Maldives is high compared to other economies of SAARC.

At first glance, SAARC countries seem to be homogeneous set of countries in term of their time of independence. All the countries in the Table 4, with the exception of Bhutan and Nepal used to be British colonies. Afghanistan is not included in the Table 4 due to different socioeconomic and geographical importance. Countries which were British colonies have almost similar governing structure and legal system from their British rulers. Eight countries of the SAARC namely; Afghanistan, Bangladesh, Bhutan, India, Maldives, Nepal, Pakistan and Sri Lanka differ greatly in land area, GDP, population, trade patterns and way of governance. They have common borders with one another and these countries have more or less similar level of human resources and economic development (Zaman et al., 2013).

	Independence			Government		
	Date	From	Name	Nature	Legal System	
Bangladesh	16-Dec- 1971	West Pakistan	People's Republic of Bangladesh	Parliamentary Democracy	English Common Law	
Bhutan	8-Aug- 1949	India	Kingdom of Bhutan	Monarchy; special treaty with India	Indian law and English Common Law	
India	15-Aug- 1947	United Kingdom	Republic of India	Federal Republic	English Common Law	
Maldives	26-Jul- 1965	United Kingdom	Republic of Maldives	Republic	Islamic law and English Common Law	
Nepal	1768 ^a		Kingdom of Nepal	Parliamentary Democracy and Constitutional Monarchy	Hindu Legal Concepts and English Common Law	
Pakistan	14-Aug- 1947	United Kingdom	Islamic Republic of Pakistan	Federal Republic	English Common Law	
Sri Lanka	4-Feb- 1948	United Kingdom	Democratic Socialist Republic of Sri Lanka	Republic	Mixture of English Common Law, Roman-Dutch, Muslim, Sinhalese, and Customary Law	

(a) Unified by Prithvi Narayan Shah

Source: Central Intelligence Agency (2010)

India is a large country in this region with vast land area, huge population, the greater share of GDP and GNP in the SAARC and a large trading country as compared to other SAARC associates (Sharma, 2007). Pakistan and Bangladesh have some type of similar characteristics in this region. These two countries have identical GDP, GNP and population structure as compare to India. Their combined share of trade in the SAARC is comparable to India (Mahmood and Azhar, 2001). Sri Lanka is a small country in this region with a small share of total GDP, GNP, population and land area of SAARC. Trade share is also not very high, but Sri Lanka has its own importance in the region (Sharma, 2007).

The estimated gross domestic product of Nepal for being at over approximately \$19 billion, agriculture, services and industry accounted for 36 %, 48%, and 15% respectively of the entire GDP. While the industry and agriculture sector is contracting, the involvement of the service sector is increasing in domestic income. Agriculture sector holds 76% of the labor force, services sector employs 18.1% and manufacturing carries industry 6.2%. Agricultural yield includes rice, corn, tea, sugarcane, root crops, milk, wheat, and buffalo meat. Industrial sector generally involves the processing of agricultural products, including grain, tobacco, sugarcane, and jute. Its labor force is roughly 10 million suffers from a shortage of trained labor (WB, 2014). Nepal's economic growth adversely affected by the political instability, 4.9 % GDP growth rate in 2010-11 is considered to be the second highest rate in the post-conflict period (Shakya et al., 2012). Domestic income comes mostly from agriculture, construction, financial and other services. The percentage of deprived persons have dropped in recent years. The percentage of people living below the poverty line have halved from 53.1% to 31% in the last

seven years (Price and Bohara, 2013). Nepal has considerable potential of tourism due to spectacular landscape and diverse, colorful culture, but progress in tourism and hospitality industry has been lifted by political instability and inadequate infrastructure. The tourism industry is contributing nearly 3% in domestic income and is the second biggest source of foreign income after remittances (Gautam, 2011). Nepal's exports of mainly leather goods, hemp natural fiber, clothing, carpets, grain and jute goods total \$886 million. Import commodities of mainly gold, petroleum products, machinery and equipment and fertilizer total 2 billion US dollar (Thapa, 2012).

The countries of the SAARC region are at the similar stage of economic development. The share of the manufacturing sector is increasing in major SAARC countries. Each country has a specific role in the specialization of a narrow range of products. The specialization of a narrow range of products will not only source of cost effectiveness, but also improves the quality of their products (Batra, 2006). This dissertation will consider data from major South Asian Association for Regional Cooperation (SAARC) countries, namely, Bangladesh, India, Pakistan, Nepal and Sri Lanka. These countries have a major share in population and economy of the region. Approximately, more than 93% share of the whole region is captured by these countries. These countries in our research is that the data base of these countries is well organized and reliable as compare to other countries. The other countries Afghanistan, Bhutan, and Maldives do not offer a reasonable comparative database (Zaman et al., 2013).

The SAARC region has huge capacity to make itself an effective trading bloc in the world. The share of SAARC countries in world population is about 23 percent and considered a large market with massive profit making opportunities. But the performance of these countries is unsatisfactory if this region compares with the trade performance of other regional alliances. After comparing the SAARC(South Asian Association for Regional Cooperation) and the ASEAN (Association of South East Asian Nations) regions, Batra (2006) comes to know that the performance of the SAARC region is unsatisfactory. By engaging in trade on the basis of product differentiation and economies of scale (due to large scale production), these countries can gain from the trade of differentiated products (Akram and Mahmood, 2012). Economists in the SAARC think that today's South Asia has a variety of common characteristics with the Europe of the nineteen century. Development of international institutions, rising nationalism and regional associations are new sense of political and economic interaction (Anderson, 2000). The mode of trade is also changing due to trade reforms in the world, accordingly, trade of differentiating goods is becoming popular in this region (Weeraratne, 2005).

2.4.2 Child Labor in SAARC

The number of child workers in SARRC countries falls by some 10 million. Notwithstanding this performance, at the end of 2010 there were still over 21.6 million child workers, and half of them were engaged in unsafe work. The overall falling trends could not stop the rising numbers of children in economic activity in South Asia from 2006 to 2012 (ILO-IPEC, 2013). These figures underscore the significance of the remaining threats facing the international community, they also give a clear expression of hope progress against the child labor is attainable, with substantial policy choices and massive commitment in the national and international level. Commitment to national and international level can only be possible due to trade openness in SAARC countries (Deacon et al., 2011).

Approximately 95% of child workers live in developing countries of which half of them belong to Asian developing countries. Within Asia, South Asia has a high frequency of child labor and this differ significantly between countries (Herath and Sharma, 2012). In SAARC countries, the incidence of child labor tends to be higher among boys than girls and in rural areas. A numerous participation rate for boys in the workforce is due to the fact that girls perform their duties in the informal sector (such as at home) which are challenging to capture by statistics (Fallon and Tzannatos, 1998). Urban children generally work on carpet, garment and textile related activities, small business and beedi (hand- rolled cigarette) industry together with their parents, at the same time, children in rural areas tend to work with their families in fishing, agriculture, forestry or small family ventures (Singh, 2005).

Child labor exhibits 12% of the entire labor force in Bangladesh. With the children engaged in economic exercises, 16.12 % are factory workers, 22.90% are seller/vendors, 5.71% are brick/stone breakers, 2.26% is in craft, 4.76% agricultural worker, 0.48% beggars and 0.68% in fish processing (Fassa, 2003). It is examined that practically 60% of child workers are engaged in the agriculture sector and 8% of the workers employed in the hand knotted carpet industry are child labor.

In Nepal, 90% of child workers are engaged in agriculture and household work and the rest of 5% are working in non-agricultural activities, including service and manufacturing industries (Bhukuth and Ballet, 2006). Of those who are engaged in non-agriculture sector, 1.8% is in transportation, communication and construction, 0.8% work as general technical workers and another 0.4% as sales workers and 0.8% work as production workers. There is a higher concentration of child labor in Nepal if compared with other countries in South Asia. Multiple indicator cluster surveys (MICS) and other state surveys found that 34.14% of children between the ages of 5-14 are involved in child labor, if it is compared with the South Asia region as a whole it is 12.04%. There are more young girls than boys are engaged in child labor activities, and the working conditions are worse in rural areas. The agricultural sector occupies 89% of the working children, while 1.38% of employed children are in the manufacturing sector, 0.31% child workers are part of the construction industry, 1.57% are working in wholesale and retail trade, 1.0% are laboring in hotline industry, and 7% are in other industries (UNICEF, 2012).

In Pakistan, 12% in manufacturing, 66% of child labor is found in agriculture and remaining 22% works in service related activities (Rosati and Rossi, 2001). It is evident that in South Asia most working children are found in non-tradable rather than in tradable sector and most of them are employed in agriculture along with their parents. The worse thing is that most of employed children work together with their parents, they are projected to danger in most circumstances. In agriculture sector, most children are exposed to toxic pesticides, while in manufacturing and construction sector they are exposed to deadly material

such as mercury, benzene and asbestos. There is no doubt that these work hazards affect children with physical and mental development during their formative years (Sharma, 2007).

The international community has serious concerns regarding child labor in SAARC region. The situation of child labor is different in different countries (Hidellage and Srivastava, 2012). Male children from the age of 10 to 15 are engaged in militant activities in Afghanistan and Pakistani areas adjacent to Afghanistan (Asal et al., 2008), as the developed states make their front doors less accessible this area for emergence of globalization in the form of increased openness through trade liberalization (Cochran, 2013). Similarly, the other countries like India and Bangladesh male children are taking part in dangerous activities, but, female from 10 to 15 years are pushed into prostitution business (McClarty et al., 2014). Militancy for boys and prostitution for girls both are associated with serious human rights violation in SAARC region (Mehdi, 2010). Trade liberalization is a tool which can bring the children out of the child humiliation in South Asia (Mansoor, 2004).

2.5 Major Developments in Child Labor Reduction

The International Labor Organization (ILO) and the United Nations Children's Fund (UNICEF) have directed their considerations to the reduction of child labor worldwide. As a result of their contributions there have been significant cosmopolitan treaties and conventions, impeding child labor and diagnosing concrete measures for governments to adhere. Once a state sign a convention, UN agencies monitor compliance and hold states answerable for violations.

Table 5: Key Achievements of the ILO to Combat Child Labor

Year	ILO Conventions
1973	The key instrument of minimum age was adopted for admission to employment (15 or the age reached on completion of compulsory schooling) ⁷ .
1989	The UN approved the convention on the rights of the child, determining the right of the child to be protected from hazardous work and economic exploitation, and the refraining of regimes from engaging any person under 15 into the armed forces.
1999	ILO members mutually approved the convention concerning the immediate and prohibition action for the abatement of the worst forms of child labor ⁸ .
2002	The ILO produces its first comprehensive report on child labor and declared 12 June as world day against child labor. This arrangement supports more than 80 countries in formulating their own programs to control the child labor.
2004	First ILO worldwide study of the costs and benefits of eradicating child labor says that benefits would exceed the costs by nearly 6 to 1.
2006	The findings of the second global report on child labor an hope for the future, which says that child labor is declining worldwide, the ILO projected a worldwide campaign to wipe out the worst forms of child labor by 2016.
2009	The 183 ILO Member States mutually accept the Global Jobs Pact as a guide to recovery from the worldwide jobs and economic crunch. The Pact calls for overblown surveillance to attain the eradication and prevention of an increase of child labor, forced labor and discrimination at work.
2010	The ILO, published its third Global Report on Child Labor, warning that the pace and profile of development is not expeditious to attain the 2016 targets of eradicating the worst forms of extensive child labor.

Source: International Labor Organization Conventions (Various Years)

⁷ Convention No. 138

⁸ Convention No. 182

The South and Southeast Asian countries are devoted to defeat the child labor, and this apart from different actions taken by individual countries of the SAARC and ASEAN. The ILO members in the SAARC and ASEAN region have ratified some of the ILO legal frameworks and the other conventions for eliminating child labor.

Cambodian state had ratified the convention 138 on the minimum age in 1999, and specified the minimum age to work to be the age of 14. Cambodian government signed the convention 182 on the worst forms of child labor in 2006, it is a remarkable step towards the elimination of the worst forms of child labor in Cambodia. In spite of this, a significant number of children to be found in Cambodian labor market (Kim,2011).

Indonesian government established a national action committee on the establishment of the worst form of child labor in 2001, and prepared a national action plan for the reduction of the worst forms of child through a three phase program over the next twenty years. The plan constitutes a time bound program to eradicate the worst forms of child labor. The legal committee of the national action plan has since taken steps to the legal framework to stop the child labor and child exploitation (Bessell, 2011).

Thailand has made considerable efforts to eliminate the worst forms of child labor, therefore, set a minimum age for domestic work at 15 years. For this purpose government passed the parliamentary regulations for domestic workers, which offers protections for domestic child workers, also updated the list of dangerous work (Hepburn and Simon, 2013) and working circumstances banned to teenagers and change the minimum age in sea vessels from 15 to 17 (Ratner et al., 2014).

The Philippine has made a considerable progress in the efforts of child labor reduction in 2014. The national child labor committee (NCLC) is one of the milestones to eradicate the child labor in the Philippines. NCLC is an operational framework to finalize the strategic plan against child labor. Moreover, Philippine also rectified the all international conventions on child labor, i.e. ILO C. 182, Worst Forms of Child Labor, UN CRC Optional Protocol on Armed Conflict, Palermo Protocol on Trafficking in Persons, ILO C. 138, Minimum Age and UN CRC Optional Protocol on the Sale of Children, Child Prostitution and Child Pornography (US Department of Labor, 2015)

In 2013, Sri Lanka made feasible efforts to eradicate the child labor in 2013. National Steering Committee (NSC) reactivated to eliminate the child labor in all 25 districts, 755 Child Protection Committees are activated all over the Sri Lanka, and the main motive of this project is to make Sri Lanka, a child labor free country by 2016. Except Palermo Protocol on Trafficking in Persons, Sri Lanka ratified the all well-known conventions to eliminate the child labor (US Department of Labor, 2015)

Countries	ILO- Convention	ILO-Convention	ILO-Convention	ILO-Convention
Countries	No 138 ⁹	No 182 ¹⁰	No 29 ¹¹	No 105 ¹²
Afghanistan	\checkmark	\checkmark	×	\checkmark
Bangladesh	×	\checkmark	\checkmark	\checkmark
Bhutan	×	×	×	×
Brunei	\checkmark	\checkmark	×	×
Cambodia	\checkmark	\checkmark	\checkmark	\checkmark
India	×	×	\checkmark	\checkmark
Indonesia	\checkmark	\checkmark	\checkmark	\checkmark
Laos	\checkmark	\checkmark	\checkmark	×
Malaysia	\checkmark	\checkmark	\checkmark	\checkmark
Maldives	\checkmark	\checkmark	\checkmark	\checkmark
Myanmar	×	\checkmark	\checkmark	×
Nepal	\checkmark	\checkmark	\checkmark	×
Pakistan	\checkmark	\checkmark	\checkmark	×
Philippines	\checkmark	\checkmark	\checkmark	\checkmark
Singapore	\checkmark	\checkmark	\checkmark	\checkmark
Sri Lanka	\checkmark	\checkmark	\checkmark	\checkmark
Thailand	\checkmark	\checkmark	\checkmark	\checkmark
Vietnam	\checkmark	\checkmark	\checkmark	×

Table 6: ILO Convention Ratification Status of Member Countries

Source: Various International Labor Organization Conventions

⁹ ILO, 1973 Minimum Age Convention No. 138

¹⁰ ILO, 1999 Convention on the Worst Forms of Child Labor No. 182

¹¹ ILO, 1930 Forced Labor Convention No. 29

¹² ILO, 1957 Abolition of Forced Labor Convention No. 105

In accordance with the Minimum Age Act, the age of working children differ between countries. For example, in Bangladesh the minimum age for work is 12 years, while in India and Pakistan, the minimum working age of 14 is practiced only to certain distinct activities occupations and processes (Chaturvedi, 1995). In Pakistan, the Employment of Children Act 1991 ban the hiring of children under age of 15 in hazardous or dangerous activities. Strategies are also installed for enforcement and execution of the act, such as the requirement that all entrepreneurs maintain a register of details regarding the recruitment of any children under the age of 15 (Sharma, 2007). In Nepal, the minimum working age of 14 does not apply to any exercises, such as brick kilns and plantation (Baker et al., 1997).

CHAPTER 3

LITERATURE REVIEW

3.1 Chapter Overview

Chapter 3 explains the literature review relevant to the child labor. It identifies, appraises, and synthesizes the research evidence and arguments. Moreover, this chapter incorporates the current knowledge in the field of both trade and child labor, including substantive findings, as well as theoretical, methodological and empirical contributions to situate the current research within the body of literature.

3.2 Introduction

A voluminous literature focuses on the new aspect of international trade and emphasizes the effects of market size on the international division of labor. Edmonds and Pavcnik (2005) claim that vast literature tended to focus on the determinants of child labor and explore the connection between international trade and the child labor. On this note, this section discusses the existing studies with a critical perspective, as well as provide a staccato review of earlier work reveals the different aspects of trade and child labor.

3.3 Theoretical Studies

In the theoretical literature on child labor the notable contributors are Basu and Van (1998), Basu et al. (2010), Ranjan (2001), Baland and Robinson (2000), Jafarey and Lahiri (2002) and Dessy and Pallage (2005). Trade liberalization, in a developing economy, is generous in inefficient and disorganized labor, likely to increase the relative rate of return to unskilled labor, thus diminishing the satisfaction to spend on skills and education. Therefore, the rewards to child labor raise with a substitution effect along with an increased supply of child labor (Doepke and Zilibotti, 2010). Trade openness does not increase the demand for child workers if children generally work in sectors that compete with imports or in the nontradable segment. On the other hand, some economists suggest that the effects of trade openness would be negligible (Doepke and Zilibotti, 2009).

Trade liberalization proponents advocate that economies have a desire to invest in skills and education in order to spur economic development and their long-run competitiveness (Tariq and Ab-Rahim, 2016). Ab-Rahim and Tariq (2016) examine that the trade liberalization might also lead to structural changes and sectoral shift from low skilled to higher skilled capital intensive manufacturing and services in the long run, thus making the recruitment of children less convenient.

A greater intensity of child labor could abate costs to gain the country a competitive advantage over others. Therefore, all economies face this incentive, increased trade intensity
could bring about a magnified incidence of child labor all over the world. Developing economies with tolerant labor standards, nominal wages and a plenteous supply of unskilled workers, including child laborers, are regarded as a haven for foreign investors, a perspective called the conventional wisdom by Doepke and Zilibotti (2010).

Trade liberalization in a developing country, will not only have a child labor substitution, but also a child labor income effect. Despite the fact that free trade is far from an accomplished fact and delicately assume that trade does increase the growth rate of the traditional economy (see Estevez, 2010) as well as, it enhances the relative rate of returns of incompetent workers. This child labor income effect can be expected to decline the benefits for parents with little skills to fire their children to work by assuming that child education and child leisure are normal goods (Basu and Van, 1998).

Basu et al. (2010) find out that parents in poor countries use extra income to move children out of work. This effect of additional income occurs even in the face of increased earnings opportunities for children. Along these lines, the patronizing view held by so many wealthy country researchers, that poor country parents somehow do not bother about the betterment of their children seems unfounded. These directions propose that the use of punitive income and sanctions against poor countries, where child labor is common, may not have the proposed impact. When children work as a result of poverty, then punishments and sanctions for the economies with child labor hardly seem to be favorable. A more impressive approach to

restrict the child labor would be to supplement treatment for income generating exercises with, possible, additional motivation for children staying in school.

The fluctuations and variations in earning channels and opportunities in developing economies are a cause of an increase in the demand for child workers. Labor markets arises from trade liberalization, some time, increase the child labor. Incidentally, children may be compelled to perform several activities customarily done by adults (Rees et al., 2012). Markusen et al. (1995) discover the association between trade liberalization and earnings tendencies, their arguments support the entry of foreign multinationals and earnings potentials of local workers. This occurs when the foreign companies apply market influence in local markets. However, Cigno et al. (2002) state that the trade liberalization not only affect the earnings opportunities of developing countries, but also increases the rich countries' capacity to control the policy in the poorer countries. As developing economies incorporate into the world's economy and progressively depend upon world export markets to sell their goods and services, developed and wealthy economies can use the trade sanctions as a tool to force the policies that attempt to reduce the child labor (Cigno et al., 2002). Ray (2000) uses a strategic framework to examine the effect of stricter child labor policy on research and development when consuming labor-intensive. Doepke and Zilibotti (2005) shows that when countries set child labor regulations in a non-cooperative setting, a country benefits from a stricter policy abroad when it has high product standards at home. In a subsequent study, Rogers and Swinnerton (2008) analyzes the implications of non-cooperative child labor policy for local production externalities and concludes that the impact of international trade on the child labor is ambiguous.

Similarly, Webbink et al. (2012) uses a strategic framework to examine the effect of stricter child labor policy on research and development and obtain a broad overview of the degree to which children in different regions of Asia and Africa are involved in different forms of child labor. Edmonds and Pavcnik (2006) present a trade and child labor model, the study justifies the Ben-Porath (1980) in that he focuses on the investigation of the effects of trade policy on child labor. In a subsequent study, Edmonds (2007) analyzes the implications of child labor policy for local production externalities and concludes that impact international trade on the child labor is not clear. The study extends previous investigations that examine the link between child labor policy and modern international trade.

Krueger (1996) has indicated that in Heckser-Olin framework, it is assumed that, developing countries are abundant in unskilled labor therefore these countries have comparative advantages in unskilled labor. Sign and Zammit (2004) find out that trade liberalization may not only provide the comparative advantages in unskilled labor intensive sectors for the developing economies, but also brings a comparative erosion of labor standards and adaptation of less labor intensive technology with high labor intensive techniques known as race to bottom hypothesis. The long term capital transfers from developed economies to developing economies plays an important role in comparative advantage. It is noteworthy that weak labor laws, low wage rates, lax labor standards and abundant supply of unskilled labor, including child workers are viewed as a haven for foreign investor. A country can gain comparative advantage on labor intensive goods by using a higher extent of child labor through cutting labor costs. Growth with the increased trade openness and penetration of FDI increases the demand of child labor and their

earnings (Edmonds and Pavcnik, 2006). As the wage rate of children engaged in the economic activity increases, the cost of opportunity for children to go to the school also increases. This relief supports the arguments of the parents about sending their children to work (Ranjan, 2001).

The new trade theory postulate that trade of differentiated products tends to be prevalent between countries that are similar in their factor endowments, technology level, skill levels, and so on. Thus, the trade of differentiated products will be the dominant trade pattern especially between countries at a similar level of economic development. It has been established that gains from trade will be large when economies of scale are strong and trade goods are highly differentiated (Krugman, 1980). Product differentiation and economies of scale that has internal to a firm thus formed the basis of the trade of differentiated products.

Krugman (1991) uses a two-region model to examine sector vise competition in labor demand in an imperfectly competitive market with increasing returns to scale. Krugman's theoretical model shows that competition in labor demand may result in driving the labor out of the market when the cost of labor is high enough. Notten et al. (2012) uses two regions to examine different levels of competition in labor behavior in an imperfectly competitive market. Their theoretical work shows that competition in poor countries may result in either driving the adult labor out of the market when availability of child labor high. Role of unskilled labor in trade is also very important and such type of framework based on new trade theory is advanced by Egger et al. (2007). The authors build a two country model of skilled and unskilled labor, which assumes monopolistic competition. The framework examines the effects of trade in intermediate products on unskilled labor, output level and welfare under the conditions of the endogenous tax and skilled to unskilled labor functions. The framework provides interesting insights into the effects of trade in intermediate goods on the foreign unskilled labor level.

In the investigations of the "race to the bottom" in child labor protection laws, the evidence to support the hypothesis remains mixed. The "race to the bottom" concept, an offshoot of the inverted u-shaped child labor curve hypothesis, suggests that developing nations compete in lowering the stringency of child labor regulations in efforts to attract foreign investments from more developed nations. Developed countries are assumed to relocate the labor-intensive industries to avoid stricter labor standards at home (Bigelow, 1997). Singh and Zammit (2004) conclude that a "race to the bottom" in child labor policy is unlikely for both small and large countries, when standard intervention applies. However, the paper suggests that the relaxation of the assumption of immobile factors implies the possibility that freer trade induces capital flight from more developed countries with stricter child labor regulations to less developed countries with weaker regulations.

Singh and Zammit (2004) asserts that studies which provides empirical evidence to support the inverted u-shaped child labor curve hypothesis lends some credence to the possibility of the race to the bottom in child labor policies among the developed nations. It is the primary argument of many anti-trade liberalization and workers' rights that trade liberalization results in a race-to-the-bottom¹³. These activists are supportive of the belief that free trade generates additional encouragement for competitive advantage, resulting in increased child labor to get the lowest labor rate possible. In the motivation for low cost labor, developing countries with weak labor regulations and abundant unskilled labor pools including children, are prime targets for corporations competing in the global economy (Singh and Zammit, 2004).

To date, Singh and Zammit (2004) is a unique study that proposes a general equilibrium analysis based on traditional trade and their study finds little support for the race to bottom arguments. There is some evidence that more openness increases the level of income, compliances with minimum wages, and reduces child labor. The authors developed a framework consistent with the Heckscher Ohlin model of international trade, but there is no model or framework for monopolistic completion to examine the effect of domestic child labor regulation.

Capital and labor market imperfections strengthen the underdevelopment trap of developing countries (Alan, 2011). Kaplan (2012) and Chamley (2012) examines the multiple

¹³ Baldwin & Krugman (2004) and Garretsen & Peeters (2007) suggest that a race to the bottom is partially materialized in new the trade theory. Developing countries have to use second hand or outdated technology from developed countries; this technology is less labor intensive but can be utilized product differentiation with the help of skilled labor.

equilibriums in the labor markets in poor developing countries due to interest rates. Van Binsbergen et al. (2012) argue that lower interest rates, offer better access to credit due to trade liberalization, may increase the employment and earnings opportunities available to poor households of a developing nation. But, Kanungo (2012) has a diverse opinion on this issue; he claims that trade liberalization attracts the foreign investor in developing countries and increases in the demand for local labor. Child labor and adult labor are substitutes (substitution axiom) thus low cost child labor is regarded as an utopia for foreign investors. Willis (1973) says "labor demand, expands the earnings opportunities of children" (Serra, 2009). Child leisure and child education are normal goods, but child labor is a luxury commodity to the poor households (luxury axiom); old age security of the parents behind the incidence of high backward society is a source of high fertility rate and lower investment in the education (Basu et al., 2010). Ventevogel et al. (2013) has found a justification in the need for old age security of the parents behind the tendency of the severe fertility rate and petty investment in the education of their children (and hence the high incidence of child labor) in a poverty-stricken community where the child labor rate is considerably high. Thus, it is recommended an improvement in legislation of compulsory education and health care services to eliminate child labor from the system.

Trade openness could thus be linked with a decreased incidence of child labor. Another probably positive effect of trade openness on the frequency of child labor through the effect of openness on interest rates and credit constraints. More open economies are likely to have lower interest rates and offer easy access to credit. This diminishes the opportunity cost of schooling and thereby the intensity of child labor (Guarcello et al., 2010). From a political perspective, an open economy has less encouragement to secure the prominent traditional culture and institutional framework that endorses the child labor (Neumayer and De Soysa, 2005). Trade liberalization might be delaying the decision duration of the factors that regulate the child labor supply. Cultural and societal standards are more glamorous in developing economies, leading to a higher social acceptability of child labor (Neumayer and De Soysa, 2005). But the awareness level of parents naturally converges to natural norms, the social acceptability of child labor incidence reduces due to the trade liberalization. Subsequently, they may prefer sending their children to an educational institute instead of engaging them in work. Trade liberalization has an effect on the labor market adjustment and ethic in developing countries.

The financier and capital owner made their decision according to world market trends and have a low level of reliance on the local supply of inexperienced and immature labor, including children (Serra, 2009). In economies where the trade intensities are very low, efficient and skilled workers and the owners of capital naturally benefit from preserving the institutional culture. While, poor legislative conditions encourage the supply of low-priced unskilled labor, including that of children (Neumayer and De Sousa, 2005). If child labor is officially banned, but continues to exist due to lack of enforcement, then as Berik and Rodgers (2010) has noted violations of labor standards are more economical in the non-tradable and less exports oriented sectors. Berik and Rodgers (2010) had also found that the prevalence of child labor declines sharply with national income. According to them, the use of child labor is negatively related with the economic development. In addition, the governments of wealthier countries seem to have more strictly controlled labor standards and better working conditions. Regulatory authorities of developing countries often have a scarcity of resources to implement the child labor bans.

Foreign venture capitalist and investors might be less involved in exploiting low-cost labor, including child labor. Market stability, market size and market expansion, economic and political stability, infrastructure and high labor expertise are also essential in determining the child labor (Hagestad and Uhlenberg, 2006). Most of the empirical studies typically fail to interrogate that, economies with lax labor standards in general and a gigantic incidence of child labor in particular attract a greater inflow of FDI (Brown et al., 2012). In as much as foreign direct investment spurs economic development (Herath, 2012), it also has a catchy effect for reducing the incidence of child labor laws as they are apparently under great interrogation of executors and doubtlessly more exposed to the authorities of trade unions, consumer, human rights, the media and other opposition groups (Davies and Voy, 2009).

Banerjee and Nag (2013) collaborate the child labor and trade liberalization in two basic ways. First, trade liberalization may raise the earnings and employment potentials to poor families in developing economies. Variations in domestic labor markets from trade liberalization may decline or incline the child labor. Second, trade liberalization enhances the control of wealthy countries in the local policies of the developing world. Trade liberalization can increase jobs and income in poor countries because of penetration of international investment or increases in the value of developing countries' export products.

It is observed that the governments of developing countries are coerced by the developed world about child labor policies. Pressure of wealthy nations can be effective in dropping the child labor when developing countries infiltrate into the global economy (Edmonds and Turk, 2002). Governments can reduce the child labor by giving grants to education facilitators and encouraging the children to go to school instead of work. For this purpose, compulsory schooling can be used as an instrument for regulatory authorities to stop children from working (Lopez-Calva, 2002).

When a country opens its borders for getting the access to the international markets, the foreign investment often, but not always, come into the country. This is the cause of an increase in the demand for local labor and hence increase the real wage rate. In addition, nowadays some developing countries have a comparative advantage in agriculture goods, and integration into international markets may increase the price of the export product to international levels. Thus, trade liberalization may increase employment and wages in these sectors (Davies and Voy, 2009). Trade liberalization can change the decisions of the factors that determine the child labor supply. Social mobility and cultural influence, are more influential and traditions in developing countries, accelerate the higher acceptability of child labor (Lopez-Calva, 2002).

Ray (2000) for Pakistan, Weeraratne (2005) for Sri Lanka, Kanungo (2012) for Bangladesh, Rees et al. (2012) for India, Dorji and Gyeltshen (2013) for Bhutan, consequently, Herath (2012), Singh and Zammit (2004), and Kaplan (2012) find that trade liberalization is one of the most common policy prescriptions offered to initiate the process of poverty eradication in today's poor countries. They agree on a single point that growth is likely the most important channel through which trade might lower poverty in the long run. These studies have emphasized the importance of short to medium run consequences of trade reforms. This focus is in part motivated by restricting mobility of individuals across regions and industries within poverty-stricken economies as a result of massive trade policy shocks differential impact of trade on poverty across regions and industries within poor economies that persist over a long period after the reforms. A high level of child labor coupled with low school attendance rates and income levels. For example, in India and Pakistan, 7-10 percent children of age 10-14 report working as their usual principal activity and 98 percent of these working children do not attend school at the beginning (Boockmann, 2010).

Trade reforms, by differently altering the local economic environment in which families operate and the resources available to families, might affect the parents' decision to invest in children. When the awareness level of parents converges to global norms (via trade liberalization), the general acceptability of child labor decreases due to social proliferation. As a result, these changes, parents start sending their children to school instead of work. Trade liberalization also affects the labor market regulations and standards in developing countries. (Edmonds, 2010).

Edmonds (2010) and Ray and Chatterjee (2012) emphasis on the effects of child labor and agricultural trade liberalization in a framework where adolescent frequently work on a family farm. This is not astonishing given that highest level of child labor worldwide rise in family farms or non-traded sectors. However, some advocates that trade liberalizations in agricultural sector (particularly in economies where a huge population depends on selfemployed agriculture) might yield specific case studies because income effects linked with selfemployed agriculture sector might have a tremendously meaningful role that would not be as pronounced during a trade reform. In addition, Dorji and Gyeltshen (2013) propose that the child work for compensation, rather than work on their clan and family farm, is the kind of job that interrupts most with schooling and thus it is a matter of greater policy interest. These are certainly disputed claims, the trade liberalization permit us to check out whether and how an economy-wide trade behaviors transfer in a country with relatively developed non-agricultural sector, and thus a tremendous tendency for children to engage themselves for wages in nonagricultural sector, affects child labor and schooling preferences.

The effect of trade, on hiring the children as labor in poor, especially agricultural economies have started an unmelodious debate. Research which reflects the analysis of actual data from developing countries and mathematical modeling both come to the general conclusion that poverty is the root of all evil, not capitalism or trade liberalization (Edmonds, 2010). Poverty limits access to education which has been shown to decrease child labor and improve the economic wellbeing of families. Through improved opportunities for education, greater integration into a global economy and trade liberalization has empirically been shown to drop

child labor volumes. The remedy of child labor is not very simple, but it is something that will take continued development and dedication to free trade and to trade liberalization overcome. Just as developed nations overcame their legacy of child labor through greater economic expansion, higher wages, and increased education opportunities, the developing world will be able to do so in the future as well (Edmonds and Schady, 2009).

If poverty is a major cause of child labor and children have to work either for the income or they cannot afford school fees, than the trade improvements in earnings opportunities can reduce child labor. When one child makes more in a poor family as a result of trade promotion, than other children of the same family may be able to stop working and they may be in a better position to attend school. Additionally, the increase in the parental earnings may support the family to reduce the work of school going children. Parents can get the substitutes for those goods which are produced by children previously, or they can utilize their increased income to alternative income earned by their children previously. Hence, trade liberalization is a best tool to help the parents in poor countries and stop the child labor without foreign intervention (Islam and Choe, 2013).

In the conventional literature, the supply of child labor has been attributed to factors such as failure of the educational system, the objectives of households to maximize present income, a dualistic economy characteristic of developing countries with the coexistence of formal and informal sectors. However, it is without any ambiguity that the root cause is helpless and abject poverty, which enforce the people to have a large size of households and adolescent to go out in the labor market and earn their own means of livelihood. To eradicate the incidence of child labor, World Development Reports called for a multifaceted approach with programs that increase income security, reduce costs of education, and improve the quality schooling (Basu et al., 2010).

Emerson and Souza (2011) examine that labor market interventions raise adults' wages are expected to mitigate the problem of child labor. Ranjan (2001), Baland and Robinson (2000) and Jafarey and Lahiri (2002) are some unique studies which focus on capital market failure and emphasize the importance of capital-market imperfection as a contributing factor to inefficient child labor. On the other hand, Dessy and Pallage (2005) has advocated in favor of the imposition of compulsory education as a means to combat the incidence of child labor. Dessy and Pallage (2005) has shown that in an economy where the benefits of having children are outweighed by rearing costs, a policy of free education with no compulsory education laws, may lead the economy to an underdevelopment trap with a high fertility rate and higher incidence of child labor. On the other hand, an obligatory educational policy is expected to abate the presence of the child labor evil from the system.

The supply side mainly results from phenomenon such as poverty, education, household, and government decisions. The income levels of countries play a very critical role and has a strong link between the incidence of child labor and household income (Fallon and Tzannatos, 1998; Udry, 2006). Most studies noticed that the poverty of the household is an important factor in determining child labor (ILO, 1992; Grootaert and Kanbur, 1995). Basu and Van (1998)

avouch that poverty expedite the parents send their children to work since they do not see any substitutes. It is usually acknowledged that when the society is characterized by poverty and inequality, then the, incidence of child labor tends to increase (Bekele and Boyden, 1988; Grootaert and Kanbur, 1995). Krueger (1996) has found that the acuteness of child labor fell sharply with national income. He concluded that the child labor is negatively related to economic development. Moreover, the regulatory authorities of rich economies suggest to have stricter labor standards and better working conditions. The regulators of developing economies often absence of resources to enforce child labor laws. The anti- trade liberalization elements argue that market integration, augmenting the labor demand, enhance the earnings opportunities of young workers and thereby necessarily leads to more child labor. Some opponents of trade liberalization recommend the practice of trade sanctions against the exporting countries as a way to eliminate the child labor (Kis-Katos and Sparrow, 2011).

One reason frequently cited in inconclusive evidence in the study of trade child labor linkages is the lack of theoretical underpinnings to ground empirical prediction of the trade and child labor relationships (Edmonds and Pavcnik, 2006). More recently, formal theoretical frameworks have been advanced to provide a basis for empirical hypothesis. In particular, the theoretical framework developed by de Bell et al. (2007), based on the Hecksher Olin type trade model, provides an explicit description of the child labor impact of inter industry trade. An important contribution of the de Bell et al. (2007) framework is the formal presentation of the child labor impact of trade due to income effects or trade induced child labor technique due to differences between the countries. Although widely recognized, the income effect (trade induced technique effect) postulated by Nardinelli (1980) has been investigated in earlier studies.



Figure 5: The Mechanism of Child Labor

Source: Acaroglu and Dagdemir (2010)

Recent development in modeling trade-child labor relationships contributes to providing theoretical underpinnings for empirical tests, most analyses are based on the traditional theory of trade. The Ranjan (2003) framework is based on the traditional framework of the trade,

which assumes constant returns to scale and perfect competition. In contrast, theoretical frameworks that examine the impact of trade on the child labor using modern trade theory are relatively scarce in the literature. There is yet a formal framework to delineate the impact of trade of differentiated products on child labor and to examine how structural parameters in a new trade model may clearly affect child labor. Futhermore, literature shows that most studies that are based on new trade focus on the issue of child labor policy rather than describing the effects of international trade on domestic child labor (Cigno et al., 2002). In particular, strategic behavior pertaining to the foreign capital inflow affects child labor regulation of domestic child labor is most typically examined (Fors, 2012).

The literature review shows the various causes of the child labor along with its economic reasons. Moreover, this review inquires that how child labor in developing continues to be affected by economic growth and trade liberalization, once it exits in developing economies due to trade liberalization as seen in Figure 5. According to Figure 5 there are two dimensions which constitute a child labor mechanism for demand and supply side. Firstly, It is made up by the factors that determine child labor supply and child labor is defined by household decisions, government decisions, education (illiteracy) and poverty. Secondly, it is made up of those factors that affects the demand side of child labor and related to the growth dynamics of economies that integrate with the global economy

3.4 Empirical Studies

The empirical literature on demand side factors of child labor is relatively scarce as compared to supply side factors i.e. household factors. Unfortunately, it is difficult to obtain firm level data on child labor, which can be beneficial to validate the results of this study and even the survey's reliability can be uncertain due to resistance by firms to disclose the information on the child labor and wages paid to these unlawful workers. Davis and Voy (2009) have empirically found a feeble connection between trade liberalization and child labor, by using instrumental variable techniques on macro level data. One of the drawbacks of macrolevel data is an absence of data for the firms which are engaged in the production of differentiated goods by using child labor. The unavailability of reliable data on child labor issue necessitates a theoretical approach. At present, empirical examination has illustrated various findings on whether trade improves the economic stability of the country given the adverse effects of child labor (Edmonds and Pavcnik, 2005). The authors provide evidence that child labor increase in the initial phase of economic growth, but decrease in the later phase of development. However, the interesting findings of Kambhampati and Rajan (2006) give impetus to further and expansive research in the area, new data and more sophisticated statistical methods in subsequent studies do not provide conclusive findings as to the validity of the inverted u-shape curve (see Basu et al., 2010).

Subsequent investigations into other trade-child labor phenomena such as the race to the bottom hypotheses face similar mixed findings. In contradiction to the child labor, Dustmann et

al. (2010) find that trade and investment data suggest labor-intensive industries relocate to other industrial countries instead of to less developed nations (LDCs). Jacobs and Graham (2010) found that there is no direct data on turnover among school cafeteria young workers, concludes that other factors such as labor standards for workers, labor training, infrastructure and political stability, as opposed to cost-savings from labor regulations, play more important roles in the relocation decisions of food processing units. However, the empirical study by Singh and Zammit (2004) identifies a high cognitive performance in 26 developing countries to support the race to bottom hypothesis. Since the empirical study by Galor (2012) is based on the predictions of a formal theoretical framework, the study provides more persuasive evidence to support the validity of the substitution effect by short-run cost-benefit analysis.

Davies and Voy (2009) express child labor issue in two important ways: First, this study mainly addresses the trade openness, and also looks at penetration by foreign direct investment (FDI) defined as the stock of FDI over gross domestic product (GDP). Second, their research uses the labor force participation rate of 10 to 14 year old adolescent as the dependent variable in their main analysis, Clark (2011) examines the number of economic sectors in developing economies, in which expression of a child labor can be traced by using the primary school and the secondary school non-attendance rates.

Estevez (2011) finds that the comparative static results show that the policies have a positive impact on the child labor reduction and increase the welfare of unskilled labor another contribution of his work. The results show that an increase in FDI increase the welfare of the

economy and it has a positive impact on the return to the education and lead to a reduction in the child labor. It will have a positive impact on the improvement of human capital stock in the future and sustainable development this work is similar to Dinopoulos and Zhao (2007).

Bonnal (2010) employs a panel data approach to test the link between child labor and international trade. The results indicate that the economies which trade more and have a higher stock of FDI have less child labor. Iram and Fatima (2008) use multivariable vector auto regression (VAR) model for investigating the causal links between FDI, openness to trade and the child labor. They find that openness to trade raises the exportable sector and increases the demand for child labor in Basu (2002) framework due to the substitution effect while in Edmonds and Pavcnik (2004) setting this relationship is negative due to income effect. On the other hand, FDI is found to lower the incidence of the child labor. Kanungo (2012) interrogate the impact of FDI on child labor by using instrumental variable techniques. They show that FDI has a negative impact on child labor, however, they identify, this is biased by the endogeneity of FDI, and the impact of FDI is conveyed through its impact on per capita income.

Ray (2000) estimates the impact of household income on the schooling and child labor in Pakistan and Peru. Estimations show that child labor is negatively correlated with the household welfare. The interesting finding of this work is that in the case of more educated mother there is a less chance of child labor in Pakistan and on the other hand, in case of Peru there is no significant effect of female education on child labor.

Source, Year	DATA: Country, Year, Source, Observations.	Definition of Child Labor	Econometric Analysis Technique	Independent Variables	Findings
Doytch et al., 2013	100 Countries Data from 1999 to 2010 is drawn from DHS, MICS and other National Surveys.	Economically Active Children, Ages 10-14	The Blundell- Bond system GMM methodology	GDP per capita FDI/GDP Control of Corruption Population Density	Income and substitution effects of FDI vary across the sectors and regions.
Estevez, 2011	118 Countries Data from 1985 to 2007 is Drawn from World Bank and ILO in the Broadest Estimation.	Economically Active Children, Ages 7-14	Linear Least Squares Method	 GDP per Capita. Trade Volume as a Percentage of GDP. Value-Added as a Share of GDP. Pupil-Teacher Ratio in Secondary Education. 	Child labor is negatively correlated with FDI inflow, but the results are not conclusive due to limitations in the data.
Davies and Voy, 2009	 145 countries data for the year of 1995. The political freedom index comes from the Freedom House (2006) Political rights index. From the CIA World Factbook (2006). 	Economically Active Children, Ages 10-14	Fixed Effects to the Pooled Data.	FDI Income Trade Openness Freedom Index Ratify (ILO) Convention 138) Freedom Index	FDI has no effect on child labor

Table 7: Empirical Evidence of Child Labor

Neumayer and De Soysa, 2005	WDI, 2001 UNICEF, 2003 UNCTAD, 2003 127 countries down to 33 at maximum.	Economically Active Children, Ages 10-14.	Ordinary Least Squares (OLS)	 GDP Per Capita. Trade Openness. Expenditure on Health %age of GDP. FDI Relative to GDP. Education Expenditure Share of GDP. 	Child labor is negatively correlated with FDI and positively associated with trade openness.
Blunch and Verner, 2000	Ghana, 1997; CWIQ, 14514 House Hold Survey 60,686 Individuals ranging from 5 to 14 years	9 to 14 Years Economically Active Labor and Not Going to School Past Four Weeks. Household Chores are Not Included in This Definition.	Uni-Variate Probilt Model main focus is on the work/no work decisions	HH Soicoeconomic Status Individual Characteristics i.e, Urban or Rural. Parental Characteristics Community Characteristics.	Positive association b/w poverty and child labor
Rossi and Rosati, 2001	Statical Information and Monitoring Program for the Child Labor (1996) 10,453 HH 30722 children engaged (from 5 to 14 years) in CL in Pakistan.	Ages 5 to 14 years	Simultaneous Equation model which combines a Tobit and Probit Model with correlated disturbances.	Individual Characteristics Parents Characteristics Sibling Number	Household income has a positive effect on school enrollment in Pakistan.

Table 7: Empirical Evidence of Child Labor (continued)

Ray, 2000	Peru Living Standards Measurement Survey 1994 (PLSS), 3623 HH, 5231 Children aged 6- 17. Integrated HH Survey 1991, 4800 HH, 5867 observations on children 10 -17 yrs.	Age: 6-17 (Peru) 10-17 (Pakistan) Child Labor Measured as Time Spent in Wage Labor and HH or Non-wage Labor.	Logit and Probit Model	HH Socioeconomic Status (Poverty) Adult (Female Education) Individual Characteristics Child's Gender/Age Community Characteristics	Expenditure on education has a positive impact on school enrollment and negative impact on child labor.
Deb and Rosati, 2001	 Ghana Core Welfare Indicators Questionnaire, 1997. 14514 HH. 13484 Children, 6701 Nuclear HH With At least One Parent Present. Human Development of India Survey, 1994, 34398 HH, 34211. Children, 16371 nuclear HH with at least one parent present. 	Ages 6 to 15 Years	Multinomial Logit Model with HH Level Random Intercept Latent	Community Schooling Iniviual Characteristics Household Characteristics Education of Parents Socioeconomic Status.	Children who belong to poor families less likely to attend school.

Table 7: Empirical Evidence of Child Labor (continued)

Source: Various Studies

Blunch and Verner (2000) find that self-employed workers are likely to engage in harmful child labor and could not be able to attend school. Estimation shows that girls have to work more than boys. Finally, there is a small but positive relationship between poverty and child labor. Deb and Rosati (2001) examine the lowest income quintile of India and Ghana by applying multinomial logit analysis. Their analysis shows that in both countries' children who belong to poor families are less likely to go to school and contrast; there is a significant increase in the household income decrease the incidence of child labor. Rossi and Rosati (2001) find that household income has a positive effect on school enrollment in Pakistan. Hence, these two studies are very similar to each other and find the same evidence in different regions.

Empirical findings such as Neumayer and De Soysa (2005) reveal that child labor is negatively correlated with FDI and positively associated with trade liberalization. Trade liberalization might be delaying the decision duration of the factors that regulate the child labor supply. Cultural and societal standards are more glamorous in developing economies, leading to a higher social acceptability of child labor. In economies where the trade intensities are very low, efficient and skilled workers and the owners of capital naturally benefit from preserving the institutional culture and legislative conditions encouraging a large supply of low-priced unskilled labor, including that of children. Trade liberalization can affect the sources of economic growth, which creates the demand for child labor. The effect of growth on child labor demand is started, in the developing countries, with global dynamics of competitive advantage. Amestoy and Crosbie (2000) use the predictions of the formal model to derive estimates equations for their empirical study in a single sector, and find evidence to support the claim that trade is not good for the child labor. More recently, Kis-Katos and Sparrow (2011) present both theory and empirical evidence to disclose the child labor effect. The authors find evidence to support the hypothesis that industries that have the most increase in the child labor sanctions the largest decrease in the net decrease the incidence of child labor. Empirical findings based on the theoretical predictions found in the studies such as Neumayer and De Soysa (2005) and Goldberg and Pavcnik (2004) are more likely to provide persuasive evidence to validate trade child labor linkages.

A recent empirical study by Chaudhuri (2011) offers an investigation into the relationship between trade patterns and child labor regulations under the traditional trade theory. The authors find that under Hechsher Oilin Vanek (1968) framework, there is statically significant evidence to support an empirical relationship between child labor regulations and trade patterns. On the other hand, Baldwin and Forslid (2010) under a monopolistic competition framework, find statically significant evidence to support a relationship between labor standards and trade of differentiated products. One shortcoming of the analysis is that the authors do not offer a formal framework to describe the relationship between child labor and trade variables. In short, the study does not have vertical underpinnings in the child labor context.

3.5 A Concluding Remark

The review of theoretical and empirical studies suggests that the trade affects the child labor depending on types of trade induced effects. Trade-induced child labor selection and scale effects deserve substantial theoretical and empirical attention due to a distinguishing characteristic of the new trade theory. Based on the above-discussed studies, the traditional trade framework shows that international trade affects the scale of economy and scale of production and it has an impact on child labor. New trade theory talks about the product differentiation and increasing returns to scale; the beauty of Krugman's framework (1979) is that it explains the effect of trade on a number of firms that are engaged in the production of differentiated products in the home and foreign countries. Under the umbrella of Krugman framework, this study attempts to investigate the trade-induced child labor selection, scale and technique effects.

Literature that studies the link between the child labor and new trade focus on the issue of child labor policy. Most investigations into the trade-child labor relationships are based on the traditional theory of trade. There is no formal model that explicitly shows the impact of trade of differentiated products on child labor quality in the open economy context. In particular, the child labor impact of trade driven by economies of scale has not been explicitly differentiated from the child labor impact of trade driven by factor endowment differentials. Lastly, the theoretical and empirical analysis of the previous studies do not provide the effects of an increase in the stringency of child labor policy on domestic local regulatory quality levels in a model of monopolistic competition. The review of literature suggests that a large body of studies is available that deals with the issue of international trade and child labor separately. Only a few studies are available that bridge the issues of both disciplines to assess their causes and effects. Besides, only a few studies provide the linkage between trade and child labor by using the framework developed by the new trade theory. To the best of undergone whilst studying knowledge, there is no study available for the SAARC and ASEAN region that examines the relationship between new trade theory and child labor. This dissertation is therefore an attempt to fill this gap by employing econometric methods for the purpose of estimating the new trade and child labor relationship in SAARC and ASEAN region.

CHAPTER 4

THEORETICAL APPROACH AND METHODOLOGY

4.1 Chapter Overview

This chapter identifies the theoretical framework to be used in the thesis. A model created by this research shows the principal actors involved in the production, trade and child labor activities. Methodology part describes the research approach, the scheme of the data collection and method of developing empirical scenarios finalizes this chapter.

4.2 Theoretical Framework

The terminology of trade liberalization is usually in practice to narrate the accelerated spread of global trade and investment. It has gathered the momentum at an extraordinary rate in the recent era of globalization, thus obliging the countries to be more interconnected (Mahmood and Azhar, 2001), and leaves with two options; either adapting development path immediately in order to gain from the prospects of prosperity (Edmonds, 2010), or blaming themselves to a permanent process of communicative, political, social and economic marginalization (Mahmood and Azhar, 2001).



Figure 6: Theoretical Framework

Source: Various Studies

For Asian countries, a significant number of the world's working adolescent are to be traced, transformation to a trade liberalization basically means two things: firstly, they have an ability to enhance their exports and secondly, they have a greater capacity to bring a greater volume of foreign direct investment (Amjad et al., 2008).

The set trend shows that major technology in developing countries comes from the developed countries. Since there are relatively few case studies to derive a serious advancement on the relationship between trade liberalization and child labor. Trade and child labor literature tell us that child labor is essential to the competitiveness of less developed countries' industry in the global commercial markets. Child labor creates a competitive advantage of international trade in developing economies, while sometimes it discourages the foreign direct investment (Rule, 1985).

Comparative advantage theory of international trade (Ricardo, 1817) advocates that, free trade enables the economies to ensure both efficiency in production and allocation of resources. In Ricardian framework, there is a huge incentive for the children to work due to the single factor of production which is labor. In this framework the child labor and adult labor, both are identical in all aspects. That's why, in this domain, economies with labor intensive industries are blamed to be the major culprit of the child labor (Hanushek, 1992).

In Marxian Framework children are identified as a desired for survival and they are considered a production material. More specifically, the Marxian vision advocates that the proletarianization¹⁴, of the workforce brings about a fertility gain. Since, the working masses try to accumulate the factor of production over which they do have control (Smith, C. A., 1984). While, Malthus's vision designed on the belief of decreasing returns to scale¹⁵ (Herrera, 2011), whereas the new trade theory assumes that the economy displays the increasing returns to scale (Krugman, 1979). In the Malthusian framework, there is a trade off between resources and human capital (Fraser et al., 2003). Less human capital has more resources to utilize but child labor cannot be protected in this framework due to decreasing returns to scale. It means every next unit of labor is more expensive as compared to the previous unit. It is also assumed that all the labor is constant in the production process and child labor and adult labor are substitutes. This substitution may be harmful in the mitigation polices of child labor in Malthusian framework (Schultz, 2009).

Conventional trade theory anticipates that an increase in exporting commodity price may either decrease or increase the child labor, but it is based on the magnitudes of the income and substitution effects (Edmonds and Pavcnik, 2005). The literature suggests that the studies which examine the relationship between child labor and trade under the new trade theory are relatively scarce. In contrast, the mostly trade models examine the effects of international trade on the child labor are based on the traditional trade theory. Traditional trade theory advocates that the exchanges of goods and services is simply determined by relative factor differentials across the

¹⁴ In Marxisam frame work the proletarianisation is mostly seen as the most significant form of downward social monbility (Rule, 1985).

¹⁵ Output increases by less than the proportional change in inputs (Neumann et al., 2010).

countries on the basis of specialization in the production of goods in which the trading country has comparative advantage. Traditional trade theories assume that technology yields constant returns to scale¹⁶ in production, and markets are classified as perfect competition¹⁷. With this philosophy, economies deal in homogenous goods and services with each other. The advancement in the global trend of trade in differentiated products¹⁸ suggests that the trade of differentiated products is a dominant phenomenon in the worldwide exchange of goods and services. It demonstrates a substantial volume of trade flows for countries that are engaged in trade in differentiated products. New trade theory illustrates the trade as being driven by the specialization in, and the demand for, differentiated goods. The new trade theory relies on increasing returns to scale in production. Under these assumptions, trade may occur between countries with similar factor endowments and identical technologies, rather than between countries that are different in both aspects.

Lancaster (1980) proposed a model, by using differentiated goods, in which consumers can vary in their "ideal variety". However, Spence (1976) and Dixit and Stiglitz (1979) suggest an isolated consumer demanding multiple varieties or differentiated good, and they introduced a terminology the "love of variety". Lancaster (1980) and Helpman (1981) also adopt the "ideal

¹⁶ Production function which displays an equal proportion of inputs and outputs in the same direction (Ethier, 1982).

¹⁷ A situation in which infinite buyers and sellers exist, they are well informed and the market price of a goods is above the control of particular buyers and sellers (Stigler, 1957).

¹⁸ The process of distinguishing a product from others, to make it more charming (Grubel et al., 1975).

variety" approach in international trade models, while Krugman (1979, 1980 and 1981) carried forward the "love of variety" idea due to Dixit and Stiglitz (1979) concept of specialization under monopolistic competition. The extensive investigation by Helpman and Krugman (1985) shows that these two approaches are afforded in trade literature, so this thesis will use the similar " love of variety" approach as given by Krugman (1979). Feenstra (2003) describes this concept in a fascinating way and explain that the total number of varieties (number of firms) in a region necessarily falls with trade liberalization and the remaining firms have to increase the production to fulfill the market demand.

The primary purpose of this research is to provide new insights regarding the impact of international trade on the child labor by using new trade theory. Furthermore, this research is an empirical effort to explore the relationship between the child labor and the selection, scale and technique effects for the selected SAARC and ASEAN countries which are engaged in the trade of both differentiated and homogeneous goods. Finally, it examines the effect of trade intensity on child labor. There has not been an empirical study that specifically addresses factors which influence the child labor, in the production and trade of differentiated products, namely the selection, scale and technique effects. Notably, the selection is specific to the trade of differentiated goods and a unique contribution in trade and child labor literature. Additionally, the scale and technique effects arising from the production of differentiated goods have not been explicitly shown to generate similar effects as the scale and technique effects arising from the production of homogenous goods. In this research, the consequences of child labor due to

trade of differentiated products are integrated with the child labor effects of trade of homogeneous goods.

4.3 Conceptual Framework

Trade expands the consumption basket and enables the consumers to use both local and imported varieties of goods. Krugman's (1979) framework of international trade states that the industrial sector, having features of increasing returns to scale and total number of firms in the industry shrinks although each one of the remaining firms produces more than before (Feenstra, 2003). At the same time, when income level of a country increases due to trade openness; the country implements stricter child labor approach, which causes amplified child labor reduction activity (Edmonds, 2007 and 2015). The increasing returns to scale in Krugman's framework compile the producer to hire more efficient and skilled labor to raise the production level. This framework has a potential to support the child labor elimination policies and discourage the child labor, because the producer wants to produce more and additional unit of adult labor is not expensive for him due to increasing returns to scale. The new theory suggests that the increase in the trade liberation shrinks the number of the firms in the locality; only the competitive firms can survive in the region other non-competitive firms have to exit from the competition. Therefore, the child labor in the competitive environment decreases.

Figure 7 shows that in the open economy, free trade generates trade induced child labor selection, scale and technique effects. When trade opens, countries expand production, but the

number of firms in the economy falls. This framework contains three predictions concerning the impact of trade on the productivity of firms: one, some firms are forced to exit (selection effect), two, surviving firms expand their outputs (scale effect), and finally these results provide a basis for the technique effect, which leads to higher income levels, which induces better technique of production undertaken by the firm.



Figure 7: Trade Induced Child Labor Effects

In Figure 7, the trade induced selection effect is an avenue through which child labor in a country can change following trade liberalization. In particular, the least efficient firms have to exit. The remaining firms which are relatively efficient and reluctant to use child labor in the

production are free to raise the productivity. Under these two circumstances the technique effect is sourced from the cross-country diffusion of efficient techniques. In simple words, the figure explains that holding the scale effect and other determinants constant, the technique effects refer to the turnabout in the child labor as a result of adjustments to the income level due to trade intensity.

Based on theoretical and conceptual frameworks, this dissertation proposes the following hypotheses about the trade-child labor relationships which describe the impact of both trade of differentiated products and homogeneous goods on child labor.

4.3.1 Hypothesis 1

Child labor is increasing in the level of the scale of production, or positive trade induced child labor scale effect.

Consider an economy with a labor-intensive sector that produces differentiated goods. Then, the production of differentiated goods corroborates that the scale of economic activities determines child labor in the economy. Given technology, an increase (decrease) in scale of labor intensive industries leads to an increase (decrease) in the labor demand (Hamermesh, 1987), child labor is a substitute of adult labor and child labor is parsimonious as compared to adult labor (Basu and Tzannatos, 2003), so an increase (decrease) in labor demand leads to an increase (decrease) in child labor demand. Theory suggests that the scale effects implies a
positive relationship between labor and the production of labor intensive goods (Hamermesh, 1987, 1996). In fact, researchers as far back as, Marx (1867) has argued that variation in child labor is primarily labor demand driven (Edmonds, 2007). In the current analysis, the direction of the effect of the scale on the child labor is similarly expected to be positive.

However, in contrast to the past studies, the current study assumes that the trade-induced scale effect represents the expansion of industries due to trade. Hence, the scale effect in the current analysis assumes that an increase (decrease) in the scale of production is not only due to the expansion (contraction) to the production of goods with comparative advantage, but it is also due to an expansion (contraction) of economies of scale. Consistent with the theories, the aggregate scale effect of trade of differentiated products is assumed to be increasing in the child labor.

4.3.2 Hypothesis 2

Child labor is negatively associated with income level due to changes in trade intensity, or negative trade induced child labor technique effect.

In an economy, on the one hand, the child labor regulation moves towards reduction and elimination of child labor in autarky situation (Weiner, 1991). On the other hand, an increase in the trade intensity increases the nation's wealth, this prosperity oscillate the nation behavior towards the labor amelioration (Karlan and Valdivia, 2011). Holding the scale effect and other

determinants constant, the technique effects refer to the turnabout in the child labor as a result of adjustments to the income level due to trade intensity. Literature suggests that, assuming child labor is unskilled and income levels are tied to trade intensity, then the trade induced child labor technique effect or income effect reduces the child labor in the economy.

Income levels influence the child labor policy, negatively associated with child labor. An increase (decrease) in income level implies stricter (more lax) policy which leads to an inverse change in child labor intensity. This implies that holding the scale and selection effect constant, higher income levels lead to lower child labor. Thus, the current analysis hypothesizes a negative relationship between child labor and income level.

4.3.3 Hypothesis 3

Child labor is decreasing with the decrease in number of firms, or positive trade induced child labor selection effect.

In the model of Krugman (1979), firms are identical in size and production technology. With trade, consumer chose to consume foreign products resulting in a decrease in the consumption of domestic varities. The fall in the number of domestic product varieties leads to the exit of unprofitable firms. Assuming the scale and techniques of production remains unchanged, a smaller the number of firms in an industry implies less labor activities, which means less child labor. Hences, everything else equal, openness to trade implies a fall in the number of the firms which leads to a fall in the child labor. In other words, the child labor is decreasing in the number of firms, citris paribus.

4.3.4 Hypothesis 4

The child labor is decreasing in the trade intensity or openness to trade.

The impact of free trade on child labor is measured by the variable defined as the ratio of trade volume to GDP, namely, trade intensity. In this analysis, it is postulated that freer trade leads to lower child labor. This hypothesis is consistent with findings in Neumayer and De Soysa (2005). There are many reasons why openness to trade may contribute to the improvement of the child labor situation. One reason is that greater openness to trade implies an increase in trade flows which leads to an expansion in the scale of overall production. Consequently, income level rises, which implies a stricter child labor policy that stimulate the amelioration and lower child labor intensity (Grootaer and Kanbur, 1995). Further, openness to trade may accelerate an inward flow or diffusion of more efficient techniques which contributes to reducing child labor intensity. Therefore, holding other factors constant, greater openness to trade leads to a negative growth in the child labor.

4.4 The Model

The model formulated in this research is an extension of Krugman (1979) framework of international trade, provides an elegant account of the trade of differentiated products based on the monopolistic competition and increasing returns to scale, by incorporating the child labor to examine the impact of trade on child labor. This section analyses the characteristics of closed economy along with an open economy. In this model, there is a Chamberlin economy in which firms are producing differentiated and imperfectly substituted goods; firms has some extent of market power and the key objective of these firms is to maximize their profits; non zero profits indulge the firms to enter or exit in the monopolistic competitive environment. Consumers, firms and regulatory authority are major players of the economy; monopolistically competitive market structure exists in the economy along with increasing returns to scale in production process, due to economies of scale each firm is producing only one type of product variety. Producers are equal, expect their design of production; similarly, firms are producing differentiated goods with a large number of potential varieties by using identical technologies.

Firm behavior is independent with other firms in the market due to production of large number goods and this production has a negligible effect on the market price and demand of other goods. In this framework labor is a single factor of production and this labor is inelastically supplied in a competitive market. Income and wages are interdependent because income comes from wages in the economy. Buyers and sellers have all possible information due to which there is an instantaneous adjustment to the change in variables. Countries, in this framework, are considered similar in different aspects like technology, preferences and size.

In the economy, there are N number of consumers having similar preferences and utility function is separate-able with the preference of variety with respect to consumption of goods. The positive utility is obtained by using i_{th} , good and a mildly negative utility ¹⁹ is associated with child labor, but the utility of leisure is excluded due to Krugman (1979) model constraints. Horizontally differentiated products symmetrically enter into utility function. Social damage comes from child labor, due to weak regulation in the society and consumers have no control over child labor. Therefore, the consumer can only maximize its utility by consuming goods. Thus the utility function can be written as.

$$U = \sum_{i=1}^{n} V(q_i) - \sum_{i=1}^{n} \zeta l_i \quad \dots (1) \qquad \qquad V > 0, V'' < 0$$

The consumer can maximize his utility by using its income which is equal to his wage, wage of child labor depends on age of child worker (θ), the consumer maximization is as follow:

$$Max_{\{q_i\}} U = \sum_{i=1}^{n} V(q_i) - \sum_{i=1}^{n} \zeta l_i (15 - \theta)$$
 (2)

¹⁹ Higher level of marginal disutility from children engaged in economic activity (Reggio, 2011).

subject to: y = w and $5 \le \theta \le 15$

where $y = \sum_{i=1}^{n} p_i q_i$, p_i denotes the price of i^{th} goods. For maximization of consumer utility the first order condition is as follows:

$$v'(q_i) = \lambda p_i \dots (3)$$
 $i = 1, 2, \dots, n$

Where λ is the marginal utility of income and langrange multiplier. The share of each variety in the budget is insignificant due to enormous varieties of goods and services; therefore, the price impact of marginal utility of income can be avoided (see Krugman, 1979). The effect of a change in price implies that the elasticity of demand for variety *i* can be written as:

$$\psi_i = \frac{dq_i}{dp_i} * \frac{p_i}{q_i} = -\left(\frac{v}{q_iv'}\right) > 0\dots(4) \qquad \qquad \frac{d\psi_i}{dq_i} < 0 \quad (\text{see, Krugman, 1979})$$

Hence, the elasticity is increasing as move up along with demand curve but consumption is falling.

Krugman (1979) suggests the linear function of labor output along with identical technologies having increasing returns to scale. The returns to scale is internal to firm having initial positive fixed cost, declining average fixed cost and the most important constant marginal cost. The output x_i , is an increasing function of labor:

$$N_i = \alpha + \beta x_i \dots (5) \qquad such that : \alpha \& \beta > 0$$

Where α is fixed cost of production and β is the coefficient of variable cost. Hence, following equation shows the relationship of child labor and output.

$$\varsigma l_i = N * x_i \ldots (6)$$

In the model locally engaged children are under consideration, it is also assumed that government or regulatory authority is not directly involved in the amelioration of labor, the regulatory authority can use child labor penalty as a policy tool and firms have to pay a high price in form of child labor penalties for engaging a child labor. In this framework, the child labor penalty is determined exogenously, in response to the implementation of child labor regulation. Firms try to avoid costly child labor penalty payments. In this model, costly child labor penalty payments are taken from the output that a firm produces by using labor including child labor. Therefore, firms allocate a proportion of output for the child labor penalty payments, only in case of engaging the child labor and allocate remaining portions for goods consumption in the market.

If a firm allocates x_i^a unit of output in amelioration of labor, then net output:

$$x_i^{net} = x_i \left(1 - \mathring{A}_i \right) \qquad \dots (7)$$

Where $A_i = \frac{x_i^a}{x_i}$ is the allocation of resources from output towards hiring more skilled workers and terminate the less skilled workers (or child labor) engaged in the production process, this process in the current research is called amelioration. The individual consumers are also taking part as workers in the production of goods, the total labor force in the economy is N. Then, the supply is equal to demand of produced output, this relationship can be expressed as:

$$(1 - Å_i) x_i = Nq_i \qquad \dots \qquad (8) \qquad \qquad 0 < Å_i < 1$$

Where $(1 - Å_i)$ is the fraction of output, which is allocated for consumption. This research specifies the functional form for child labor to describe the relationship between child labor and output, which is consistent with the approach of specifying the production technology in Krugman (1979,1980) models. This child labor intensity takes the following form:

$$\varsigma l_i = (1 - \mathring{A}_i)^{\sigma} \qquad \dots (9) \qquad \sigma > 1$$

The parameters σ measures the responsiveness of a change in child labor due to changes in the fraction of output allocated towards the amelioration. The greater the value of responsiveness measures (σ) means that lesser child labor, similarly the greater the value of amelioration (Å) the lesser will be child labor. Generally, in labor intensive industries such as manufacturing and agriculture, labor demand increases at an increasing rate with the increasing allocation of resources towards the production of goods and child labor demand is directly associated with labor demand in the economy due to substitution effect. In simple words, the scale of production can increase the demand for child labor, keeping other things constant in an economy. Therefore, for the functional form specified above, the assumption that σ is greater than one is reasonable and has a basis in the practical sense.

Tabl	le 8:	Exp	lanation	of Im	portant	Sym	bol	S
------	-------	-----	----------	-------	---------	-----	-----	---

Variables	Explanation
Amelioration (Å)	Amelioration is the allocation of resources from output towards hiring more skilled workers and terminate the less skilled workers (or child labor) engaged in the production process, this process can be illustrated as $(0 < \text{Å}_i < 1)$.
Measures the Responsiveness (σ)	The parameters σ measures the responsiveness of a change in child labor due to changes in the fraction of output allocated towards the amelioration. The greater the value of responsiveness measures (σ) means that lesser child labor.
Child Labor Penalty (<i>t</i>)	The child labor penalty, t , is taken to be high enough so that firms choose to undertake child labor amelioration activity. If regulatory authority does not impose child labor regulation then it is equal to zero. In the absence of child labor regulation or at zero child labor penalty firms have no incentive to restrict the child labor in the society.

The child labor penalty, t, is taken to be high enough so that firms choose to undertake labor amelioration activities and avoid from child labor. If regulatory authority does not impose child labor regulation then it is equal to zero. In the absence of child labor regulation or at zero child labor penalty firms have no incentive to restrict the child labor in the society. This model without child labor entanglement would be quite similar to the Krugman (1979) model of monopolistic competition. The profit function, denoted π , is the firm's revenue less labor cost and child labor penalties given by the following equation:

$$\pi_{i=} p_i \left(1 - \mathring{A}_i\right) x_i - w(\alpha + \beta x_i) - t \varsigma l_i x_i \qquad \dots (10)$$

Since all the firms are identical, symmetry across firms implies that

$$p = p_i$$
$$x = x_i$$
$$Å = Å_i$$
$$\varsigma l = \varsigma l_i$$

Henceforth, subscripts are suppressed.

4.5 Monopolistic Competition and the Child Labor

The profit function is the firm's revenue less labor cost and child labor penalties such that:

$$\pi_{i=} p_i \left(1 - \mathring{A}_i\right) x_i - w\alpha - (w\beta + t\varsigma l_i) x_i \quad \dots (11)$$

Where in monopistic competition P = AR = ATC = LRAC

$$p = p_i \left(1 - \mathring{A}_i \right) x_i \qquad \dots (12)$$

4.6 Trade Induced Child Labor Effects

In the autarky case the child labor penalty is considered as preponderant moral sentiment (see also Appendix A), while in the open economy case trade provides a sustainable, scalable and replicable solution to combat the child labor locally and globally. Changes in the trade intensity can have important implication effects on the consumer welfare as well as on economic preference. If a country wants to have a better quality care for vulnerable children, for example, in the form of no child labor is desirable, then, trade liberalization can be the best compliment for the policy makers. However, trade liberalization affects the production decisions at the aggregate level. Hence, in the developing economy, there exists a trade-off between the achievements of labor amelioration and economic objectives.

4.6.1 Effect of Trade Openness on Child Labor

The effect of a change in trade intensity on child labor is given by the following:

$$cl = \left(\frac{w\beta}{T(\sigma-1)}\right) \dots (1.1)$$
$$\frac{\partial cl}{\partial T} = -T^{-2} \left(\frac{w\beta}{\sigma-1}\right) < 0 \dots (1.2)$$

$$\frac{\partial cl}{\partial T} = -T^{-2}(cl) < 0 \dots (1.3)$$

This result is consistent with the theoretical expectation that the greater the trade liberalization, the lesser will be the child labor.

Trade Induced Child Labor Effects	Explanation
Scale Effect	Holding all other factors constant a change in the growth of the economy due to trade liberalization yields a trade induced child labor scale effect.
Technique Effect	Holding other factors equal a change in the income growth of an economy due to trade liberalization leads to trade induced child labor technique effect.
Selection Effects	Holding scale and technique effect constant a change in the trade intensity leads to a change in the number of domestic firms, which implies a trade induced child labor selection effect.
Total Impact of Trade	Total impact of trade on child labor depends upon the sum of the magnitude of trade induced child labor selection, scale and technique effect.

Table 9: Trade Induced Child Labor Effects

In the previous studies, i.e. Krugman (1979,80) discusses the two types of trade induced effects, namely scale and selection due to trade liberalization (Feenstra, 2003). Copeland and Taylor (2001, 2003 and 2013) use Heckscher-Ohlin framework and find three types of trade induced effects i.e. scale, technique and composition. Aralas (2010) sums up the Krugman (1979 and 1980) and Copeland and Taylor (2001 and 2003) work and concluded there are four types of trade induced effects scale, selection, technique and composition. In this section, this research checks the effect of trade intensity on the child labor by using the trade induced child labor selection, scale and technique effects on the similar grounds.

4.6.2 Selection Effect

Thus, in the ongoing analysis, the trade intensity has a negative effect on the number of firms. The new theory suggests that the increase in the trade liberation shrinks the number of the firms in the locality; only the competitive firms can survive in the region other non-competitive firms have to exit from the competition. The competitive strategy is always more profitable than the exploitative strategy. Estevez (2011) concludes that child labor is an inappropriate option for producers than unskilled adult labor even after accounting for productivity differences. Therefore, the child labor in the competitive environment decreases and it is called a trade induced child labor selection effect.

The change in trade intensity on the number of firms is as given by the following:

$$n = N \left[\alpha + \frac{\beta N q_i}{\left(\frac{w\beta}{T(\sigma-1)}\right)^{\frac{1}{\sigma}}} \right]^{-1} \dots (2.1)$$
$$\frac{\partial n}{\partial T} = -N \left[\alpha + \frac{\beta N q}{\left(\frac{w\beta}{T(\sigma-1)}\right)^{\frac{1}{\sigma}}} \right]^{-2} \left[\frac{\beta N q}{\sigma} \left(\left(\frac{w\beta}{T(\sigma-1)}\right)^{-\left(\frac{1+\sigma}{\sigma}\right)} \right) \frac{w\beta}{T^2(\sigma-1)} \right] \dots (2.2)$$

$$\frac{\partial n}{\partial T} = -N \left[\alpha + \frac{\beta N q}{c l \sigma} \right]^{-2} \left[\frac{\beta N q}{\sigma} \left((cl)^{-\left(\frac{1+\sigma}{\sigma}\right)} \right) T^{-1}(cl) \right] < 0 \qquad \dots (2.3)$$

Dividing equation (2.3) by (1.3)

$$\frac{\frac{\partial n}{\partial T}}{\frac{\partial cl}{\partial T}} = \frac{-N\left[\alpha + \frac{\beta Nq}{cl^{\frac{1}{\sigma}}}\right]^{-2} \left[\frac{\beta Nq}{\sigma} \left((cl)^{-\left(\frac{1+\sigma}{\sigma}\right)}\right) T^{-1}(cl)\right]}{-T^{-2}(cl)}$$
$$\frac{\frac{\partial cl}{\partial n}}{\frac{\partial cl}{\partial n}} = N\left[\alpha + \frac{\beta Nq}{cl^{\frac{1}{\sigma}}}\right]^{2} \left[\frac{\beta Nq}{\sigma} \left((cl)^{\left(\frac{1+\sigma}{\sigma}\right)}\right) T^{-1}\right] \dots (2.4)$$
$$\frac{\frac{\partial cl}{\partial n}}{\frac{\partial cl}{\partial n}} > 0 \dots (2.5)$$

4.6.3 Scale Effect

It typically remains the truism that the trade increases scale of the economy and leads to an increase in the number of products produced, or an increase in the scale of production (Francois, 1990).

$$x = (1 - \text{\AA})^{-1} \text{Nq} \dots (3.1)$$
$$x = \left(1 - \left(1 - \left(\frac{w\beta}{T(\sigma - 1)}\right)^{\frac{1}{\sigma}}\right)\right)^{-1} \text{Nq} \dots (3.2)$$
$$\frac{\partial x}{\partial T} = \frac{Nq}{T\sigma} (1 - \text{\AA}) (\text{cl})^{\frac{2-\sigma}{\sigma}} \dots (3.3)$$
$$\frac{\partial x}{\partial T} > 0$$

Equation 3.3 shows the trade openness increases the scale of production in the economy. The increase in scale of production due to trade brings income in the country and this is source of better technique of production in the economy.

$$\zeta l = \text{\AA} \cdot x \dots (3.4)$$
 (see, Apendex A)
 $\frac{\partial \text{cl}}{\partial x} = \text{\AA} \dots (3.5)$

Equation 3.5 shows that increase in scale of production is positively associated with child labor. The silver lining of this result is a source of improvement in the production techniques of an economy.

4.6.4 Technique Effect

Theoretically, an increase in the trade intensity brings a positive change in the income level of the nation and has a direct effect on the performance and techniques of the production. These techniques excel the skills of the labor in the trading countries. This analysis hypothesized that trade induced technique effect in the trading economy positively affects the amelioration. The amelioration is basically improvement in the labor standard; it means skilled, educated and adult labor is demanded for the production purpose in this framework. Child labor is supposed unskilled, uneducated and low productive labor in this analysis. The increase in the amelioration means the decrease in the child labor. Thus, the trade liberalization increases the marginal benefit of reducing child labor. The effect of a change in the trade intensity to the level of amelioration can be shown as follows:

$$\mathring{A} = 1 - \left(\frac{w\beta}{T(\sigma-1)}\right)^{\frac{1}{\sigma}} \quad \dots (4.1)$$

$$\frac{\partial \mathring{A}}{\partial T} = \frac{(T^{-2})}{\sigma} \left(\frac{w\beta}{T(\sigma-1)} \right)^{\left(\frac{1}{\sigma}-1\right)} \left(\frac{w\beta}{\sigma-1} \right) > 0 \dots (4.2)$$
$$\frac{\partial \mathring{A}}{\partial T} = \left(\frac{T}{\sigma} \right) (cl)^{\left(\frac{1}{\sigma}-1\right)} (cl)^{1} > 0 \dots (4.3)$$

$$\frac{\partial \mathring{A}}{\partial T} = \left(\frac{T}{\sigma}\right) (cl)^{\left(\frac{1}{\sigma}\right)} > 0 \quad \dots \quad (4.4)$$

Dividing equation (4.4) by (1.3)

$$\frac{\frac{\partial \text{\AA}}{\partial \text{T}}}{\frac{\partial \text{cl}}{\partial \text{T}}} = \frac{\left(\frac{\text{T}}{\sigma}\right)(\text{cl})^{\left(\frac{1}{\sigma}\right)}}{-\text{T}^{-2}(\text{cl})}$$
$$\frac{\partial \text{\AA}}{\partial \text{cl}} = -\left(\frac{\text{T}^{3}}{\sigma}\right)(\text{cl})^{\left(\frac{1-\sigma}{\sigma}\right)}$$
$$\frac{\partial \text{cl}}{\partial \text{\AA}} = \frac{-\sigma}{T^{3}(\text{cl})^{\left(\frac{1-\sigma}{\sigma}\right)}}$$

$$\frac{\partial \mathrm{cl}}{\partial \mathrm{\AA}} < 0$$

The result shows that trade positively affects the amelioration or technique effect, while effect is negatively associated with child labor. However, at this point it's worth considering that the increase in labor standards or amelioration means a decrease in the child labor, in other words, it can be stated that trade induced child labor technique effect is negative as this research hypothesized earlier. The increase in trade intensity necessitates the firm to increase the allocation of output for labor amelioration and reduce the child labor.

4.7 The Estimation Model of Trade of Differentiated Products

In this section, this research expands the disintegration of the child labor effects of trade of differentiated products to examine the link between child labor and economic factors that define the trade of differentiated products. This analysis shows how the theoretical anisotropy in chapter four transformed into estimating equations for empirical analysis. The estimating equation will describes the trade effects on number of firms (differentiated products) intimated in the earlier section. SAARC and ASEAN countries in the sample are assumed to engage in the trade of both differentiated and homogeneous products with trading partners. This is one of the unique contribution to show the effects of trade on child labor empirically, by using trade induced effects. To achieve an empirical equation, the theoretical relationship needs to be transformed that can be estimated.

4.7.1 Explanation of Variables

There is no consistent measure, interpretation of the lawful age and working conditions of child labor across the countries. Two primitive categories are usually recognized in the child labor studies (i) economic activities and (ii) hazardous work. Child labor activity is inherently more difficult to measure than simple labor activity. Sometime, it rules out the necessary chores that children undertake as household member subject to important seasonal variations (Anker, 2000). According to the international definition of employment, the child labor can be identified as economically active children if they work for at least one hour in a reference week²⁰. Thus, child labor in this research is a substantial as promulgated by the ILO and world development indicators, which is based on the agency's Minimum Age Convention No 138 (1973). Child labor ζl_{it} is surprisingly difficult to define, this dissertation establishes a normal definition and considered children under the age of 14, engage in economic activity directly in all sectors of the economy or school going children in employment, are part of child labor. This thesis takes children in employment percentage of children ages 5-14 as child labor based on available data base. This analysis looks at child labor in 9 SAARC and ASEAN countries and tests the robustness for the choice years 1999 to 2013. The intertemporal variations in child labor data²¹

²⁰ The calendar week containing the 12th day of the month, which is used in the current population survey as the time period for documenting the employment and labor force status of respondents.

²¹ Most of countries have some independent observations for a given years. Some observations are taken by imputation rather than actual variation in child labor due to infrequent surveys, generally, for SAARC countries. For controlling the time invariant country characteristics a panel data is recommended by this study.

are driven by imputation²² and some adjustments based on country wise household surveys at national level. Data adjustment is made to make the data comparable because data of all countries are not available for the same year.



Figure 8: Child Labor and Firms Density

Source: UNICEF (2012) and World Development Indicators (2012).

²² The missing data were imputed with the average of the observations in countries of the same region. Then the regional average is combined with the information available from the country to compute the child labor status (Pasteels, 2013).

In the theory, the differentiated products derive from the variation in the number of firms engaged in the production of differentiated goods or in the number of product varieties (Startz, 1989 and Feenstra, 2003). Along these lines, in the data, the differentiated product can be characterized by two measures: one, by the change in the number of domestic firms, and two, by the change in the number of product varieties produced domestically (Feenstra, 2003). Krugman (1979) talks about the preference of variety based on selection of variety (differentiated products). This is the same concept as explained by Feenstra (2003). Thus, in the data, the selection effect can be represented by two measures: one, by the change in the product varieties produced domestic firms. In current analysis change in the preference of variety is explained by number of firms. Consistent with other measure, the measure of the number of firms is in intensive form and it is number of listed domestic companies per squared kilometer (companies/km²).

Domestically incorporated companies in the country's stock market are used as a proxy for a number of firms. SEL_{it} is used to show the effect of change in number of firms on child labor. This effect is due to a change in the number of firms in the economy on the basis of selection of product variety. SEL_{it} includes all kinds of companies, which are engaged in production of goods. The theory suggests that in general, the number of firms is directly associated with labor demand (Brainard and Riker, 1997; Fehr et al., 1998 and Felbermayr and Prat, 2011), similarly, child labor is directly associated with the demand for labor (Humphries, 2010). Figure 8 presents the graphical observation of raw data to support the understanding of selection effect with respect to child labor. In our empirical model SEL_{it} is associated with the country-specific number of listed companies. Thus, the child labor and trade model suggests a definition of product variety which can be best defined by firm-level production that is the number of firms engaged in production of differentiated products. Accordingly, this research uses the number of listed domestic companies to represent the selection effect, sourced from the World Development Indicators (WDI) for the years 1999-2013.

	Explanation	Proxy	Variable Measurement
	Change in the number of product	Product Price	-
Salastian affast	varieties		
Selection effect	Change in the number of firms	Listed companies	Companies/km ²
		in stock markets	
	Industrial production	Industrial	-
Scale effect		production index	
	Size of economy	GDP	GDP/km ²
	Change in techniques of production	FDI	-
Technique effect	Change income level	GNP or NFFI	NFFI/Population
	Capital intensity	K/L	-
	Trade barriers	Custom duties	-
Trada openness	Trade shares	Trade intensity	(X+M)/GDP
rade openness	Trade volumes	X+M	-
	Net exports	X-M	-

Table 10: Variable Explanation and Measurement

A further concern identifies the rationale and visualization of the scale and technique effects in the data. The data used for scale effect can be minimal or extensive depending on the

level of accuracy and should complement with the aggregate effects of the change in child labor that is due to both of homogeneous goods and differentiated goods. To simplify the discussion in this document, it is anticipated from Mukhopadhyay and Chakraborty (2005) and Copeland and Taylor (2001) that, the scale and technique effects as represented by gross domestic product (GDP) and gross national product (GNP), respectively, measure the effect of trade. It is worth noting at this stage trade consists of both homogeneous and differentiated products.

For this study, it might be important to discuss the problem associated with the gross domestic product (GDP) which is used to show the scale effect. GDP explicate the variation in the scale of production due to a change in the supply of both differentiated and homogeneous goods, it is prudent to assume that the child labor scale effect illustrated by the change in gross domestic product is the aggregate effect of the changes from both the production of homogeneous and differentiated goods. The scale effect in this study is GDP per kilometer squared, because there are notable cross country differences in the GDPs of SAARC and ASEAN countries that can avoid the pragmatic approach to evaluate the impact of scale effect. Graphical presentation of raw data for this variable in Figure 9 shows an association of scale effect and child labor.

However, a similar reasoning strategy in this context can be applied to the technique effect, conveniently, the income effect measured the change in child labor due to changes in income or gross national income. More precisely, the technique effect is the improvement in child labor as firms undertake the amelioration of labor to meet labor regulation standards, so,

the higher the income level the more stringent child labor regulation will be. Thus the change in child labor can be represented by the change in the technique of production (see Karlan and Valdivia, 2011) where an increase in income level is associated with a decrease in child labor.



Figure 9: Child Labor and Scale of Production

Source: UNICEF (2012) and World Development Indicators (2012).

In the presence of GDP, the GNP is the source of high correlation among these variable. This study takes the difference of GDP and GNP which is net foreign factor income (NFFI). In simple word it is the difference between payment given to home country labor from foreign people and payment given to foreign country labor from home country people. This technique is more appropriate in the Krugman framework, because in this framework key factor of production is labor. In this way a country which has more skilled labor, will take more income from foreign countries or in other words more skilled labor will use better technique.

4.7.2 Child Labor Effects in the Closed Economy

The empirical evidence of child labor in the close economy is essential to the debate on the performance of different effects. Model-A shows that the change or growth in total child labor can be decomposed into, or is the sum of, the scale, selection and technique effect (respectively). The model can be quite easily solved numerically for a variety of parameter values. It is possible to proceed entirely numerically from this point. If we take a somewhat different approach, however, it is possible to characterize the properties of the model analytically. Model-A allows to express the demand of child labor in an autarky situation with respect to time (t) across the countries.

$$\varsigma l_{it} = \alpha_0 + \beta_1 SCL_{it} + \beta_2 SEL_{it} + \beta_3 TEC_{it} + u_{it} \dots (A)$$

4.7.3 Child Labor Effects in the Open Economy

Model-A does not express the effect of trade liberalization, so this study introdues a trade variable. TR_{it} is used to examine the effect of overall trade on the level of child labor. TR_{it} is defined as the openness of trade in term of trade intensity and calculated as the import plus export ratio to GDP $\left(\frac{(X+M)}{GDP}\right)_{it}$.



Figure 10: Child Labor and Trade Openness Source: UNICEF (2012) and World Development Indicators (2012).

Trade intensity is thus the share of total trade in the GDP. The greater the share of trade in the GDP the greater will be trade intensity of a country and thus more open will be the economy to foreign competition. Use of trade intensity is very common in the literature, Fung and Maechler (2007) and Alam et al. (2011) and Copeland and Taylor (2001) utilize this variable to express the trade liberalization. Following this tradition, the current exploration also include this variable in the model. The graphical observation of raw data supports the understanding of trade openness and child labor in Figure 10. In the current exploration, the trade intensity is exercised to measure two kinds of effects. Firstly, it establishes the link between trade liberalization and child labor. Secondly, it is to explained in interaction forms to express the responses trade induced selection effect, scale, and technique effects. So, Model-A can be rewritten as:

$$\varsigma l_{it} = \alpha_0 + \beta_1 SCL_{it} + \beta_2 SEL_{it} + \beta_3 TEC_{it} + \beta_4 TR_{it} + u_{it} \dots (B)$$

4.7.4 Trade-Induced Child Labor Effects

Model-B only shows the effect of selection of product variety, scale of production, technique of production and trade on the child labor. To derive trade related child labor effects, an interaction term TR_{it} can be introduced along with the variables representing the scale of production, technique of production, and number of firms represent the trade-induced child labor scale, technique and selection effects.

Adding an interaction term to Model C drastically changes the interpretation of all of the coefficients of selection, scale and technique effect variables in Model A. It is now interpreted as the unique effect of selection, scale and technique effects on child labor only when trade = 0. But the interaction means that the effect of these variables on child labor is different for different values of trade. So, the unique effect of selection, scale and technique effects on child labor is compared to the values of trade. So, the unique effect of selection, scale and technique effects on child labor is not limited, but also depends on the values of trade-interacted variables (trade induced child labor effects).

Following is the linear model that can be used to examine the trade-induced child labor effects:

$$\varsigma l_{it} = \alpha_0 + \beta_1 SELTR_{it} + \beta_2 SCATR_{it} + \beta_3 TECTR_{it} + u_{it} \dots (C)$$

Where, $SELTR_{it}$ variable shows the change that trade brings about in a number of firms. Trade creates a fall in the number of firms due to economies-of-scale internal to the firms. This is because economies of scale make it valuable for a country to specialize in the production of only a limited range of products. The effect of a change in the number of firms on the level of child labor as a result of the change in trade intensity is called trade-induced child labor selection effect. In our empirical specification domestic companies per squared kilometer along with the openness of trade is trade-induced child labor selection effect (*SELTR*)_{it}.

 $SCLTR_{it}$ variable shows that change in the trade intensity brings a change in scale of production. The effect of the change in scale of production at the level of child labor due to a change in trade intensity is called trade-induced child labor scale effect. In our empirical specification $SCLTR_{it}$ is the i_{th} country's gross domestic product per squared kilometer interacted with trade intensity (this is trade-induced child labor scale effect) at time t.

NFFI is used to determine the child labor technique effect. To find the trade-induced child labor technique effect, we interact the net foreign factor income (NFFI) per capita with trade intensity $(TECTR)_{it}$. Copeland and Taylor (2001) use this trade-induced technique effect

in their work on trade and environment. Trade-induced child labor technique effect shows the effect of change in the income level on the level of child labor.

4.8 Measurement and Data Sources

The dependent variable, child labor, is the proportion of 5 to 14 year old children involved in economic activity that exists generally in developing economies. Using multiple imputations to adjust for missing values for child labor data, collected from various sources²³. Data on gross domestic product, gross national product in 2000 dollars and other macroeconomic variables is sourced from the world development indicators, for the years 1999-2013.

Gross domestic product measures the scale effect induced by a change in size of the economy. In order to avoid strong correlation between GNP and GDP, the difference between the GNP and GDP will be used to disconnect the income effect from the scale effect. The new explanation of technique effect is measured as net foreign factor income (NFFI= GNP-GDP) or the net amount of nation earned in a fiscal year. Domestically incorporated companies in the country's stock market are used as a proxy for a number of firms. Consistent with other

²³ Data is gathered by the access to specific databases, i.e. ILO (IPEC reports), The United Nations Children's Fund (UNICEF), Understanding Children's Work (UCW), Department of Census & Statistics from different countries and World Development Indicators etc. A comprehensive internet searches and e-mails to key informants are also made to collect the related child labor data.

measure, the measure of the number of firms is in intensive form and it is number of listed domestic companies per squared kilometer (companies/km²).

Dependent Variable			
Variable Name	Variable	Variable Explanation	
	Code		
Child Labor	ςl_{it}	Percentage of children (aged 5-14) engaged in child labor	
Data Source	Various So	ources	
Independent Variables			
Variable Name	Variable	Variable Explanation	
	Code		
Selection Effect	SEL _{it}	Country specific number of listed companies per	
		squared kilometer.	
Scale Effect	SCL _{it}	Gross Domestic Product (GDP) per squared	
		kilometer	
Technique Effect	TEC _{it}	Net foreign factor income (NFFI) per capita	
Trade	TR _{it}	Import plus export ratio to GDP	
Trade Induced Technique Effect		Technique Effect interacted with the openness of	
	I ECI R _{it}	trade	
Trade Induced Selection Effect	SELTR _{it}	Selection Effect interacted with the openness of trade	
Trade Induced Scale Effect	SCLTR _{it}	Scale Effect interacted with the openness of trade	

Table 11: Variable Description

Note: Data for independent variables is derived from World Development Indicators

Openness of trade²⁴ in term of trade intensity (TR_{it}) is calculated as the import plus export ratio to $\text{GDP}\left(\frac{(X+M)}{GDP}\right)_{it}$. Trade intensity is thus the share of total trade in the GDP. Alam et al. (2011) and Tariq and Ab-Rahim (2016) utilize this variable to express the trade liberalization. Following this tradition, the current exploration also includes this variable in the model. The effect of a change in the number of firms on the level of child labor as a result of the change in trade intensity is called trade-induced child labor selection effect. In this study domestic company per squared kilometer along with the openness of trade is used to express the trade-induced child labor selection effect $(SELTR)_{it}$. While, the effect of the change in scale of production on child labor due to a change in trade intensity is trade-induced child labor scale effect. This study interacts the trade intensity with the country's gross domestic product per squared kilometer to show the trade-induced child labor scale effect at time t. Finally, to find the trade-induced child labor technique effect, this study interacts the net foreign factor income (NFFI) per capita with trade intensity $(TECTR)_{it}$. Copeland and Taylor (2001) use this tradeinduced technique effect in their work on trade and environment. Trade-induced child labor technique effect shows the effect of change in the income level on the level of child labor.

²⁴ It is difficult to segregate X+M into different sectors or infer a transition in trade sectors from the evolution of the sector selection probability. A change in the sector selection probabilities may indicate that a transition in trade patterns. Observations on trade flows are not likely to fit well with a specified regression equation that is constant over the observation period. A switching regression model is not appropriate for Krugman (1979) framework, the transition in trade patterns in which a shifting from comparative advantage to product differentiation occurs. Infrequent surveys and data constraints also make this study specific to simple definition of trade openness. Literature on trade and child labor also support this proxy (see also, Doytch et al., 2013; Estevez, 2011; Davies and Voy, 2009 ; Neumayer and De Soysa, 2005)

The empirical investigation within given epistemological domains relies on the findings resulted from theoretical contribution which was conducted previously, in the following ways. First a unique empirical model articulates the discernible effects of trade for SAARC and ASEAN countries which are engaged in production and trade of homogeneous and differentiated goods. Second, the data in the gross domestic product and gross national product can be pursued for both homogeneous and differentiated products effects. Furthermore, this study deciphers the estimates of scale effects (by using gross domestic product) and technique effect (by using net foreign factor income) as estimates that shows the integrated child labor effects of an economy and these effects exist in the production and trade of homogeneous and differentiated products.

4.9 Method

The analysis is carried out using panel data models, for getting a more accurate measurement of whether child labor has curtailed over the period 1999 to 2013 in selected ASEAN and major SAARC countries. Panel data give an increased precision estimation by accreting time periods of data for each individual. For all that, a valid statistical inference can only be achieved by controlling correlation of regression model errors over time period for a given entity or individual.

Many early studies of trade and child labor, relied solely on cross-sectional data due to limitation of data, however, the most recent databases of the International Labor Organization (ILO), International Program on the Elimination of Child Labor (IPEC), The United Nations International Children's Emergency Fund (UNICEF), World Development Indicators (WDI) and Understanding Children's Work (UCW) has enough information for child labor, therefore the trend is now altering and the stream of studies is rotating around panel data methods. The panel data allow the dynamics of individual country behavior as compared to cross section assortment at one specific point in time. While simultaneously, panel data analysis allows consistent estimates for fixed effect model, in conjunction with consistency, unobserved country specific heterogeneity, which sometime correlated with regressors, can be resolved in a fixed effects model. However, contrary to this expectation, if unobserved country specific effects are assumed to be distributed randomly of regressors, random effects model implies additional orthogonality.

In order to ensure a valid statistical inference, the panel-robust statistical inferences are corrected for both the correlation of errors over time for major SAARC and selected ASEAN countries and heteroskedasticity across these countries. The Hausman specification test and Sargan-Hansen test²⁵ is used to check whether fixed or random effects model is more appropriate.

²⁵ Sargan-Hansen test is an extension of Hausman test, with cluster robust version to robust to heteroskedasticity and within group correlation (Schaer and Stillman, 2010)

Table 12: Panel Data Models with a Small Number of Countries.

Static Models	
Fixed effects (FE)	The fixed-effects model controls for all time-invariant differences between the
	individuals, so the estimated coefficients of the fixed-effects models cannot be biased
	because of omitted time-invariant characteristics.
Random Effects (RE) The random effects model is used when some omitted variables are constant	
	but vary between counties, and others may be fixed between countries but vary over
	time. RE estimator is a weighted average of fixed and between effects.

Fixed-Effect Versus Random-Effects Models

The Hausman test checks a more efficient model against a less efficient but consistent model to make sure that the more efficient model also gives consistent results.

Dynamic Models	
Mean Group (MG)	Mean Group estimator (MG) which is based on estimating N time-series regressions and
	averaging the coefficients (Pesaran and Smith, 1995). It consists of estimating separate
	regressions for each country and computing averages of the country-specific coefficients,
	which will provide consistent estimates of the long-run coefficients. The least restrictive
	procedure and it allows for heterogeneity of all the parameters (imposes no cross-country
	restriction). Does not take account of the fact that certain parameters may be the same
	across groups.
Dynamic Fixed Effect	Dynamic Fixed Effect imposes homogeneity of all slope coefficients, allowing only the
(DFE)	intercepts to vary across countries
Pooled Mean Group	PMG estimator which is a combination of pooling and averaging of coefficients (Pesaran
(PMG) ²⁶	et al., 1999). Intermediate estimator (between DFE and MG). Allows the intercepts,
	short-run coefficients and error variances to differ freely across groups, but constrains
	the long-run coefficients to be similar across groups.

Note: This study is based on static models, to present a comparative statics analysis of the effects of an increase in the trade intensity which generates trade-induced scale, technique and selection effects.

Source: Hsiao (2014)

²⁶ A basic requirement of the PMG estimator is that it T should be large (Pearson, Shin and Smith 1998). Using Monte Carlo experiment, it can be shown that the Mean Group (MG) estimator is consistent but it is unlikely to be a good estimator when either N or T is small (Pearson, Shin and Smith 1998). Since the basic assumption of the PMG estimator has been violated, hence the results cease to be reliable (Hsiao et al., 1998).

4.10 Unobservable Variables

Unobservable parameters such as the exogenous demographic attributes, labor market attachments, education system and child labor preference in the production process (as a compliment or substitute) can be considered as time invariant country specific effects represented by the unobserved heterogeneity, denoted as ε_i .

It is also noted that common-to-all-countries effects such as relative income idiosyncrasies and unconventionalities in amelioration process and dynamic technological improvements, may be considered as time-specific effects, denoted as \beth_t . Machine error in reading child labor concentrations and human error in the calculation is a source of idiosyncratic error represented by μ_{it} . To expound the unobserables and individual effects of the model are specified in the following way:

$$u_{it} = \varepsilon_i + \beth_t + v_{it}$$

 \beth_t is a time-specific effect, ε_i is a country specific effect, and v_{it} idiosyncratic measurement error for country at time. The most familiar fixed effects and random effects panel data treatments for count data are suggested by Hausman et al (1984). An alternative of Hausman test is the Sargan-Hansen test (Schaffer and Stillman, 2011), the Sargan Hansen test is

used cluster-robust standard error as it allows the determination of the most appropriate estimator between the fixed and random effects models.

If country-specific effects contained in the unobserved random component, which may be a case of many practical applications in international economics, are correlated with predictors, then fixed effects estimator allows for consistent estimation of the model. In contrast to the fixed effects model, sometime the unobserved country-specific heterogeneity is distributed randomly of the predictors, in this case, consistent and efficient estimation contingent upon the model of the random effects estimator. Note that the pooled ordinary least squares (OLS) estimator is inconsistent if the true model is the fixed effect model. Pooled ordinary least squares (OLS) estimator is inconsistent, while the pooled OLS is consistent but less efficient where the random effects model is more appropriate.

CHAPTER 5

EMPIRICAL RESULTS

5.1 Chapter Overview

This chapter is a presentation of empirical results, which evaluates the prospective of trade induced effects on child labor in the SAARC and ASEAN countries. Estimating equations, based on the predictions of the theoretical framework refers to the previous chapter, explain the various effects of trade. This chapter consists of descriptive statistics, discussion of empirical results and alternative specifications for SAARC and ASEAN countries.

5.2 Descriptive Statistics

The descriptive statistics of the variables employed are reported in Table 13. Total number of observations used in the empirical analysis, means, standard deviation, minimum and maximum values of variables provide a useful information of child labor and other explanatory variables. The summary statistics of the variables contains the data from 9 SAARC and ASEAN countries²⁷. The data consists of 15 years (1999 to 2013) with a total number of 135 observations.

²⁷ Major SAARC countries, namely, Bangladesh, India, Pakistan, Nepal, and Sri Lanka and selected ASEAN countries, namely, Cambodia, Indonesia, Philippine, and Thailand.
Variable Name	Variable Code	Mean	Max	Min	Std. Dev.
Child Labor	ςl _{it}	15.91	52.3	1.47	14.47
Selection Effect	SEL _{it}	$1.21e^{-03}$	$4.37e^{-03}$	7.18e ⁻⁰⁵	$1.04e^{-03}$
Scale Effect	SCL _{it}	$3.66e^{+03}$	$8.73e^{+03}$	$7.53e^{+02}$	$1.71e^{+05}$
Technique Effect	TEC _{it}	67.97	178.62	0.26	24.87
Trade	TR _{it}	70.45	150.32	25.54	36.76
Trade Induced Technique Effect	TECTR _{it}	$2.36e^{+03}$	$6.44e^{+03}$	$2.14e^{+02}$	$1.61e^{+03}$
Trade Induced Selection Effect	SELTR _{it}	$1.26e^{+02}$	$7.07e^{+02}$	8.22e ⁺⁰¹	$1.59e^{+02}$
Trade Induced Scale Effect	SCLTR _{it}	$2.46e^{+05}$	$6.64e^{+05}$	$2.04e^{+04}$	$1.71e^{+05}$
Unemployment	UNEP _{it}	4.78	11.9	0.2	2.89
Literacy rate	EDU _{it}	71.32	97.23	32.57	20.27

Table 13: Descriptive Statistics of Variables

Note: Literacy rate measures the percentages of the population, age 15 and above, who can read and write. The unemployment comprise all labor force that is without work but available for and seeking employment.

Based on Table 13, the selection effect (number of listed companies/km²) varies from 7.18e⁻⁰⁵ points to $4.37e^{-03}$ points, while the technique effect (NFFI per capita) varies from 0.26 points to 178.62 points and the scale of the countries (GDP/km²) varies from a low of 753 to a high of 8730 with a mean of 3660. The explanatory variable trade changes between 25.54 percent to 150.32 percent with a mean of 70.45 percent which implies that some economies are more open to trade than others, the trade induced selection effect variable has a minimum of 82.2 and a maximum of 707 with a mean of 1260, while the average of trade induced scale and

technique effects are 246000 and 2360 respectively. The mean of unemployment and literacy rate indices are 4.78 percent and 71.32 percent, respectively.

5.3 Empirical Results

The estimation procedure of this research consists of five steps. In the first step, an exposure assessment of the simple model by capturing selection, scale and technique effects obtained for the child labor in SAARC and ASEAN respectively in closed economy case. In the second step, a variable of trade combined with selection, scale and technique effects variables in both regions to find the effect of change in trade on child labor. Next, in the third step, the synergies of both SAARC and the ASEAN countries examine the selection, scale and technique effects the child labor. Subsequently, the fourth step presents trade interactions to check the trade induced effects across the SAARC and ASEAN countries. As the theoretical framework suggests that the trade induced effects are the key determinants of child labor. Finally, a number of other explanatory variables added to the closed economy and open economy models to investigate the relationship between these variable and child labor in SAARC and ASEAN countries.

5.3.1 Estimation Results for Model A and Model B

This regression for ASEAN countries is based on 60 observations. A low R^2 of models A and B have low predictive value under these circumstances in theory for ASEAN countries. While the regression for SAARC countries is based on 75 observations. While the regression in a set of SAARC plus ASEAN countries is based on 135 observations, the significant improvements are witnessed in R^2 for Models A (0.295) and B (0.298), it shows that a sufficient amount of variation is explained by Model A and Model B.

Table 14 presents initial estimates of selection, scale and technique effects with the help of Model A for autarky and Model B for open economy. In calculating the selection, scale and technique effects with child labor it as assumed that the relative position of selected ASEAN and major SAARC countries in the world remains constant between 1999 to 2013. Consistency across the regressions in the sign of the estimated coefficients reported in the majority of variables for both SAARC and ASEAN countries and t-values reflect the relevance of these variable.

Table 14 indicates that there is a negative; however it is statistically insignificant relationship between child labor and selection effect for SAARC countries and for ASEAN countries, while a statistically significant relationship is found in a set of ASEAN plus SAARC countries for a closed economy (Model A). Furthermore, the results for open economy (Model B) indicates that there is a negative and statistically significant relationship exists between child labor and selection effect for SAARC countries, for ASEAN countries and in a set of ASEAN plus SAARC countries.

Dependent Variable: Child Labor						
	Model A			Model B		
	ASEAN	SAARC	SAARC& ASEAN	ASEAN	SAARC	SAARC& ASEAN
SEL	-0.025	-0.002	-0.007***	-0.036*	-0.007***	-0.0197*
	(-1.34)	(-1.03)	(-3.21)	(-1.88)	(-8.68)	(-1.69)
TEC	-0.004 (-0.75)	-0.001*** (-2.90)	-0.0008* (-1.79)	0.010 (0.54)	-0.005*** (-8.28)	0.0001* (1.75)
SCL	0.010*** (4.01)	0.0039*** (5.65)	0.0062*** (7.56)	0.10*** (6.19)	0.008*** (8.30)	-0.0004*** (-3.33)
TRADE	-	-	-	0.108** (1.97)	0.510*** (4.91)	0.013 (0.26)
Hausman Test	33.16 (0.00)	0.73 (0.000)	6.96 (0.073)	32.89 (0.000)	65.01 (0.000)	3.911 (0.418)
Observations	60	75	135	60	75	135
$\frac{R^2 - overall}{R^2 - overall}$	0.043	0.280	0.295	0.009	0.287	0.298

Table 14: Estimation Results for Model A and Model B

indicating significance at 1-percent

** indicating significance at 5-percent

* indicating significance at 10-percent

SEL: Country specific number of listed companies per squared kilometer.

TEC: Net foreign factor income (NFFI) per capita

SCL: Gross domestic product (GDP) per squared kilometer

TRADE: Import plus export ratio to GDP

Note: $\varsigma l_{it} = \alpha_0 + \beta_1 SEL_{it} + \beta_2 SCL_{it} + \beta_3 TEC_{it} + u_{it} \dots$ Model A $\varsigma l_{it} = \alpha_0 + \beta_1 SEL_{it} + \beta_2 SCL_{it} + \beta_3 TEC_{it} + \beta_4 Trade_{it} + u_{it} \dots$ Model B An insignificant negative relationship between child labor and technique effect is found for ASEAN countries, while there is a negative and statistically significant relationship exists for SAARC countries and in a set of ASEAN plus SAARC countries in Model A. Further investigation reveals that child labor in SAARC countries is less responsive to technique effect as compared to ASEAN countries and empirical evidence shows an insignificant relevance of predictor and dependent variable for ASEAN countries. The linear interaction term of trade induced technique effect is positive in the set of both SAARC and ASEAN countries.

The estimates for Model A indicate that a one unit increase in the scale effect variable raises child labor by about 1% in ASEAN, 0.32% in SAARC and 0.62% in the composition of SAARC and ASEAN countries, by keeping other effects constant. Whereas, the estimates for Model B indicate that a one unit increase in the scale effect variable raises child labor by about 1% in ASEAN, 0.8% in SAARC and 0.04% in the composition of SAARC and ASEAN countries, by holding all other things equal.

Another variable of interest which shows a positive association with child labor is trade variable, trade variable reports that an increase in the trade to GDP ratios raises the child labor. The results in Table 14 reveals that there is a positive and statistically significant association between child labor and trade openness for ASEAN countries, for SAARC countries and in a set of ASEAN plus SAARC countries. In the Model B for child labor, response to the trade variable is statistically significant at the 10 percent level of significance in a separate setting of SAARC and ASEAN countries. Whereas, the estimates of the trade variable are statistically insignificant at the 1, 5 and 10 percent level of significance in a combine set of SAARC plus ASEAN countries. The reason of this weak relevance is that the standard error of trade induced selection effect coefficient is very large, so this statistic is unable to give a significant response.

The results of Table 14 imply a substantial evidence to suggest that the fixed effect model is more appropriate because unobserved country specific heterogeneity is associated with predictors in a separate setting of SAARC and ASEAN countries. While, in combine setting, there is a sufficient evidence to suggest that the estimates derived from random effect estimators are consistent and efficient.

5.3.2 Discussion of the Results

There are two notable features of results for Model A and Model B. One, for econometric model in Table 14, F-test statistics confirm that at the 1 percent level of significance there is sufficient evidence against the null hypothesis that difference in cross country variances is not zero in a separate setting of SAARC and ASEAN countries. These results validate the existence of country specific effect, therefore, pooled ordinary least squares (OLS) estimates are inappropriate in the analysis. Two, in Table 14, the scale, selection and technique effects are used with trade intensity variable and these trade induced variables use demeaned data by removing their sample mean with the exception of selection effect which is taken at steady state values. Consequently, the interpretations of scale and technique variables are made according to the mean value. In a broader sense, Model B explains the coefficient estimates of scale and technique effects which are measured from the average trading country. In other words, this is a normal trading country, a country whose trade intensity is equal to sample mean value.

The selection effect term indicates a negative relationship between the number of listed companies and child labor. The paradox results of selection effect, consider that the reduction in the number of competent firms engaged in the production of differentiated goods can be a source of increase in child labor or an increase in the competent firms can be a source of reduction in child labor in both SAARC and ASEAN countries. These results are consistent for model A and B in all cases.

The theoretical model predicts that at high income levels, the countries have better techniques of production; this in turn implies the labor accumulation fall as development proceeds and finally a significant reduction in child labor. In Model A, the results presents the authentication of this prediction while in Model B, this prediction is verified only for SAARC countries. Based on models A and B of Table 14, this analysis finds a positive relationship between the scale of economic activity as measured by GDP/km² and child labor. The scale effect is consistent with a positive sign in both regions, while in model B, when it measures for both SAARC and ASEAN together, the sign is surprisingly negative; but the magnitude of this effect is very small. So, the effect of scale effect variable on child labor is positive. Hence, a positive scale effect as an increase in this scale of production have a cumulative impact on child labor as previously expected from theoretical predictions.

Relatively simple hypotheses regarding the effect of international trade on child labor investigated by adding measures of openness of trade in Table 14. Theoretical framework in this thesis establishes a negative relationship between trade and child labor, while empirical evidence shows an insignificant relevance of predictor and dependent variable. The results demonstrate that, the trade variable is not significant at conventional levels and signs are also unexpected in a set of SAARC plus ASEAN countries. These findings report that trade openness (greater global integration) may be correlated with an increase in child labor in a set of both regions, although this association is not valid at conventional levels of significance, but there appears to be a positive relationship between trade and child labor. This increasing trend in trade variable probably reflects a labor intensive production technique in major SAARC and selected ASEAN countries, which can be a source of an increase in child labor in these countries. The findings of the research indicate that the positive effect of trade on child labor phenomenon does not match with theoretical expectation. The main reason, why does the relationship change its direction and child labor enter the labor force at higher stages trade openness, is that the factors increasing child labor supply in developing countries have become more effective as a result of integration with global economy as compared to demand side factors. Busse and Wittwer (2001) and Neumayer and De Soysa (2005) provide the evidence that trade openness can be a cause of high incidence of child labor in developing countries. While the trade openness brings modest changes in the magnitude of some of the coefficients for scale and technique effects variables in Table 14. Overall the estimates, given in Table 14, offer an enough evidence for a positive association between openness and child labor.

5.3.3 Estimation Results for Interacted and Non-interacted Models

Table 15 indicates that there is a negative, but statistically insignificant relationship between child labor and selection effect for SAARC countries and for ASEAN countries, while a statistically significant relationship is found in a set of ASEAN plus SAARC countries in noninteracted form (Model A). Furthermore, the results of interacted²⁸ form model (Model C) indicates that there is a positive and statistically insignificant relationship exists between child labor and selection effect for SAARC and ASEAN countries, while a negative and statistically insignificant relationship is found in a set of ASEAN plus SAARC countries.

The results in Table 15 show that there is a negative, but statistically insignificant relationship between child labor and technique effect for ASEAN countries, while there is a negative and statistically significant relationship exists for SAARC countries and in a set of ASEAN plus SAARC countries for a non-interacted model (Model A). However, investigation from interaction terms reveals that child labor in SAARC countries is less responsive to technique effect as compared to ASEAN countries and empirical evidence shows an insignificant relevance of predictor and dependent variable for ASEAN countries. The linear interaction term of trade induced technique effect is positive in the set of SAARC plus ASEAN countries.

²⁸ These interactions terms capture the combined effect of two variables i.e., selection and trade (trade induced child labor selection effect), scale effect and trade (trade induced child labor scale effect) and technique effect and trade (trade induced child labor selection effect), on child labor.

Dependent Variable: Child Labor							
Model A				Model C			
Non interacted			Interacted				
	ASEAN	SAARC	SAARC& ASEAN	ASEAN	SAARC	SAARC& ASEAN	
SEL	-0.025 (-1.34)	-0.002 (-1.03)	-0.007*** (-3.21)	-	-	-	
SELTR	-	-	-	0.423 (0.972)	2.29 (0.149)	-0.0174 (-0.77)	
TEC	-0.004 (-0.75)	-0.001*** (-2.90)	-0.0008* (-1.79)	-	-	-	
TECTR	-	-	-	-0.012 (-0.640)	-0.002*** (-6.53)	0.00005* (1.86)	
SCL	0.010*** (4.01)	0.0039*** (5.65)	0.0062*** (7.56)	-	-	-	
SCLTR	-	-	-	-0.0017 (-0.69)	-0.111* (-1.74)	-0.0004*** (-3.44)	
Hausman Test	33.16 (0.000)	0.73 (0.000)	6.96 (0.073)	45.47 (0.000)	1.52 (0.678)	0.14 (0.86)	
Observations	60	75	135	60	75	135	
R ² Overall	0.04	0.28	0.30	0.08	0.23	0.25	

Table 15: Estimation Results for Interacted and Non-interacted Models

*** indicating significance at 1-percent

** indicating significance at 5-percent

* indicating significance at 10-percent

SEL: Country specific number of listed companies per squared kilometer.

SELTR: Selection Effect interacted with the openness of trade

TEC: Net foreign factor income (NFFI) per capita

TECTR: Technique Effect interacted with the openness of trade

SCL: Gross domestic product (GDP) per squared kilometer

SCLTR Scale Effect interacted with the openness of trade:

Note: $\varsigma l_{it} = \alpha_0 + \beta_1 SEL_{it} + \beta_2 SCL_{it} + \beta_3 TEC_{it} + u_{it} \dots$ Model A

 $\varsigma l_{it} = \alpha_0 + \beta_1 SELTR_{it} + \beta_2 SCLTR_{it} + \beta_3 TECTR_{it} + u_{it} \dots$ Model C

The estimates for the SAARC countries reveals that one unit change in the scale effect, brings 0.010 while this effect is 0.039 units in child labor in the ASEAN countries. Whereas, taking interaction of scale effects brings -0.111 units change with a 90% confidence interval for SAARC countries and it imply -0.001 for ASEAN countries with high standard error. In a set of SAARC plus ASEAN countries this change varies from 0.0062 to -0.0004 units for nonintracted and interacted models respectively.

5.3.4 Discussion of the Results for Interacted and Non-interacted Models

The trade variable in previous section measures the predicted change in child labor for a 1% change in the ratio of exports plus imports to GDP. This measure indicates that a 1 percent change in the share of trade in GDP increase the child labor by 51 percent major SAARC countries and 10.8 percent in the selected ASEAN countries and 1.32 percent in both SAARC and ASEAN countries. On the other hand, the estimates in Table 15 indicate that a change in trade induced effects lowers the child labor. One possible cause for such a change is trade openness. The results may seem different in isolation as the estimates from Table 15 indicate that trade produces the trade induced effects toward child labor. For trade variable a positive sign is traced for the given sample of countries, it is also out of theoretical prediction. Although we have only a sample of countries, it seems reasonable to expect both positive and negative can be expected. As a check on theoretical explanation this thesis examined each country's trade intensity data, surprisingly, both child labor and share of trade in GDP are declining in most of countries of theses regions and due to this reason trade estimates are positive. While the

other findings are roughly consistent with our theory and predicts that trade induced child labor effects can be a disintegrated into scale, selection and technique effects.

Table 15 shows a negative relationship between child labor and selection effect for SAARC, ASEAN and ASEAN plus SAARC countries in non-interacted form. Results of interacted form indicates a positive association between child labor and selection effect for SAARC and ASEAN countries, while a negative relationship is found in a set of ASEAN plus SAARC countries. The results show that, the strategy for identifying the trade related impacts of scale, selection and technique variables is successful for ASEAN and SAARC countries in a separate setting. The data set observe a noticeable heterogeneity in the number of firms for most of the SAARC and ASEAN countries. Therefore, for the selection effect variable this research is using the number of listed companies per square kilometer or it is a firm intensity in the country. For trade induced child labor selection effect, the selection effect is interacted with trade variable.

Neutral technological progress increases the scale of production (Solow, 1956) and creates a positive trade induced scale effect (Krugman, 1979). According to the results hazardous effects of the scale effect is offset by a negative trade induced child labor scale effect. Therefore, estimates indicate that increases in economic activity driven by the trade induced scale effect lower the child labor. Recall that there are great differences among SAARC and ASEAN countries; therefore, the scale effect is measured in the intensive form GDP/km^2 and trade induced child labor scale effect is derived with the interaction of trade variable. As this

study examines that the deviation in the scale of production across countries can be isolated from technique effect. Therefore the recognition that scale should be examined in intensive form due to theoretical restriction. While, the technique effect is measured by net foreign factor income (NFFI) which is the difference between payment to the foreign people and local country people. This isolation reduces the correlation between the scale and technique effect variables.

Only, selection and scale effect exercise is not tests of new trade theory, if surprisingly, child labor is reduced by an increase in the scale of production and competency of firms, then it will be more interesting to observe the effect of trade induced child labor technique effect in these two regions. The results are reassuringly close to what one may have expected with these effects. More speculatively, these thoughts, trials may also offer a possible explanation for child labor in SAARC and ASEAN countries. Note that, the coefficient of trade induced child labor technique effect Table 15 is negative and significantly different from zero in both SAARC and ASEAN regions. On the other hand, in case of a combine data set of SAARC plus ASEAN countries, there is a positive trade induced child labor technique effect. This is a somewhat surprising result because it is predicted that trade has an overall negative technique effect keeping scale and selection effect constant.

The further investigation reveals that child labor in SAARC countries is much less responsive to technique effect as compared to ASEAN countries. The influence of trade induced technique effect is negative with respect to child labor in a separate setting of SAARC and ASEAN countries. This result is consistent with theoretical predictions as it implies that trade induced child labor technique effect is negative. While, in a set of SAARC plus ASEAN countries, the child labor increases due to technique effect in these countries, so the hypothesis of negative trade induced child labor technique effect cannot be accepted in a group of both regions. It is worth noting that the coefficient of trade induced technique effect is insignificant for ASEAN countries and Model C for ASEAN countries has a slightly smaller R-squared. While for SAARC countries coefficient of technique effect is very small. So in a set of SAARC plus ASEAN countries, the different sign can be possible. Another point to be noted here that Cambodia and Nepal have high trade to GDP ratios with a high incidence of child labor as compared to other countries. So, these countries togather in a group of SAARC and ASEAN countries effects the trade induced child labor technique effects.

The linear interaction term of trade induced technique effect is positive in the set of both SAARC and ASEAN countries and while in a separate setting of these regions has a negative association. It shows that, if a country has a relatively low level of production skills and techniques relative to the rest of the world, then keeping all else held constant, the impact of further openness can make these countries vulnerable to children. On the other hand, Estevez and Levy (2014) examine that, if a country has a sufficiently skilled labor and better technique of production relative to the rest of the world, then the impact of trade induced child labor technique effect can make this country a safer place for children.

Several features of result are presented in the Table 15. First, the trade interactions effects confirm the model's basic predictions regarding selection, scale and technique effects. In

particular, the desired signs from trade interacted regressors are obtained, while the significance levels are reduced by the inclusion of the trade interactions in the separate setting of SAARC and ASEAN countries. Second, the coefficient of trade interacted variables has a great influence, whereas in non- interacted form the coefficient looks very small. Third, the sign pattern of the interaction terms is as expected from theory.

These results indicate that trade induced child labor selection, scale and technique effects are still at work determining, but in non-interacted form selection, scale and technique effects considerations is also matter. The majority of signs and statistical significance of the estimates in Table 15 are according to theoretical predictions. Moreover the signs for the selection, scale and technique are plausible, because in most of the cases, these trade induced effects signs lead to a reduction in child labor.

The remaining part of analysis suggests that freer trade is more likely to be good rather than bad for the child labor and is subject to several provisos. This work has several robust maintained assumptions that may be deceitful, because the data on child labor is not perfect, and may be varied on different data source. Clearly much more need to be done along these lines mentions in this section. In order to meet these demands the next section of this thesis present alternative specification test. Overall the results are surprisingly similar to those presented in this section. The main features of this analysis remain intact along with selection, scale and technique effects.

5.4 Alternate Empirical Specifications

To account for the possibilities of robustness to the scale, technique and selection effects, this study proposes an alternative specification in section 5.4.1, by adding education and unemployment. Moreover, section 5.4.2 compares the test results across different groupings of SAARC and ASEAN countries which give a deep understanding of trade induced effects.

5.4.1 Alternative Specification

In section 5.3, this study compares the alternative econometric models regarding to their ability to explain the robustness of the results obtained in the previous section. The alternative specifications are: Model A_2 and C_2 . In the model A_2 the literacy rate is introduced to represent the education and to examine the robustness of other variables. Education variable was ignored in Model A_1 to check the effect of pure demand side factor scale, selection and technique effect, which does not include any interaction terms, specifies an estimating equation that represents the scale, selection and technique effect in close economy. Model C_1 is a simple trade induced child labor model with no trade variable, whereas trade intensity is interacted in Model A_1 variables i.e. scale, selection and technique effects of child labor from other factors are also important for understanding the impact of trade in the given trade and child labor setting. This section briefly summarize the results of models A_1 , A_2 , C_1 and C_2 .

In the models A_1 for child labor, responses to the selection, scale, and technique effects are statistically significant at the 10 percent level of significance. The coefficient estimates of the scale, and selection effects are statistically significant at the conventional level. Whereas, the coefficient estimate of the technique effect is not statistically significant at conventional levels, although a 95 percent level of confidence is a standard level but the response of technique effect is statistically significant at 90 percent confidence level.

In the model A_2 for child labor, responses to the selection, scale, and technique effects are statistically significant at the 10 percent level of significance. Whereas, the coefficient estimate of the variable for educaton is not statistically significant at any level of significance. The Model A_2 with education variable, generate statistically insignificant estimates for the child labor. Strikingly, this analysis finds no significant effects of literacy on child labor reduction. The results obtained from EDU variable are not very robust, so, this variable has a trivial effect on other variables. While, in Model A_1 this variable was ignored and the estimates present better significant impacts. In simple words, the scale, selection and technique's effects in autarky situation are not affected by inclusion or exclusion of education variable. For the trade and child labor analysis, random effects estimation of Model A_2 shows that selection effect variable is statistically significant at the 1 percent level of significance. In the cases of child labor, the sole insignificant estimate is education. An explanation of education in this model gives the lack of evidence to support the theoretical prediction in these cases.

Dependent Variable: Child Labor				
Model Specification:	A_1	A_2	C_1	C_2
SEL	-0.007***	-0.007***		
	(-3.21)	(-2.98)		
TEC	-0.0008*	-0.001*		
	(-1.79)	(-1.70)		
SCL	0.0062***	0.006***		
	(7.56)	(7.38)		
EDU		0.061		
		(0.62)		
SELTR			-0.0174	-0.0139
			(-0.77)	(-0.53)
TECTR			0.00005*	0.00001*
			(1.86)	(1.78)
SCLTR			-0.0004***	-0.0005***
			(-3.44)	(-3.43)
UNEP				-0.0001
				(-0.30)
Breusch-Pagan Test	654.46	655.19	741.89	550.18
-	(0.000)	(0.000)	(0.000)	(0.000)
Hausman Test	6.96	3.02	0.14	1.96
	(0.073)	(0.554)	(0.86)	(0.743)
R ² Overall	0.30	0.37	0.25	0.12
***0				

Table 16: Alternative Specification

***Significance at the 99-percent confidence level.

**Significance at the 95-percent confidence level.

*Significance at the 90-percent confidence level.

SEL: Country specific number of listed companies per squared kilometer.

SELTR: Selection Effect interacted with the openness of trade.

TEC: Net foreign factor income (NFFI) per capita.

TECTR: Technique Effect interacted with the openness of trade.

SCL: Gross domestic product (GDP) per squared kilometer.

SCLTR: Scale Effect interacted with the openness of trade:

EDU: Percentage of the population age 15 and above who can read and write.

UNEP: Share of the labor force that is without work but available for and seeking employment.

Note: $\varsigma l_{it} = \alpha_0 + \beta_1 SEL_{it} + \beta_2 SCL_{it} + \beta_3 TEC_{it} + u_{it} \dots$ Model A₁ $\varsigma l_{it} = \alpha_0 + \beta_1 SEL_{it} + \beta_2 SCL_{it} + \beta_3 TEC_{it} + \beta_4 Edu_{it} + u_{it} \dots$ Model A₂ $\varsigma l_{it} = \alpha_0 + \beta_1 SELTR_{it} + \beta_2 TECTR_{it} + \beta_3 SCLTR_{it} + u_{it} \dots$ Model C₁ $\varsigma l_{it} = \alpha_0 + \alpha_1 SELTR_{it} + \alpha_2 TECTR_{it} + \alpha_3 SCLTR_{it} + \alpha_4 UNEP_{it} + u_{it} \dots$ Model C₂ In model C_1 , only trade-induced effects are estimated. However, and interestingly, the estimates of the selection effect variable are statistically insignificant at the 1, 5 and 10 percent level of significance, the reason of this weak relevance is that the standard error of trade induced selection effect coefficient is very large relative to the statistic, so this statistic is unable to give a significant response. The coefficient of selection has an unexpected negative sign, but without significance. It shows that there is no direct effect of trade induced variables on child labor. While, responses to the scale and technique effects are statistically significant at the 10 percent level of significance.

Model C_1 is extended by including an extra variable unemployment to see the effect of trade induced child labor effects. The new model is Model C_2 and it shows the approximately similar results in term of significance and signs for trade-induced effects i.e. the selection, scale, and technique effects. Hence, this new variable has no significant effect on child labor reduction. The coefficient of unemployment is insignificant and opposite to theoretical explanation. The reason for this unexpected sign and significance is the lack of theoretical support and incongruity for the model which is checking the trade induced child labor effects by using demand side factors for child labor demand while unemployment in Model C_2 address the supply side factors of child labor. That's why the supply side factor is misfit with demand side factor. The remaining part of analysis compares the test results across different groupings of SAARC and ASEAN countries for a deep understanding of trade induced effects.

5.4.2 Robustness of Results

Table 17 shows that in column 1, without Cambodia and Nepal, the selection effect is positive and consistent with the theoretically predicted strategy that a decrease (increase) in the number of firms leads to a decrease (increase) in the child labor. Results in column 1 express that holding all other variables constant, a unit change in the selection variable brings approximately 20 percent positive change in the level of child labor in the same direction for the data set of selected SAARC and ASEAN countries. It means, if the number of firms decrease the child labor will also decrease. It is important to discuss that the purpose selection effect is not as offering universal explanations or predictions, while this effect enhances the understanding of complex child labor situation in literature.

Next, the scale effect is explained by gross domestic product per kilometer square (GDP/km²), the result of scale effect is against the theoretical prediction as it was assumed that an increase in scale of production, increases the demand of child labor due to direct association of labor demand with child labor demand. Estimates of scale effect in models show there is an inverse association between trade induced scale effect and child labor. The coefficient indicates that holding constant all of the other predictors in the model, a unit increase in trade induced scale effects brings 0.01 percent reduction in child labor. It was assumed that this effect can be harmful for child labor, but unpredictably, it is helpful in the child labor reduction. This result strengthens the argument that the increase in scale of production under Krugman framework is beneficial due to demand of skilled labor in the economy.

Dependent Variable: Child labor						
Estimation Equation: $\varsigma l_{it} = \alpha_0 + \beta_1 Sel_{it} + \beta_2 Tec_{it} + \beta_3 Scl_{it} + \beta_4 Trade_{it} + u_{it} \dots$ Model B						
	(1)	(2)	(3)	(4)		
SEL	0.2078***	0.0057	2.014	-0.0197*		
	(3.92)	(0.10)	(0.61)	(-1.69)		
TEC	-0.0001***	0.0001*	0.0001	0.0001*		
	(-2.90)	(1.90)	(0.99)	(1.75)		
SCL	-0.0001***	-0.0007***	-0.0001***	-0.0004***		
	(-4.23)	(-3.80)	(-2.64)	(-3.33)		
Trade	0.369***	0.182***	0.057	0.0132		
	(4.03)	(2.22)	(0.288)	(0.26)		
Breusch–Pagan Test	333.07	245.67	230.67	246.26		
	(0.000)	(0.000)	(0.000)	(0.000)		
Hausman Test	15.86	0.92	1.90	3.911		
	(0.003)	(0.921)	(0.740)	(0.418)		
Observations	105	90	105	135		
R ² Overall	0.31	0.26	0.59	0.29		
Note: Figure in the parentheses are t-statistics, excent for Breusch-Pagan test Hausman test which are n						

Table 17: Robustness of the Results

Note: Figure in the parentheses are t-statistics, except for Breusch–Pagan test, Hausman test, which are p-values.

*, ** and *** indicate the respective 10%, 5% and 1% significance level.

SEL: Country specific number of listed companies per squared kilometer.

TEC: Net foreign factor income (NFFI) per capita

SCL: Gross Domestic Product (GDP) per squared kilometer

TRADE: Import plus export ratio to GDP

Column 1: Includes the SAARC countries, namely, Bangladesh, India, Pakistan, and Sri Lanka and selected ASEAN countries, namely, Indonesia, Philippine, and Thailand.

Column 2: Includes the SAARC countries, namely, Bangladesh, Pakistan, and Nepal, and ASEAN countries, namely, Cambodia, Indonesia, and Thailand.

Column 3: Includes SAARC countries, namely, Bangladesh, Pakistan, Nepal, and Sri Lanka and selected ASEAN countries, namely, Cambodia, Philippine, and Thailand.

Column 4: Includes major SAARC countries, namely, Bangladesh, India, Pakistan, Nepal, and Sri Lanka and selected ASEAN countries, namely, Cambodia, Indonesia, Philippine, and Thailand.

The technique effect is described by the lagged per capita net foreign factor income (NFFI). Regression estimates for column 1 suggests that there is a negative association between technique effect and child labor. In column 1, a one unit increase in technique effect variable can cause up to a 0.01 percent decline in child labor. The t-value (t = -2.90) denotes that the estimated coefficients, for trade induced technique effect, are statistically different from zero at conventional levels of significance.

The trade variable is represented by a trade to GDP ratio. In main model, which assumes linearity in all regressors, the coefficient of trade openness is statistically significant at a one percent level of significance. This result shows a substantial evidence to suggest that a one unit change in trade intensity brings a 36.96 percent positive change in child labor. The estimated coefficient is significantly different from theoretical proposition. Trade variable has unexpected and significant positive sign. This could result from a non-distinction between the trade of differentiated goods and trade of homogeneous goods, similar result has already been reported by Kis–Katos (2006), as she concluded that the real income effect of trade can also increase the aggregate child labor and the impact of trade on the child labor also depends on the techniques of productions with a change in income levels across countries.

The second column in Table 17 shows the set of three countries each from SAARC and ASEAN countries. This set encompasses Pakistan, Bangladesh and Nepal from SAARC region while from the ASEAN region, Indonesia, Cambodia and Thailand are included because on average incidents of child labor is high as compared to other countries. These results show that

the coefficient of selection effect is not significant, however the sign of this variable is positively associated with child labor. The sign of selection effect variable, with respect to child labor, is positive and consistent with the theoretically predicted strategy that a decrease (increase) in the number of firms leads to a decrease (increase) in the child labor.

The next variable in terms of strength of association with child labor is scale effect, it is described by gross domestic product per kilometer square (GDP/km²), The result of scale effect is against the theoretical prediction as it was previously assumed to increase in scale of production, increases the demand of child labor due to direct association of labor demand with child labor demand. Estimates of scale effect in the Model B show a statistically significant negative relationship between scale effect and child labor. The coefficient estimates indicate that holding the all remaining factors constant, a unit increase in scale effect variable brings 0.07 percent reduction in child labor. It was assumed that child labor can increase due to scale effect, while child labor is decreasing due to this effect. This result strengthens the significance that the scale of production in the economy when it is aligned with the openness of trade.

The technique effect shows a positive association with child labor. In column 2, a one unit increase in technique effect variable brings a 0.01 percent rise in child labor. The p-value denotes that the estimated coefficients, for technique effect, are statistically different from zero at the 10 percent level of significance.

The coefficient estimate of trade intensity is statistically significant at the 1 percent level of significance. This result provides a substantial evidence to suggest that a one unit increase in trade intensity variable brings a 18.23 percent increase in child labor. The relationship of trade and child labor receives a considerable theoretical attention because available theoretical and empirical literature on child labor under the new trade theory is not well equipped. The stressing sign of trade variable can reverse the conclusion. Instead of positive effect on child labor, trade may turn the effect into negative one. This result is consistent with the Kis–Katos (2006) findings, as she examine that the real income effect of trade can also increase the aggregate child labor and the impact of trade on the child labor also depends on the techniques of productions with varying income levels across the SAARC and ASEAN countries.

Column 3 of Table 17 shows the responses of changes in the scale, selection and technique effects on the level of child labor, in seven SAARC and ASEAN countries taking three countries from ASEAN and four from SAARC region. The random effects model is found to be more efficient and appropriate in Hausman test. Random effects estimations indicate that there are two opposing theoretical predictions in column 3. Firstly, the technique effect which is statistically insignificant and secondly, the scale effect which is an optimal reverse prediction in favor of child labor reduction activities. This analysis explains that selection effect is a reliable predictor of future real economic activity. The direction of the effect of the selected variable on the child labor is positive, consistent with the theoretically predicted strategy that a decrease (increase) in the number of firms leads to a decrease (increase) in the level of child labor.

predictor in column 3 of Table 17. The only scale effect variable in column 3 shows a statistical significant relevance with child labor. This effect is statistically and economically significant at the conventional significance level (p < 0.05). The sign and statistical significance are consistent with previous research of Edmonds and Pavcnik (2006).

The standard errors of other variables are relatively large and these estimates are not significant at the 5 percent level of significance, therefore, the technique and selection effects are statistically insignificant at conventional levels of significance. Similarly, the association between trade and child labor is not statistically significant. P-values indicate that the null hypothesis cannot be accepted at the 5 percent level of significance. In this analysis, selection and technique variables are not statistically significant; while the scale effect is statistically significant. It would suggest that only scale effect exists, but the remaining selection effect and technique effects are absent.

Table 17 reports the statistically significant estimates for the technique and scale effects in column 4 at the 1 percent level of significance. Both effects have a negative association with child labor. The findings of scale effect endorse the negative relationship between scale and child labor. For the scale effect, the coefficient estimates are statistically significant at the 1 percent level in column 4. There is strong evidence to suggest that a one unit increase in scale of production brings 0.04 percent decrease in child labor. A one unit increase in the trade induced selection effect variable leads to 1.97 percent decrease in child labor in column 4.

Estimates for trade variable is statistically insignificant at conventional levels. More formally, t-test cannot reject the null hypothesis that the coefficient estimate of the trade variable is equal to zero. On the basis of these estimates, there is an absence of trade intensity effect independently. The absence of a trade variable on child labor may be described by the possibility that the variations in the level of trade intensity do not affect child labor directly, but trade is a source of trade induced child labor effect as explained in this trade and child labor framework.

In determining the appropriateness of the estimator i.e fixed effect estimator versus random effect estimator Hausman test and Sargan-Hansen test with cluster-robust standard errors are recommended in this study. The Hausman test, presented in column 1 of Table 17, shows that at conventional levels of significance null hypothesis can be rejected, as it validates the addition orthogonality condition under random effects estimator. In comparison, it is important to note that, statistically significant coefficient estimates are derived from estimations by excluding Cambodia and Nepal under fixed effects estimation in Table 17. The remaining estimates of Table 17 gives a sufficient evidence against fixed effect estimates. Therefore, in all remaining models, there is a sufficient and efficient. The interesting finding in these results is that, statistically insignificant coefficients are obtained on exclusion of India and Indonesia from selected ASEAN and major SAARC countries has adverse effect on estimation results. On the other hand, the estimation of column 2 shows that, there is evidence to suggest that estimates

generated by the random effects models are consistent and efficient. In this case, therefore, in explaining the following regression results, a similar trend is found in full, sample of all selected ASEAN and major SAARC countries, therefore this analysis demonstrates the estimates based on the random effect models.

Table 16 and 17 points two different estimation options i.e. fixed and random effects model with dependent variable child labor. In Table 17, the results of selection, scale and technique effects along with trade intensity are presented. While in the Table 16 a sensitivity analysis of scale, selection and technique effects in both autarky and open economies are presented by including some other variables in estimation equation. Table 16 comprising of four estimating models Model A₁, Model A₂, Model C₁ and Model C₂. Model A₁ and A₂ don't deal with trade intensity separately while in Model C₁ and Model C₂ trade intensity is interacted with selection, scale and technique effects to check the effect of trade induced child labor scale, selection and technique effect. These results suggest that trade induced effects exist in SAARC and ASEAN countries when it is calculated for average country in the sample. The findings of Table 17 and 15 propose the following links of trade and child labor.

First, Selection, scale and technique effects of close economy and trade induced child labor selection, scale and technique effects seemed simply to be correlated with child labor.

Second, there is substantial evidence to support the main idea that the selection effect is an important variable in the treatment of child labor. All models estimated with a number of firms in the close and open economy, it is seen that the main findings are robust with respect to a number of firms (selection effect) and trade induced child labor selection effects. In the models A_1 to C_2 , the coefficient estimates of the selection effect variable are significant in the closed economy model while the trade interacted model this variable lost the significance.

Third, the estimates of the coefficients of scale effect variables in the closed and open economy are statistically significant at all levels of significance. These findings indicate the relative importance of scale effect with respect to significance.

Fourth, The exceptions found the mixed evidence to responses the effects of the technique and scale effects on child labor in the closed and open economy. Comparatively, the coefficient estimates of the effects of scale and technique variables look very small but they are theoretically important and statistically significant for an explanation of the child labor phenomenon.

Fifth, the education and unemployment provide insignificant explanation. Results show that for the child labor, adding new variable, i.e. education and unemployment have no effect on selection, scale and technique effect variables.

Finally, trade intensity variable is not part of the alternative specification, this specification uses the unemployment and education instead of trade variable. The result found

that substituting variable are irrelevant in models A_2 and C_2 . Only the trade variable has a capacity to explain the effect of trade with the help of trade induced child labor effect.

CHAPTER 6

CONCLUSION AND RECOMMENDATIONS

6.1 Chapter Overview

This chapter contains several sections: the first section reiterates the findings and highlights the key results of the research work, followed by discussions on trade induced child labor selection, scale and technique effects. The chapter closes with a discussion of the limitations of the study and policy and recommendations based on the results of this research.

6.2 Summary of the Results

The premise of the analysis explores that the international trade is integrated with the trade of differentiated goods described by the framework of new trade theory and trade of homogenous goods described by the theoretical framework of traditional trade. However, the composite effects of an integrated open economy can be disintegrated into various trade induced effects. In doing so, this dissertation aims to ascertain whether the impact of trade of differentiated products on child labor can be decomposed into selection, scale and technique effects. The past empirical studies, which are based on the traditional trade framework, show that the child labor is linked to the comparative advantage in trade. On the contrary, in this

research the comparative advantage is altered by product differentiation which need a better skills hence less chances of child labor.

The analysis complements the existing literature on the subject of child labor by discussing the role of trade induced child labor effects in Krugman (1979) model. According to new trade theory, a country that starts out with a largely unskilled labor risks being excluded from trade in differentiated goods with the other countries, as the latter form exclusive clubs. Trade openness can thus further impoverish developing countries with particularly unfavorable initial conditions. For all these reasons, child labor is likely to rise. The results seem to be fairly robust to alternate model specifications. The analysis casts doubts about the wisdom of using trade openness to control child labor. Instead, the results suggest that trade induced child labor selection, scale and technique effects are effective in reducing child labor in SAARC and ASEAN countries. However, the analysis could not find empirical evidence that exposure to international trade reduce the incidence of child labor.

Theoretical framework has presented an imperfect competition model of a small developing country to examine the incidence of child labor. A general conclusion which can be drawn from the framework is that the trade will be more effective, if its effects applied at a more disaggregated level. This conclusion has been reached by comparing the different effects of trade on developing countries. Furthermore, the framework suggests that trade openness to combat the child labor can be examined carefully by using trade induced child labor effects. The study also focuses on the mechanisms other than income through which trade might affect child labor that are often the focus of the anti-globalization movement. The analysis finds no support for the view that selection, scale and technique effects instigated by trade are associated with increases in child labor. The intuition looks realistic and when it is investigated solely on SAARC and ASEAN countries.

The results confirm that if firms are engaged in production of differentiated products in an integrated economy, then an empirical estimation of the total impact of trade on the child labor needs to be accounted for a selection effect in addition to scale and technique effects. This has a national and global significance because it highlights the association between the selection effect of the trade and child labor. The ex-ante theoretical realization is admissible with ex post parameters for selection, scale and technique effects. In some measures, expectations for the effects are aligned with theoretical forecast, while in some cases the predictors appear to have an opposite effect from the one intended.

The selection effect is the impact of economic integration with trade driven market structure through product differentiation associated with choice process. Openness of trade not only affects the number of firms (a number of varieties) inside the economy, but also the incidence of child labor across the international borders. A theoretical explanation of selection effect is that, holding the scale and technique effects constant, trade in differentiated goods induces a trade related child labor selection effect i.e. openness to trade implies an access to foreign market, which leads to a change in the number of domestic firms or in the number of product varieties. The change in the number of firms or number of varieties brings a change in child labor or simply it is called a trade induced child labor selection effect.

The results show that, selection effect hypothesis holds true in the case of SAARC and ASEAN countries when Cambodia and Nepal are excluded from the data set. The empirical result in conjunction with the theoretical backdrop gives the consonant sign for the selection effect in the selected group of seven SAARC and ASEAN countries. While in full group of major SAARC and selected ASEAN countries, statistically significant coefficient estimates found mixed evidence, the results are robust across the different empirical models and across different combinations of countries. The empirical estimates confirm the prediction of trade-induced child labor selection effect postulated by the trade framework. In other words, there seems a robust indication to support the hypothesis that the change in the incidence of child labor is due to the selection effect. Furthermore, the current analysis proposes a linear relationship between the selection effect and child labor and the sign of the trade induced child labor framework.

The findings are robust to the selection effect for the trade framework. The results endorse that a change in the number of firms brings a proportionate change in the child labor, holding other things constant. The emancipation of child labor from selection effects is found nearly opposite in their direction. Furthermore, theoretically important outcomes can be described by underlying assumptions of the trade and child labor frameworks which is transformed into econometric models for validation. The theoretical expectations are consistent with the new trade theory, by considering that countries are engaged in the trade of both homogeneous and differentiated goods. An adequate explanation of empirical evidence favors the hypotheses postulated by a theory that child labor is affected by selection effects in an open economy. The estimation shows the impact of international trade on child labor needs to be controlled for the selection effect, simultaneously, the commonly accepted scale and technique effect of international trade. The estimates of selection effect are shown to be robust across five different specifications and across 9 SAARC and ASEAN countries with a different set of combination.

The findings are based on previous work on trade and child labor literature, which seek to resolve the heated debate and serious concern about the international trade. This study departs considerably from much previous work by professeing the estimation of the impact of international trade on child labor which needs to account for a selection effect, a variable that can be favorable to describe the patterns of the exchange of goods across borders. The potential of unbiased estimation lies in the effect to resolve the important issue which may be significant for policy recommendations.

The findings of trade induced child labor scale effect strongly refute the presumption that child labor may increase with the increase in scale of production. In simple words, the scale of production due to trade liberalization is not a hazard for a reduction in child labor. Trade induced child labor scale effect finds a considerate feedback to the treatment of child labor in two ways. First, the main econometric model of the current analysis finds a statistically significant negative relationship between the scale of production and child labor level, which is not consistent with theoretical predictions. However, in the context of child labor, the scale effect can be confounded by the fact that interaction of trade intensity enables this effect to control child labor. The ongoing econometric models for trade and child labor suggest that an increase (decrease) in the level of scale of production lowers (raises) the child labor, holding everything else constant. Second, the responses of the trade induced child labor scale effect variable are statistically significant for the functional form which specifies linearity in the scale variable. These results present the regional evidence on the transformation of economic growth to a child labor reduction in the SAARC and ASEAN countries, with the emphasis on the role of trade induced child labor scale and technique effects. In the context of international trade effects, the evidence suggests that the realization of data has a substantiate assertions that trade driven scale effects play a significant role in falling child labor.

The ongoing analysis finds that there is a lack of statistical evidence for trade openness. These revelations are sufficiently startling to show the need of further investigation. The expected favorable effects of trade liberalization on child labor depend on the Kis-Katos and Sparrow (2007) which has been examined both on theoretical and empirical grounds by many researchers. Rahman and Khanam (2012) found that the theoretical arguments regarding the effects of trade on child labor is ambiguous and empirical evidences also provide us mixed results. The findings of trade openness are not too encouraging. These findings may raise concerns regarding the effect of trade on child labor. From a normative perspective, child labor is largely a phenomenon of poverty, an improvement in the income opportunity due to trade

opportunities can help the developing countries to reduce child labor. The findings in this research corroborate that technique effect is statistically significant for child labor. The regression results, for the countries in the sample, show that estimates of the technique and scale effects can be revealed with the new trade theory framework as contrasted with the traditional trade theory framework. Estimations of the scale and technique effects for the child labor generate mixed findings regarding the inherent characteristics and sources of child labor.

The explanation of favorable effects of trade liberalization on child labor is that when domestic borders opens for foreign countries, it brings better technology in the home country. So, adaptation of less labor intensive technology produced abroad substitute for labor intensive technologies used domestically, thus a reduction in child labor in the home market. A second possible justification is that openness to trade can bring labor amelioration that is embedded with imported production technology. The diffusion of better techniques of production in the domestic market alleviates child labor. Interestingly, these possibilities are inconsistent with the result of positive coefficient estimates of trade, which are statistically significant in the models for child labor. The results show that, the coefficient estimates of the trade intensity variable are consistently positive in signs for all specifications. The results show that international trade has adverse effect on child labor, depending on the available data and speciation of the model.

These results negate the theoretical prediction that greater openness to trade leads the lesser the child labor. The theoretical framework purpose analysis suggests that openness to trade or trade liberalization reduces the child labor. While available raw data rule out the
theoretical evidence. So, the widespread trade and child labor realization need to be decomposed into trade induced child labor selection, scale and technique effects.

Theoretical analysis of child labor in this study develops a theoretical model that clarifies the role of trade of differentiated products on child labor as distinct. There are some unique contributions related to study and use of theory, one, this research proposes an empirical framework for analyzing the trade and child labor relationship which integrates the effects differentiated products on child labor. Two, it tests the theoretical treatment of child labor by using selection, scale and technique variables. More specifically, it tests the selection effect on the child labor level. Three, it investigates the impact of trade on the child labor in SAARC and ASEAN countries, namely, Bangladesh, India, Nepal, Pakistan, Sri Lanka, Cambodia, Indonesia, Philippine, and Thailand. Lastly, it provides an evidence to answer the central question, how the trade induced effects can influence the child labor.

6.3 Recommendations of the Study

Based on the results of the analysis, it looks that trade is harmful for child labor. While trade induced child labor effects are beneficial for child labor. If it is assumed that trade is harmful for child labor, then policy makers recommend trade sanctions on countries. Maskus (1997) and Estevez (2011) provide a detailed discussion of this point and conclude that the sanction has probably counterproductive because sanctions are very likely to harm children rather than help them. Sanctions diminish the life alternatives and newly unemployed children

have to live in a household with a lower family income and less nutrition. In fact, the trade sanctions in developing countries are deceptive, disingenuous and misguided because of political forces. The data of trade to GDP ratios for SAARC and ASEAN countries suggest that trade to GDP ratio is unfavorable. So, it is recommended that these countries should increase the volume of trade to improve the trade to GDP ratio.

This research is not arguing that child labor is good and desirable per se. In fact, it is assumed that in an ideal world, children would not have to work and be able to spend all their time on education. What this research makes clear is that such an ideal world can be afforded only if countries are free to trade. The harsh truth is that comparative advantage of trade is considered as a root causes of child labor, still exist and will continue for the foreseeable future. A complete elimination of the demand for child labor is not realistic (not possible due to some informal sectors of an economy), while the sectors that are engaged in production and trade of differentiated products can be incentivized through the use of targeted policies that takes into account the association between the productivity of firms in a specific sector and its effect on the demand for child labor. However, it is essential to have a reasonable confidence that trade policies put in place will reduce the child labor.

Firms that are engaged in production of differentiated goods need to be incentivized through the use of targeted trade policy. The firms where the productivity gap between skilled (adult workers) and unskilled (child labor) small, trade liberalization can have adverse effects on the demand for child labor. Trade liberalization can be successful, however, in industries where the productivity gap between skilled (adult) and unskilled (child workers) labor is relatively large and the more productive firms see little benefit in employing children. In this scenario, the trade will result in the exit of firms that employ proportionally more children, which will lead to a decrease in the demand for child labor. Child labor penalty, which increase production costs and are generally accompanied by negative welfare effects, can also be used to decrease the profitability of firms that heavily rely on child labor.

6.4 Limitations of the Study

The selection effect is an avenue through which child labor can change following trade liberalization. In particular, if the least-efficient firms are the ones to exit, then the demand of child labor will decrease. Note that the "symmetry" assumption that all firms are of the same size and efficiency (it follows that the exit of some would not automatically change child labor). This assumption was made for analytical convenience, but contradicts the empirical fact that every country has a very wide range of firms operating within it.

The empirical literature on demand side factors of child labor is relatively limited due to the scarcity of reliable data. Unfortunately, it is difficult to obtain firm level data on child labor, which can be beneficial to validate the results and even the survey's reliability can be uncertain due to resistance by firms to disclose the information on the child labor and wages paid to these unlawful workers. Data of adult labor is widely available while the data for economically active children with respect to different age groups and occupation are virtually non-existent. Such information is crucial for calculating the trade induced child labor effects. The lack of perfect data hampered the study's ability to determine the trade induced child labor effects. So, there is an evident need for time series data on child labor.

6.5 Directions for Future Research

A brief review of different trade induced effects suggests that further analysis of the microfoundations of the trade-child labor links is a rich area for future research, shedding further light on the broad linkages empirically examined in this research. The empirical estimates cannot be able to find any direct circumstances of the trade that stimulated higher incidences of child labor that trying to identify such circumstances in the future could be the direction of future research. It is important to discuss that the purpose selection effect is not as offering universal explanations or predictions, while this effect enhances the understanding of complex child labor situation in literature. There may be some legitimate concern with the impact of labor practices on SAARC and ASEAN region workers, which may be equally motivated by humanitarian concerns for children. The selection effect discussed in this dissertation can be equally motivated on moral grounds related to the consumer for SAARC and ASEAN countries is another direction for the future researchers. As a closing comment, it is important to stress the fact that many directions for future research are now open. This study believes that the dynamics of educational achievement must be modeled together with labor market dynamics. This is a topic of great interest.

6.6 Policy Implications

The concluding part finds out that child labor is falling in SAARC and ASEAN region, but it may have a number of informal explanations. The trade induced effects under the new trade theory can implement a range of strategies to improve the compliance. Due to trade induced child labor effects, SAARC and ASEAN countries would no longer need child labor to gain a comparative advantage because most of the countries in both regions can compete on a more level of the playing field. This dissertation sets out the following policy implication which takes into account the validity of the research.

6.6.1 Policy Implication of Scale and Technique Effects

Trade induced child labor effects are allowing the earnest support to reduce the child labor, but they need a further technical support and rigorous empirical evaluation. The question arises which trade induced effect appears to have a greatest reduce the child labor. First and foremost, the empirical evidence points out the role of trade induced child labor technique and scale effects in giving rise to the elimination of inefficient child labor. From a theoretical perspective, countries without access to better technique of production may not be able to reduce the child labor. Estevez and Levy (2014) also examine that advancement in the technique of production is biased toward skilled workers, and less chances of unskilled labor or child labor in the production of differentiated goods. The results of this study suggest that SAARC and ASEAN countries should have a sufficiently skilled labor and better technique of

production relative to the rest of the world, then the impact of trade induced child labor technique effect will make this country a safer place for children.

The economies of scale (scale effect) have a centrifugal effect towards the child labor, tends to push economic activity into spreading out and would be opposed by the centripetal pull of access to world markets, which tend to promote concentration of economic activity. In the production of differentiated products the economy continues to grow and the supply of low skilled jobs dries up and sustained growth will result in an increase in high skilled jobs in all sectors. This, in turn, will decrease child employment. It is therefore expected that the impact of growth will lead to a shift toward more skilled workers and less child labor. So, this study recommends that the increase in scale of production need to be continue in SAARC and ASEAN countries.

6.6.2 Policy Implications of Selection Effect

In general the child labor enforcement strategies draw resources from other productive activities in the economy. These resources might be better devoted to other strategies with the help of trade induced child labor selection effect. This study suggests that the aggregate welfare of trade induced child labor selection effect may be higher by allowing some child labor to exist rather than seeking for complete elimination of child labor. Progress toward the elimination of child labor can be made with a fall in the number of firms. This analysis also finds out that an additional specialization in the form of differentiated products can be a source of trade between

the countries. The differentiation requires better labor skills and techniques. So, it nullifies the role of unproductive labor practices (child labor) and the policy implication of selection effect is fairly subtle. So, a successful product differentiation strategy is recommended to reduce the child labor in SAARC and ASEAN countries.

6.6.3 Policy Implication of Trade Openness

Indeed, trade liberalization improves the earnings opportunities for local citizen of countries and trade liberalization raises the country's exposure to international competition. In this way inefficient firms in import competing industries may be forced out of competition in business. Regardless of the long run, gains from the re-allocation of resources, in the short run, these arrangements may be a sterilizing factor for child labor treatment. So, it is recommended that trade facilities in SAARC and ASEAN should be flourished.

Finally, this research proposes that the trade induced child labor effects can be worthwhile to address the underlying economies that gives a rise to offend child labor practices. As discussed previously, trade sanctions arguably lie at the heart of the most egregious forms of child labor exploitation in developing countries. If these countries have an easy access to global markets, they would be able to enhance the scale of production and finally it improves production technique due to income improvements. The end result, of course, is adequate child labor reduction.

REFERENCES

- Abduh, M., & Omar, M. A. (2012). Islamic Banking and Economic Growth: The Indonesian Experience. *International Journal of Islamic and Middle Eastern Finance and Management*, 5 (1), 35-47.
- Abebe, T., & Bessell, S. (2011). Dominant Discourses, Debates and Silences on Child Labour in Africa and Asia. *Third World Quarterly*, 32 (4), 765-786.
- Acaroglu, H., & Dagdemir, O. (2010). The Effects of Globalization on Child Labor in Developing Countries. Business and Economic Horizons, 2, 37-47.
- Ahmed, A., Aziz, A., & Zaman, K. (2013). The Long Run Effects between Unemployment and Economic Growth in Selected SAARC Countries. *The Economic Research Guardian*, 3 (2), 2-17.
- Ahmed, S., Ahmed, V., & Sohail, S. (2010). Trade Agreements between Developing Countries: A Case Study of Pakistan-Sri Lanka Free Trade Agreement. University Library of Munich, Germany, MPRA_paper_29209. Retrieved from http:// http://mpra.ub.unimuenchen.de/29209/1/MPRA paper 29209.pdf.
- Aizenman, J., Jinjarak, Y., & Park, D. (2011). International Reserves and Swap Lines: Substitutes or Complements? *International Review of Economics & Finance*, 20 (1), 5-18.
- Akabayashi, H., & Psacharopoulos, G. (1999). The Trade-Off between Child Labour and Human Capital Formation: A Tanzanian Case Study. *The Journal of Development Studies*, 35 (5), 120-140.
- Akbar, A., Wicaksono, B., & Dachlan, E. G. (2012). Maternal Mortality and Its Mainly Possible Cause Pre-Eclampsia/Eclampsia In Developing Country (Surabaya-Indonesia as the model). Pregnancy Hypertension. An International Journal of Women's Cardiovascular Health, 2 (3), 184.
- Akhter, N., & Ghani, E. (2010). Regional Integration in South Asia: An Analysis of Trade Flows Using the Gravity Model. *The Pakistan Development Review*, 49(2), 105-118.
- Akram, A. (2013). *PAK-SAARC Intra-Industry Trade* (No. 2013: 93). Pakistan Institute of Development Economics. Retrieved from http://72.9.146.122/pdf/working%20paper/working paper-93. pdf.
- Akram, A., & Mahmood, Z. (2012). Determinants of Intra-Industry Trade between Pakistan and Selected SAARC Countries. *The Pakistan Development Review*, 51 (1), 47-59.
- Aktar, S., & Abdullah, A. S. M. (2013). Protecting Child Labour in Bangladesh: Domestic Laws versus International Instruments. *Bangladesh e-Journal of Sociology*, 10 (1), 153.
- Alam, S., Rehman, S., & Butt, M. S. (2011). Trade Liberalization, Environmental Degradation and Sustainable Development in Pakistan. European Journal of Social Sciences, 19(1), 84-96.

Alan, M. (2011). Imperfect Competition in the Labor Market. Handbook of Labor Economics, 4, 973-1041.

- Alchian, A. A., & Demsetz, H. (1972). Production, Information Costs, and Economic Organization. The American Economic Review, 62 (5), 777-795.
- Aldaba,F., Lanzona, L., and Tamangan, R. (2003). A National Policy Study on Child Labour and Development in the Philippines. Makati City: Philippine Institute for Development Studies and the International Labour Organization. Retrieved from http://www.ilo.org/public/english/region/asro/manila/ipec/info/studies/cl and development in rp.pdf.
- Amestoy, M., & Crosbie, M. (2000). Child Labor in the Orange Juice Industry in Brazil. Business Students Focus on Ethics, 8, 151-163.
- Amjad, R., Arif, G. M., & Mustafa, U. (2008). Does the Labor Market Structure Explain Differences in Poverty In Rural Punjab? *The Pakistan Development Review* 36 (1), 39-68.
- Anderson, B. (2000). Doing the Dirty Work? The Global Politics of Domestic Labour. Palgrave Macmillan.
- Andvig, J. C. (1998, January). Child Labour in Sub-Saharan Africa an Exploration. *Forum for Development Studies*, 25 (2), 327-362.
- Ang, D., Brown, D., Dehejia, R., & Robertson, R. (2012). Public Disclosure, Reputation Sensitivity, and Labor Law Compliance: Evidence from Better Factories Cambodia. Review of Development Economics, 16 (4), 594-607.
- Anker, R. (2000). The Economics of Child Labour: A Framework for Measurement. *International Labour Review*, 139(3), 257-280.
- Ansari, M. I. (1992). Growth effects of Recent Structural Changes in the Canadian Economy: Some Empirical Evidence. *Applied Economics*, 24 (11), 1233-1240.
- Aralas, S. B. (2010). Essays in Trade and Environment: The Environmental Effects of Intraindustry Trade. Michigan State University.
- Asal, V., Fair, C. C., & Shellman, S. (2008). Consenting to a Child's Decision to Join a Jihad: Insights From a Survey of Militant Families in Pakistan. *Studies in Conflict & Terrorism*, 31 (11), 973-994.
- Aspinall, E., & Berger, M. T. (2001). The Break-Up of Indonesia? Nationalisms After Decolonisation and the Limits of the Nation-State in Post-Cold War Southeast Asia. *Third World Quarterly*, 22 (6), 1003-1024.
- Athukorala, P. C. (2011). Production Networks and Trade Patterns in East Asia: Regionalization or Globalization? *Asian Economic Papers*, 10 (1), 65-95.
- Averett, S. L., Peters, H. E., & Waldman, D. M. (1997). Tax Credits, Labor Supply, and Child Care. *Review* of Economics and Statistics, 79 (1), 125-135.

- Azmat, G., & Petrongolo, B. (2014). Gender and the Labor Market: What Have We Learned from Field and Lab Experiments?, IZA Discussion Paper No.8373. Retrieved from http://www.econstor.eu/bitstream/10419/96811/1/dp8135.
- Baker, R., Panter-Brick, C., & Todd, A. (1997). Homeless Street Boys in Nepal: Their Demography and Lifestyle. *Journal of Comparative Family Studies*, 28, 129–146.
- Baland, J. M., & Robinson, J. A. (2000). Is Child Labor Inefficient? *Journal of Political Economy*, 108 (4), 663-679.
- Balassa, B. (1979). Intra-industry Trade and the Integration of Developing Countries. In the World Economy; in: Herbert Giersch, ed., On the Economics of Intra-Industry Trade (Mohr, Tubingen),3, 245-270.
- Balassa, B. (1979). The Changing Pattern of Comparative Advantage in Manufactured Goods. *The Review of Economics and statistics*, 61, 259–266.
- Balassa, B. (2013). The Theory of Economic Integration. London: Routledge.
- Baldwin, R. E., & Forslid, R. (2010). Trade Liberalization with Heterogeneous Firms. *Review of Development Economics*, 14 (2), 161-176.
- Bales, K. (2012). *Disposable people: New Slavery in the Global Economy*. California: University of California Press.
- Banerjee, R., & Nag, R. N. (2013). Globalization, Child Labour and Development Policies: A Theoretical Analysis. Foreign Trade Review, 48 (1), 83-104.
- Basri, M. C., & Hill, H. (2008). Indonesia–Trade Policy Review 2007. *The World Economy*, 31 (11), 1393-1408.
- Basu, A. K., Chau, N. H., & Grote, U. (2006). Guaranteed Manufactured without Child Labor: The Economics of Consumer Boycotts, Social Labeling and Trade Sanctions. *Review of Development Economics*, 10 (3), 466-491.
- Basu, K. (2002). A Note on Multiple General Equilibria with Child Labor. *Economics Letters*, 74(3), 301-308.
- Basu, K., & Tzannatos, Z. (2003). The Global Child Labor Problem: What Do We Know and What Can We Do? *The World Bank Economic Review*, 17 (2), 147-173.
- Basu, K., & Van, P. H. (1998). The Economics of Child Labor. American Economic Review, 88 (3), 412-427.
- Basu, K., & Zarghamee, H. (2009). Is Product Boycott a Good Idea for Controlling Child Labor? A Theoretical Investigation. *Journal of Development Economics*, 88 (2), 217-220.
- Basu, K., Das, S., & Dutta, B. (2010). Child labor and Household Wealth: Theory and Empirical Evidence of an Inverted-U. *Journal of Development Economics*, 91 (1), 8-14.

- Batra, A. (2006). India's Global Trade Potential: The Gravity Model Approach. *Global Economic Review*, 35 (3), 327-361.
- Baum, C. F., Schaffer, M. E., & Stillman, S. (2007). Enhanced Routines for Instrumental Variables/GMM Estimation and Testing. *Stata Journal*, 7(4), 465-506.
- Bayangos, V., & Jansen, K. (2011). Remittances and Competitiveness: The Case of the Philippines. World Development, 39 (10), 1834-1846.
- Becker, M. L. (2014). Circumstances, Experiences and Processes Surrounding Women's Entry Into Sex Work in India. *Culture, Health & Sexuality*, 16 (2), 149-163.
- Beegle, K., Dehejia, R., & Gatti, R. (2009). Why Should We Care about Child Labor? The Education, Labor Market, and Health Consequences of Child Labor. *Journal of Human Resources*, 44 (4), 871-889.
- Bekele, A., & Boyden, J. (1988). Combating Child Labour. International Labour Organization.
- Ben-Porath, Y. (1980). The F-connection: Families, Friends, and Firms and The Organization of Exchange. *Population and Development Review*, 6, 1-30.
- Berger, M. T. (2003). Decolonisation, Modernisation and Nation-Building: Political Development Theory and the Appeal of Communism in Southeast Asia, 1945–1975. *Journal of Southeast Asian Studies*, 34 (03), 421-448.
- Berik, G., & Rodgers, Y. V. D. M. (2010). Options for Enforcing Labour Standards: Lessons from Bangladesh and Cambodia. *Journal of International Development*, 22 (1), 56-85.
- Bessell, S. (2011). Influencing International Child Labour Policy: The Potential and Limits of Children-Centred Research. *Children and Youth Services Review*, 33 (4), 564-568.
- Best, J. (2012). Damned Lies and Statistics: Untangling Numbers from the Media, Politicians, and Activists. California: Univ of California Press.
- Bhagwati, J. (2001), 'Free Trade and Labor. Retrieved from http://time.dufe.edu.cn/wencong/bhagwati/ft_lab.pdf.
- Bharadwaj, A. (2014). Reviving the Globalization and Poverty Debate: Effects of Real and Financial Integration on the Developing World. *Advances in Economics and Business*, 2 (1): 42-57.
- Bhukuth, A., & Ballet, J. (2006). Is Child Labour a Substitute for Adult Labour?: A Case Study of Brick Kiln Workers in Tamil Nadu, India. *International Journal of Social Economics*, 33 (8), 594-600.
- Bigelow, B. (1997). The Human Lives behind the Labels: The Global Sweatshop, Nike, and the Race to the Bottom. *Phi Delta Kappan*, 79 (2), 112-119.

Blau, D. M. (2012). The Child Care Labor Market. Journal of Human Resources, 27 (1), 9-39.

- Bleakley, H., & Lin, J. (2012). Thick-market Effects and Churning in the Labor Market: Evidence from US Cities. *Journal of Urban Economics*, 72(2), 87-103.
- Blunch, N. H., & Verner, D. (2000). Revisiting the Link between Poverty and Child Labor: The Ghanaian Experience. Working Paper. Ghanaian Experience." World Bank Policy Research Working Paper No. 2488. Washington, D.C.: World Bank.
- Boeri, T., & Van Ours, J. (2013). *The Economics of Imperfect Labor Markets*. Princeton: Princeton University Press.
- Bonnal, M. (2010). Export Performance, Labor Standards and Institutions: Evidence from a Dynamic Panel Data Model. *Journal of Labor Research*, 31 (1), 53-66.
- Boockmann, B. (2010). The effect of ILO minimum Age Conventions on Child Labor and School Attendance: Evidence from Aggregate and Individual-Level Data. *World Development*, 38 (5), 679-692.
- Boserup, E., Tan, S. F., & Toulmin, C. (2013). Woman's Role in Economic Development. London: Routledge.
- Brainard, S. L., & Riker, D. A. (1997). *Are US Multinationals Exporting US Jobs?* NBER Working Paper no. 5958. Cambridge, Mass.: National Bureau of Economic Research.
- Brown, D. K., Deardorff, A. V., & Stern, R. M. (2002). *The Determinants of Child Labor: Theory and Evidence*. RSIE Discussion Paper, No. 486, School of Public Policy, University of Michigan.
- Brown, D. K., Deardorff, A. V., & Stern, R. M. (2012). Labor Standards and Human Rights: Implications for International Trade and Investment. *IPC Working Paper Series No. 119, University of Michigan.*
- Brown, D.K., A. Deardorff, and R. Stern (2001): "Child Labor: Theory, Evidence, and Policy," Research Seminar in International Economics. Discussion Paper # 474, University of Michigan.
- Bruton, H. J. (1992). *The Political Economy of Poverty, Equity, and Growth: Sri Lanka and Malaysia:* Oxford/New York: Oxford University Press.
- Busse, M., & Wittwer, G. (2001). Do Labour Standards Affect Comparative Advantage? Evidence for Labour-Intensive Goods. *World Development* 30 (11), 1921-1932.
- Cai, F., & Wang, M. (2010). Growth and Structural Changes in Employment in Transition China. *Journal of Comparative Economics*, 38 (1), 71-81.
- Cameron, A. C., & Trivedi, P. K. (2005). *Microeconometrics: Methods and Applications*. Cambridge University Press.
- Canning, D., & Schultz, T. P. (2012). The Economic Consequences of Reproductive Health and Family Planning. *The Lancet*, 380 (9837), 165-171.
- Casson, M. (2012). Multinationals and World Trade (Routledge Revivals): Vertical Integration and the Division of Labour in World Industries. London: Routledge.

Central Intelligence Agency. (2010). The CIA World Factbook 2010. Skyhorse Publishing Inc.

- Central Intelligence Agency. (2010). *The CIA World Fact Book 2010*. Washington, D.C.: U.S. Government Printing Office.
- Chamley, C. (2012). A Paradox of Thrift in General Equilibrium without Forward Markets. *Journal of the European Economic Association*, 10 (6), 1215-1235.
- Chamsuk, W., Phimonsathien, T., & Fongsuwan, W. (2015). A Structural Equation Model of Factors that Affect the ASEAN Competitive Advantage of the Thai Automotive Parts Industry. *Research Journal of Business Management*,9(1), 218-232.
- Chan, A. (2003). Racing to the Bottom: International Trade without a Social Clause. *Third World Quarterly*, 24 (6), 1011-1028.
- Chand, R. (2010). SAARC Agricultural Vision 2020. Agricultural Economics Research Review, 23(2), 197-208.
- Chaturvedi, S. (1995). Children at Work: A Report on the UNICEF & ILO/IPEC Workshop on Child Labour and Street Children: UNICEF East Asia and Pacific Regional Office.
- Chaudhuri, S. (2011). Labor Market Reform and Incidence of Child Labor in a Developing Economy. *Economic Modelling*, 28 (4), 1923-1930.
- Chen, M. A., Jhabvala, R., & Lund, F. (2002). Supporting Workers in the Informal Economy: A Policy Framework. Geneva: International Labour Office.
- Chongvilaivan, A. (2012). Learning by Exporting and High-Tech Capital Deepening in Singapore Manufacturing Industries, 1974–2006. *Applied Economics*, 44 (20), 2551-2568.
- Cigno, A., Rosati, F. C., & Guarcello, L. (2002). Does Globalization Increase Child Labor? World Development, 30 (9), 1579-1589.
- Ciuriak, D., Lapham, B., Wolfe, R., Collins-Williams, T., & Curtis, J. M. (2011). *Firms in International Trade: Towards a New- New Trade Policy*. Working Paper.
- Claessens, S., Djankov, S., & Lang, L. H. (2000). The Separation of Ownership and Control in East Asian Corporations. *Journal of Financial Economics*, 58 (1), 81-112.
- Clark, R. (2011). Child Labor in the World Polity: Decline and Persistence, 1980–2000. Social Forces, 89 (3), 1033-1055.
- Cochran, V. C. (2013). A Crossroad to Economic Triumph or Terrorism: The Afghanistan-Pakistan Transit Trade Agreement. Global Security Studies, 4 (1), 1-15.
- Collier, P., & Dollar, D. (2002). *Globalization, Growth, and Poverty: Building An Inclusive World Economy*. Oxford University Press.

- Cook, S., & Pincus, J. (2014). Poverty, Inequality and Social Protection in Southeast Asia: An Introduction. *Journal of Southeast Asian Economies (JSEAE)*, 31 (1), 1-17.
- Cooter, R. D., & Siegel, N. (2012). Not the Power to Destroy: An Effects Theory of the Tax Power. Virginia *Law Review*, 98 (6), 1195-1251.
- Copeland, B. R., & Taylor, M. S. (2001). International Trade and the Environment: A Framework For Analysis (No. w8540). National Bureau of Economic Research.
- Copeland, B. R., & Taylor, M. S. (2003). *Trade, Growth and the Environment* (No. w9823). National Bureau of Economic Research.
- Copeland, B. R., & Taylor, M. S. (2013). *Trade and the Environment: Theory and Evidence*. Princeton University Press.
- Cortes, P., & Tessada, J. (2011). Low-Skilled Immigration and the Labor Supply of Highly Skilled Women. *American Economic Journal: Applied Economics*, 3 (3), 88-123.
- Cosimo, Inc.. Islam, T., Raihan, A., & Musharuf Hossain Mollah, M. (2014). Is Bangladesh A Beneficiary of South Asian Free Trade Area (SAFTA)? *Journal of Economics and Development*, 15 (3), 36.
- Counts, E. C. (2002). New Global Estimates on Child Labour. ILO International Programme on the Elimination of Child Labour (IPEC), Geneva.
- Davies, R. B. (2005). Abstinence from Child Labor and Profit Seeking. *Journal of Development Economics*, 76 (1), 251-263.
- Davies, R. B., & Vadlamannati, K. C. (2013). A Race to the Bottom in Labour Standards? An Empirical Investigation. *Journal of Development Economics*. 103, 1-14.
- Davies, R. B., & Voy, A. (2009). The Effect of FDI on Child Labor. *Journal of Development Economics*, 88 (1), 59-66.
- de Bell, L., van Hoestenberghe, K., & Roelfsema, H. (2007) *Trade Liberalization, Regionalism and Child Labor, with an Application to West Africa.* Utrecht: The Netherlands.
- Deacon, B., De Lombaerde, P., Macovei, M. C., & Schröder, S. (2011). Globalisation and the Emerging Regional Governance of Labour Rights. *International Journal of Manpower*, 32 (3), 334-365.
- Deb, P., & Rosati, F. C. (2002). Determinants of Child Labor and School Attendance: The Role of Household Unobservables. *Understanding Children's Work Programme Working Paper, December*.
- Despotis, D. K. (2005). Measuring Human Development via Data Envelopment Analysis: The Case of Asia and the Pacific. *Omega*, 33 (5), 385-390.
- Dessy, S. E., & Pallage, S. (2005). A Theory of the Worst Forms of Child Labour. *The Economic Journal*, 115 (500), 68-87.

- Dewey, J. (2013). *The School and Society and the Child and the Curriculum*. Chicago: University of Chicago Press.
- Dick, H. W., & Rimmer, P. J. (1998). Beyond the Third World City: The New Urban Geography of South-East Asia. Urban Studies, 35 (12), 2303-2321.
- Diller, J. M., & Levy, D. A. (1997). Child labor, Trade and Investment: Toward the Harmonization of International Law. *The American Journal of International Law*, 91 (4), 663-696.
- Dinda, S. (2004). Environmental Kuznets Curve Hypothesis: A Survey. *Ecological Economics*, 49 (4), 431-455.
- Dinopoulos, E., & Zhao, L. (2007). Child Labor and Globalization. *Journal of Labor Economics*, 25(3), 553-579.
- Dixit, A. K., & Stiglitz, J. E. (1977). Monopolistic Competition and Optimum Product Diversity. *The American Economic Review*, 67, 297-308.
- Dixit, A. K., & Stiglitz, J. E. (1979). Monopolistic Competition and Optimum Product Diversity: Reply. *American Economic Review*, 69 (5), 961-963.
- Doepke, M., & Zilibotti, F. (2005). The Macroeconomics of Child Labor Regulation. American Economic Review, 95 (5), 1492-1524.
- Doepke, M., & Zilibotti, F. (2009). International Labor Standards and the Political Economy of Child-Labor Regulation. *Journal of the European Economic Association*, 7 (2-3), 508-518.
- Doepke, M., & Zilibotti, F. (2010). Do International Labor Standards Contribute to the Persistence of the Child-Labor Problem? *Journal of Economic Growth*, 15 (1), 1-31.
- Doran, K. B. (2013). How Does Child Labor Affect the Demand for Adult Labor? Evidence from Rural Mexico. *Journal of Human Resources*, 48 (3), 702-735.
- Dorji, T., & Gyeltshen, T. (2013, March). Demographic and Socio-Economic Factors Associated With Child Labor in Bhutan. In Conference Proceedings of the 2nd International Conference on Population and Development (Vol. 1, No. 1).
- Doytch, N., Mendoza, R. U., & Thelen, N. (2013). The Impact of FDI on Child Labor: Insights from an Empirical Analysis of Sectoral FDI Data and Case Studies. *Asian Institute of Management Working Paper*, (13-006).
- Dufhues, T., Buchenrieder, G., Quoc, H. D., & Munkung, N. (2011). Social Capital and Loan Repayment Performance in Southeast Asia. *The Journal of Socio-Economics*, 40 (5), 679-691.
- Dunning, J. H. (2012). International Production and the Multinational Enterprise (RLE International Business): London: Routledge.

- Durotoye, A. (2014). The MINT Countries as Emerging Economic Power Bloc: Prospects and Challenges. *Developing Country Studies*, 4 (15), 99-106.
- Dustmann, C., Glitz, A., & Vogel, T. (2010). Employment, Wages, and the Economic Cycle: Differences between Immigrants and Natives. *European Economic Review*, 54 (1), 1-17.
- Eckel, C., & Neary, J. P. (2010). Multi-Product Firms and Flexible Manufacturing in the Global Economy. *The Review of Economic Studies*, 77 (1), 188-217.
- Edmonds, E. and Schady, N. (2009). "Poverty Alleviation and Child Labor". Policy Research Working Paper 4702, World Bank, Washington DC.
- Edmonds, E. V. (2005). Does Child Labor Decline with Improving Economic Status? *Journal of Human resources*, 40(1), 77-99.
- Edmonds, E. V. (2006). Child Labor and Schooling Responses to Anticipated Income in South Africa. *Journal of Development Economics*, 81 (2), 386-414.
- Edmonds, E. V. (2007). Child Labor. Handbook of Development Economics, 4, 3607-3709.
- Edmonds, E. V., & Pavcnik, N. (2005). Child Labor in the Global Economy. *The Journal of Economic Perspectives*, 19 (1), 199-220.
- Edmonds, E. V., & Pavenik, N. (2005). The Effect of Trade Liberalization on Child Labor. *Journal of International Economics*, 65 (2), 401-419.
- Edmonds, E. V., & Pavcnik, N. (2006). International Trade and Child Labor: Cross-Country Evidence. Journal of International Economics, 68 (1), 115-140.
- Edmonds, E. V., & Schady, N. (2009). Poverty alleviation and child labor. *American Economic Journal: Economic Policy*, 4(4), 100-124.
- Edmonds, E. V., & Shrestha, M. (2012). The Impact of Minimum Age of Employment Regulation on Child Labor and Schooling. *IZA Journal of Labor Policy*, 1 (1), 1-28.
- Edmonds, E. V., & Turk, C. (2002). Child Labor in Transition in Vietnam (Vol. 2774). World Bank Publications.
- Edmonds, E.V. and Schady, N. (2008). 'Poverty Alleviation and Child Labor', *Policy Research Working Paper*, No. 4702. Washington, DC, World Bank.
- Edmonds, Eric (2010), 'Trade, Child Labor, and Schooling in Poor Countries,' in G. Porto and B. Hoekman, eds., Trade Adjustment Costs in Developing Countries: Impacts, Determinants, and Policy Responses, The World Bank Press, 179-196.
- Edmonds, E. V. (2015). Economic Growth and Child Labor in Low Income Economies. Retrieved from: http://glm-lic.iza.org/file/view/publications/2015-0814edmonds_izadraft.pdf.

- Egger, H., Egger, P., & Greenaway, D. (2007). Intra-Industry Trade with Multinational Firms. *European Economic Review*, 51 (8), 1959-1984.
- Elson, D., & Pearson, R. (1981). 'Nimble Fingers Make Cheap Workers': An Analysis of Women's Employment in Third World Export Manufacturing. *Feminist Review*, 7 (1), 87-107.
- Emerson, P. M., & Souza, A. P. (2011). Is Child Labor Harmful? The Impact of Working Earlier in Life on Adult Earnings. *Economic Development and Cultural Change*, 59 (2), 345-385.
- Emmerson, D. K. (1984). "Southeast Asia": What's in A Name? *Journal of Southeast Asian Studies*, 15(01), 1-21.
- Engels, F. (1892). The Condition of the Working Class in England, from Personal Observation and Authentic Sources. Reprinted 1969. London: Panther.
- Eskeland, G. S., & Harrison, A. E. (2003). Moving to Greener Pastures? Multinationals and The Pollution Haven Hypothesis. *Journal of Development Economics*, 70 (1), 1-23.
- Estevez, K. (2010). *Essays on Child Labor, Productivity, and Trade* (Doctoral dissertation). University of Florida, Florida.
- Estevez, K. (2011). Nutritional Efficiency Wages and Child Labor. Economic Modelling, 28(4), 1793-1801.
- Estevez, K., & Levy, T. (2014). Intra-industry Trade and the Demand for Child Labor. *International Journal* of Economic Theory, 10(3), 275-294.
- Ethier, W. J. (1982). National and International Returns to Scale in the Modern Theory of International Trade. *The American Economic Review*, 72 (3), 389-405.
- Euwals, R., Knoef, M., & Van Vuuren, D. (2011). The Trend in Female Labour Force Participation: What Can be Expected for the Future? *Empirical Economics*, 40 (3), 729-753.
- Evers, H. D. (1987). The Bureaucratization of Southeast Asia. *Comparative Studies in Society and History*, 29 (04), 666-685.
- Fallon, P., & Tzannatos, Z. (1998). Child Labor: Issues and Directions for the World Bank. World Bank, Washington, D.C.
- Fassa, A. G. (2003). *Health Benefits of Eliminating Child Labour*. Geneva, Switzerland: International Labour Office.
- Faustino, H. C., & Leitão, N. C. (2011). Fragmentation in the Automobile Manufacturing Industry: Evidence from Portugal. *Journal of Economic Studies*, 38 (3), 287-300.
- Feenstra, R. C. (1998). Integration of Trade and Disintegration of Production in the Global Economy. *The Journal of Economic Perspectives*, 12 (4), 31-50.

- Feenstra, R. C. (2003). Advanced International Trade: Theory and Evidence. Princeton: Princeton University Press.
- Fehr, E., Kirchler, E., Weichbold, A., & Gächter, S. (1998). When Social Norms Overpower Competition: Gift Exchange in Experimental Labor Markets. *Journal of Labor economics*, 16 (2), 324-351.
- Felbermayr, G., & Prat, J. (2011). Product Market Regulation, Firm Selection, and Unemployment. *Journal of the European Economic Association*, 9 (2), 278-317.
- Felman, J., Gray, S., Goswami, M., Jobst, A. A., Pradhan, M., Peiris, S., & Seneviratne, D. (2014). ASEAN-5 Bond Market Development: Where Does It Stand? Where is it Going? *Asian-Pacific Economic Literature*, 28 (1), 60-75.
- Feridun, M., & Sissoko, Y. (2011). Impact of FDI on Economic Development: A Causality Analysis for Singapore, 1976–2002. International Journal of Economic Sciences and Applied Research, (1), 7-17.
- Foldvari, P., van Leeuwen, B., Marks, D., & Gall, J. (2013). Indonesian Regional Welfare Development, 1900–1990: New Anthropometric Evidence. *Economics & Human Biology*, 11 (1), 78-89.
- Fors, H. C. (2012). Child Labour: A review of Recent Theory and Evidence With Policy Implications. *Journal of Economic Surveys*, 26 (4), 570-593.
- Francois, J. F. (1990). Trade in Producer Services and Returns due to Specialization under Monopolistic Competition. *Canadian Journal of Economics*, 23 (1), 109-124.
- Fraser, E. D., Mabee, W., & Slaymaker, O. (2003). Mutual Vulnerability, Mutual Dependence: The Reflexive Relation between Human Society and the Environment. *Global Environmental Change*, 13 (2), 137-144.
- Freeman, R. B. (2000) Single Peaked vs. Diversified Capitalism: The Relation between Economic Institutions and Outcomes, NBER Working paper 7556, NBER, Cambridge, MA.
- Fung, K. C., & Maechler, A. M. (2007). Trade Liberalization and the Environment: The Case of Intra-Industry Trade. *The Journal of International Trade & Economic Development*, 16(1), 53-69.
- Galor, O. (2012). The Demographic Transition: Causes and Consequences. *Cliometrica*, 6 (1), 1-28.
- Gamlin, J., Camacho, A. Z., Ong, M., & Hesketh, T. (2015). Is domestic work a worst form of child labour? The findings of a six-country study of the psychosocial effects of child domestic work. *Children's Geographies*, 13(2), 212-225.Gandolfo, G. (2014). The Models. In International Trade Theory and Policy. *Springer Berlin Heidelberg*, 2,177-212.
- Gautam, B. P. (2011). Tourism and Economic Growth in Nepal. NRB Economic Review, 23 (2), 18-30.
- Gencer, E. A. H., & Anderson, W. P. (2013). An Intra-Industry Trade Model in a Vertical Differentiation Framework. *The Annals of Regional Science*, 52 (1), 1-26.

- Gereffi, G. (1999). International Trade and Industrial Upgrading in the Apparel Commodity Chain. Journal of International Economics, 48 (1), 37-70.
- Giampietro, M., & Pimentel, D. (1993). The Tightening Conflict: Population, Energy Use and the Ecology of Agriculture. NPG Forum Series. Teaneck, NJ: Negative Population Growth. Retrieved from http://www.npg.org/forum_series/tightening_conflict.htm.
- Gill, I. S., Kharas, H. J., & Bhattasali, D. (2007). *An East Asian Renaissance: Ideas Fsor Economic Growth*. Washington, DC: The World Bank.
- Glut, T. A. (1995). Changing the Approach to Ending Child Labor: An International Solution to an International Problem. *Vand. J. Transnat'l L.*, 28, 1203-1210.
- Gödecke, T., & Waibel, H. (2011). Rural–Urban Transformation and Village Economy in Emerging Market Economies During Economic Crisis: Empirical Evidence from Thailand. *Cambridge Journal of Regions, Economy and Society*, 4 (2), 205-219.
- Goldberg, P. K., & Pavcnik, N. (2004). Trade, Inequality, and Poverty: What Do We Know? Evidence from Recent Trade Liberalization Episodes in Developing Countries. *Journal of Economic Literature*, 45 (1), 223–269.
- Grootaert, C., & Kanbur, R. (1995). Child Labour: An Economic Perspective. *International Labor Review.*, 134(2), 187-203.
- Grossmann, H., & Michaelis, J. (2007). Trade Sanctions and the Incidence of Child Labor. *Review of Development Economics*, 11 (1), 49-62.
- Grubel, H. G., Lloyd, P. J., & Lloyd, P. J. (1975). Intra-Industry Trade: The Theory and Measurement of International Trade in Differentiated Products (Vol. 12): London, Macmillan.
- Guarcello, L., Mealli, F., & Rosati, F. C. (2010). Household Vulnerability and Child Labor: The Effect of Shocks, Credit Rationing, and Insurance. *Journal of Population Economics*, 23 (1), 169-198.
- Hafner-Burton, E. M. (2013). Forced to be Good: Why Trade Agreements Boost Human Rights. Ithaca, NY: Cornell University Press.
- Hagestad, G. O., & Uhlenberg, P. (2006). Should We Be Concerned about Age Segregation? Some Theoretical and Empirical Explorations. *Research on Aging*, 28 (6), 638-653.
- Hall, B. L., Kulig, J., & Kalischuk, R. G. (2011). Rural Youth and Violence: A Gender Perspective. *Rural & Remote Health*, 11 (3), 1716.
- Hall, J. (2013). Introduction. In Children's Human Rights and Public Schooling in the United States. *Sense Publishers*, 5, 1-16.
- Hameed, A., & Nazir, A. (2009). Economic Globalisation and Its Impact on Poverty and Inequality: Evidence from Pakistan. *ECO Economic Journal*, 1 (1), 1-21.

- Hamermesh, D. S. (1987). *The Demand for Labor in the Long Run*. In O. Ashenfelter and R. Layard, eds., Handbook of Labor Economics, Amsterdram: North-Holland, 429-71.
- Hamermesh, D. S. (1996). Labor Demand. Princeton, Princeton University Press.
- Hanushek, E. A. (1992). The Trade-off between Child Quantity and Quality. *Journal of Political Economy*, 100 (1), 84-117.
- Hart, H. M. (1936). Processing Taxes and Protective Tariffs. Harvard Law Review, 49, 610-618.
- Hausman, J., B. Hall, and Z. Griliches (1984) "Econometric Models for Count Data and an Application to the Patents-R&D Relationship. Econometrica, 52,909-938.
- Heller, P. S. (1976). Factor Endowment Change and Comparative Advantage: The Case of Japan, 1956-1969. *The Review of Economics and Statistics*, 58 (3), 283-292.
- Helpman, E. (1981). International Trade In the Presence of Product Differentiation, Economies of Scale and Monopolistic Competition: A Chamberlin-Heckscher-Ohlin Approach. *Journal of International Economics*, 11(3), 305-340.
- Helpman, E., & Krugman, P. R. (1985). *Market Structure and Foreign Trade: Increasing Returns, Imperfect Competition and the International Economy:* Cambridge, Mass.: MIT Press.
- Hensman, R. (2013). Workers, Unions, and Global Capitalism: Lessons from India. Columbia: Columbia University Press.
- Hepburn, S., & Simon, R. J. (2013). *Human Trafficking around the World: Hidden in Plain Sight*. New York, Columbia University Press.
- Herath, G. (2012). Child labour in Developing Countries: Review of Theoretical and Empirical Issues. Child Labour in South Asia, 11-25 Retrieved from http://books.google.com.pk/books?hl=en&lr=&id=IkCo4ww8q98C&oi=fnd&pg=PA11&dq=Herath,+G. +(2012).+Child+labour+in+developing+countries:+review+of+theoretical+and+empirical+issues.+Child +labour+in+South+Asia,+11_25.&ots=wrRHUQbhzk&sig=QlDwBZnd7HfMIP6JuWF8WGGfcMQ#v= onepage&q&f=false.
- Herath, G., & Sharma, K. (2012). Child Labour in South Asia. Hampshire: Ashgate Publishing, Ltd.
- Herrera, R. (2011). A Critique of Mainstream Growth, Theory. : Ways Out of the Neoclassical Science (Fiction) and Toward Marxism. *Research in Political Economy*, 27, 3-63.
- Herrmann, M. E. (1991). Looking Down from the Hill: Factors Determining the Success of Congressional Efforts to Reverse Supreme Court Interpretations of the Constitution. *Wm. & Mary L. Rev.*, 33, 543-1341.

- Heymann, J., Raub, A., & Cassola, A. (2013). Does Prohibiting Child Labor Increase Secondary School Enrollment? Insights from a New Global Dataset. *International Journal of Educational Research*, 60, 38-45.
- Hidellage, M. V., & Srivastava, A. (2012). Basic Needs Empowerment. Theme Paper, 11, 22, 25.
- Hindman, H. D. (2009). *The World of Child Labor: An Historical and Regional Survey*. New York: M. E. Sharpe.
- Hindman, H. D. (2011). *The World of Child Labor: An Historical and Regional Survey*. New York: M.E. Sharpe, Inc.
- Hoekman, B., Jensen, J., & Tarr, D. (2013). A Vision for Ukraine in the World Economy. Defining a Trade Policy Strategy that Leverages Global Opportunities (No. PP 2013/25). European University Institute. Retrieved from http://cadmus.eui.eu/bitstream/handle/1814/28961/RSCAS PP 2013 25.pdf?sequence=1
- Hopkins, E. (1994). *Childhood Transformed: Working-Class Children in Nineteenth-Century England*: England, Manchester University Press.
- Hort, S. O., & Kuhnle, S. (2000). The Coming of East and South-East Asian Welfare States. *Journal of European Social Policy*, 10 (2), 162-184.
- Hsiao, C. (2014). Analysis of panel data (No. 54). Cambridge: Cambridge University Press.
- Hughes, A. (2012). Corporate Ethical Trading in an Economic Downturn: Recessionary Pressures and Refracted Responsibilities. *Journal of Economic Geography*, 12 (1), 33-54.
- Hulme, D., Hanlon, J., & Barrientos, A. (2012). Just Give Money to the Poor: The Development Revolution from the Global South. Colorado: Kumarian Press.
- Humphries, J. (2010). *Childhood and Child Labour in the British Industrial Revolution*. Cambridge: Cambridge University Press.
- Hutton, T. A. (2003). Service Industries, Globalization, and Urban Restructuring within the Asia-Pacific: New Development Trajectories and Planning Responses. *Progress in Planning*, 61 (1), 1-74.
- Iftikhar, M. N., Shah, I. A., & Ullah, S. (2010). Impact of Globalization and Economic Reforms on Human Development: A Case Study of Pakistan. *Journal for Global Business Advancement*, 3 (1), 45-59.
- ILO (2002). Every Child Counts: New Global Estimates on Child Labor. Geneva: ILO.
- ILO (2003): Third Synthesis Report on Working Conditions in Cambodia's Garment Sector. Bangkok: ILO.
- ILO (2006). Handbook on Decent Work in the Informal Economy in Cambodia. Bangkok: ILO.
- ILO (2012) ILO Global Estimate of Forced Labour—Results and Methodology. International Labour Office (ILO), Special Action Programme to Combat Forced Labour (SAP-FL), Geneva, Switzerland.

- ILO (1992). World Labour Report 1992. Geneva: ILO.
- ILO-IPEC (2013). Marking Progress Against Child Labour. Global Estimates and Trends 2000- 2012, Geneva, Switzerland.
- ILO (2015). What is child labour. Available at: http://www.ilo.org/ipec/facts/lang--en/index.htm.
- Iqbal, J., Mehmood, Z., & Nosheen, M. (2012). How do Wages and Employment Adjust to Trade Liberalization? A Case Study of Pakistan. *Journal of Business & Economics*, 4 (1), 1-29.
- Iqbal, M. J. (2006). SAARC: Origin, Growth, Potential And Achievements. *Pakistan Journal of History and Culture*, 27(2), 127-140.
- Iram, U., & Fatima, A. (2008). International Trade, Foreign Direct Investment and the Phenomenon of Child Labor: The Case of Pakistan. *International Journal of Social Economics*, 35 (11), 809-822.
- Islam, A., & Choe, C. (2013). Child Labor and Schooling Responses to Access to Microcredit In Rural Bangladesh. *Economic Inquiry*, 51 (1), 46-61.
- Iversen, V. (2002). Autonomy in Child Labor Migrants. World Development, 30 (5), 817-834.
- Jacobs, K., & Graham-Squire, D. (2010). Labor Standards for School Cafeteria Workers, Turnover and Public Program Utilization. *Berkeley J. Emp. & Lab. L.*, 31(2), 447-489.
- Jafarey, S., & Lahiri, S. (2002). Will Trade Sanctions Reduce Child Labour?: The Role of Credit Markets. *Journal of Development Economics*, 68 (1), 137-156.
- Jayanthakumaran, K., Sangkaew, P., & O'Brien, M. (2013). Trade Liberalisation and Manufacturing Wage Premiums: Evidence from Thailand. *Journal of Asian Economics*, 29, 15-23.
- Johnson, D. J., & Bahemuka, J. M. (2013). Vulnerable Childhood in a Global Context: Embracing the Sacred Trust. *Vulnerable Children*, 1,1-10.
- Kambhampati, U. S., & Rajan, R. (2006). Economic Growth: A Panacea for Child Labor? World Development, 34 (3), 426-445.
- Kampan, P., & Tanielian, A. R. (2014). Securing the Future of the Community: Child Protection in ASEAN. Asian Social Science, 10 (11), 172-184.
- Kanungo, A.K. (2012). FDI Inflows into South Asia: A Case Study of India's Investments in Bangladesh. Retrieved from. http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2140737.
- Kaplan, G. (2012). Moving Back Home: Insurance Against Labor Market Risk. *Journal of Political Economy*, 120 (3), 446-512.

Kapstein, E. B. (1996). Workers and the World Economy. Foreign Affairs, 75 (3), 16-37.

- Karlan, D., & Valdivia, M. (2011). Teaching Entrepreneurship: Impact of Business Training on Microfinance Clients and Institutions. *Review of Economics and Statistics*, 93 (2), 510-527.
- Kelsall, T., & Seiha, H. (2014). *The Political Settlement and Economic Growth in Cambodia* (No. esid-037-14). BWPI, The University of Manchester.
- Kenayathulla, H. B. (2012). Cost Benefit Analysis in Malaysian Education. Jurnal Internasional Manajemen Pendidikan, 4(02). Retrieved from. http://pascasarjana.unisnu.ac.id/wp-content/uploads/2014/05/02.-Cost-Benefit-Analysys-in-Malaysian-Education. Pdf.
- Khan, R. E. A., & Khan, T. (2012). Illegal Labor Markets: A Case Study of Trafficking of Women and Children in South Asia. *Journal of Alternative Perspectives in the Social Sciences*, 4 (1), 139-166.
- Kijima, Y., Otsuka, K., & Sserunkuuma, D. (2011). An Inquiry into Constraints on a Green Revolution in Sub-Saharan Africa: The Case of NERICA Rice in Uganda. *World Development*, 39 (1), 77-86.
- Kim, C. Y. (2011). Child Labor, Education Policy and Governance in Cambodia. International Journal of Educational Development, 31 (5), 496-504.
- Kis-Katos, K. & Schulze, G. (2006). Where Child Labor Supply Finds Its demand? University of Freiburg: Institute for Economic Research, Working Paper.
- Kis-Katos, K. (2007). Does Globalization Reduce Child Labor? The Journal of International Trade & Economic Development, 16 (1), 71-92.
- Kis-Katos, K., & Beegle, R. (2011). Child Labor and Trade Liberalization in Indonesia. *Journal of Human Resources*, 46 (4), 722-749.
- Knox, P., Agnew, J., & McCarthy, L. (2014). The Geography of the World Economy. London: Routledge.
- Konaç, E. H. (2001). Trade and Labour Standards and the OIC Countries. *Journal of Economic Cooperation*, 22 (3), 79-100.
- Krishna, P., Mitra, D., & Sundaram, A. (2010). Trade, Poverty and the Lagging Regions of South Asia: National Bureau of Economic Research. Retrieved from. http://www.researchgate.net/publication/46467296_Trade_Poverty_and_the_Lagging_Regions_of_Sout h_Asia/file/e0b49525d1d4d312a9.pdf.
- Krueger, A. B. (1996). Observations on International Labor Standards and Trade., in Proceedings of the Annual World Batik (Conference on Development Economics, 1996), Washington DC, World Bank.
- Krugman, P. (1981). Trade, Accumulation, and Uneven Development. Journal of Development Economics, 8(2), 149-161.
- Krugman, P. (1991). Increasing Returns and Economic Geography. *Journal of Political Economy*, 99, 483-499.

- Krugman, P. (1992). Does the New Trade Theory Require a New Trade Policy? *The World Economy*, 15 (4), 423-442.
- Krugman, P. (1993). First Nature, Second Nature, and Metropolitan Location. *Journal of Regional Science*, 33 (2), 129-144.
- Krugman, P. R. (1979). Increasing Returns, Monopolistic Competition, and International Trade. *Journal of International Economics*, 9 (4), 469-479.
- Krugman, P., & Venables, A. J. (1996). Integration, Specialization, and Adjustment. European Economic Review, 40 (3), 959-967.
- Krugman, Paul R. (1980), Scale Economies, Product Differentiation, and the Pattern of Trade. *American Economic Review*, 70, 950-959.
- Kucera, D. (2002). Core Labour Standards and Foreign Direct Investment. International Labour Review, 141(1-2), 31-69.
- Kucera, D., & Roncolato, L. (2011). Trade Liberalization, Employment and Inequality in India and South Africa. *International Labor Review*, 150 (1-2), 1-41.
- Kumar, N. (2014). Toward an Asian Economic Community? In the Political Economy of Asian Regionalism, 169-184.
- Kumar, S. (2013). Emerging Trends and Patterns of Trade: An Economic Analysis of SAARC. Journal of International Economics (0976-0792), 4 (1). Retrieved from. http://eds.b.ebscohost.com/ehost/pdfviewer/pdfviewer?sid=eac88fda-8894-4461-b7ac-49320f0b968c%40sessionmgr113&vid=0&hid=110.
- Kurokawa, Y. (2011). Is a Skill Intensity Reversal a Mere Theoretical Curiosum? Evidence from the US and Mexico. *Economics letters*, *112*(2), 151-154.
- Kurokawa, Y. (2011). Variety-Skill Complementarity: a Simple Resolution of the Trade-Wage Inequality Anomaly. *Economic theory*, 46(2), 297-325.
- Laksmana, E. A. (2011). Indonesia's Rising Regional and Global Profile: Does Size Really Matter? Contemporary Southeast Asia: *A Journal of International and Strategic Affairs*, 33 (2), 157-182.
- Lancaster, K. (1980). Intra-Industry Trade under Perfect Monopolistic Competition. *Journal of International Economics*, 10 (2), 151-175.
- Lee-Wright, P. (2013). Child Slaves. London: Earthscan.
- Leung, S., Vo, T. T. and Viseth, K. R (2005) Integration and Transition Vietnam, Cambodia and Lao PDR, International and Development Economics Working Paper 05-1, Canberra: Asia Pacific School of Economics and Government, Australian National University.

- Li, C. M., & Wang, R. F. (2011). The Status Quo and Development Strategy of Muay Thai in China. Advanced Materials Research, 159, 348-352.
- Lieten, G. K. (2011). Introduction: The Worst Forms of Child Labour in Latin America. *In Hazardous Child Labour in Latin America*, 1-19. Springer Netherlands.
- Light, R. J. (1973). Abused and Neglected Children in America: A Study of Alternative Policies. *Harvard Educational Review*, 43(4), 556-598.
- Lin, J. Y. (2011). New Structural Economics: A Framework For Rethinking Development. *The World Bank Research Observer*, 26 (2), 193-221.
- Lopez-Calva, L. F. (2002). A Social Stigma Model of Child Labor. *Estudios Economicos*, 17(2), 193-217.
- Lucassen, J. (2000). In Search of Work. *IISG Research Paper*. Retrieved from http://socialhistory.org/sites/default/files/docs/publications/respap46_0.pdf.
- Luong, Q. V. (2011). *How Can Child Labor Lead to an Increase in Human Capital of Child Laborers and What Are Policy Implications?* (Doctoral dissertation, UNIVERSITY OF CALIFORNIA, BERKELEY). Retrieved from http://escholarship.org/uc/item/07m2v2db#page-42.
- Maeve, G. (2011). *Child Labour in Cambodia*. Retrieved from http://humanrights.ie/children-and-the-law/child-labour-in-cambodia.
- Mahmood, Z. (1998). WTO and Pakistan: Opportunities and Policy Challenges. *The Pakistan Development Review*, 37 (4), 687-701.
- Mahmood, Z., & Azhar, M. (2001). On Over Invoicing of Exports in Pakistan. *The Pakistan Development Review*, 40 (3), 173-185.
- Mansoor, F. (2004). The WTO Versus the ILO and the Case of Child Labour. *Web Journal of Current Legal Issues Ltd, 2.* Retrieved from http://www.peacepalacelibrary.nl/ebooks/files/WEBJCLI MANSOOR WTO-versus-the-ILO.pdf.
- Marchand, B. U., Rees, R., & Riezman, R. (2013). The Effect of Parental Labor Supply on Child Schooling: Evidence From Trade Liberalization in India. *Review of Economics of the Household*, 11 (2), 151-173.
- Markusen, J. R., & Venables, A. J. (1998). Multinational Firms and the New Trade Theory. *Journal of International Economics*, 46 (2), 183-203.
- Markusen, J. R., J. R. Melvin, W. M. Kaempfer, and K. Maskus (1995), *International Trade: Theory and Evidence*. New York: McGraw-Hill.
- Marx, K ([1867] 1981). Capital (Das Kapital). New York, NY: Vintage Books.

- Maskus, K. (1997). Should Core Labor Standards be Imposed through International Trade Policy (World Bank Development Research Group Policy Research Working Paper no. 1817), World Bank, Washington, D.C. Retrieved from http://www.worldbank.org/research/trade/wp1817.html.
- McCarthy, J. J. (Ed.). (2001). Climate Change 2001: Impacts, Adaptation, and Vulnerability: Contribution of Working Group II to the Third Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, Cambridge University Press.
- McClarty, L. M., Bhattacharjee, P., Blanchard, J. F., Lorway, R. R., Ramanaik, S., Mishra, S., & Becker, M. L. (2014). Circumstances, Experiences and Processes Surrounding Women's Entry Into Sex Work in India. *Culture, Health & Sexuality*, 16(2), 149-163.
- McElduff Jr, T. P., & Veiga, J. (2012). The Child Labor Deterrence Act of 1995: A Choice Between Hegemony and Hypocrisy. *Journal of Civil Rights and Economic Development*, 11 (2), 13.
- Mehdi, S. S. (2010). Illegal Migration, Human Smuggling and Trafficking: From Bangladesh to Pakistan and Beyond. Retrieved from http://www.befare.org/2nd%20Policy%20Review.pdf.
- Melgar-Quinonez, H. R., Zubieta, A. C., MkNelly, B., Nteziyaremye, A., Gerardo, M. F. D., & Dunford, C. (2006). Household Food Insecurity and Food Expenditure in Bolivia, Burkina Faso, and the Philippines. *The Journal of Nutrition*, 136 (5), 1431-1437.
- Meschi, E., Taymaz, E., & Vivarelli, M. (2011). Trade, Technology and Skills: Evidence From Turkish microdata. *Labour Economics*, 18, 60-70.
- Miettinen, J., Shi, C., & Liew, S. C. (2011). Deforestation Rates In Insular Southeast Asia between 2000 and 2010. Global Change Biology, 17 (7), 2261-2270.
- Miles, R. E., Snow, C. C., Meyer, A. D., & Coleman, H. J. (1978). Organizational Strategy, Structure, and Process. *Academy of Management Review*, 3 (3), 546-562.
- Moav, O. (2005). Cheap Children and the Persistence of Poverty. The Economic Journal, 115 (500), 88-110.
- Mohapatra, S., & Dash, M. (2011). "Child Labour"-A Product of Socio-Economic Problem for India, Findings and Preventives-A Case of Bhubabaneswar (A State Capital of India). *Educational Research* (ISSN: 2141-5161), 2(6), 1199-1209.
- Moran, T. H. (2001). Parental Supervision: The New Paradigm for Foreign Direct Investment and Development (Vol. 64). Washington, D.C: Peterson Institute.
- Mosley, W. H., & Chen, L. C. (1984). An Analytical Framework for the Study of Child Survival in Developing Countries. Population and Development Review, 10, 25-45.
- Mukhopadhyay, K., & Chakraborty, D. (2005). Is Liberalization of Trade Good for the Environment? Evidence from India. *Asia Pacific Development Journal*, 12 (1), 109.

- Murakami, S. (2013). Social Capital in Vietnam, Cambodia and Laos. *The Senshu Social Capital Review: Exploring Social Capital Towards Sustainable Development in East Asia*, 4, 65-110.
- Mustafa, U., Malik, W., Sharif, M., & Ahmad, S. (2001). Globalisation and Its Implications for Agriculture, Food Security, and Poverty in Pakistan. *The Pakistan Development Review*, 40 (4), 767-786.
- Nalla, M. K., & Mamayek, C. (2013). Democratic Policing, Police Accountability, and Citizen Oversight in Asia: An Exploratory Study. Police Practice and Research, 14 (2), 117-129.
- Nardinelli, C. (1980). Child labor and the Factory Act. Journal of Economic History, 40 (4), 739-55.
- Nardinelli, C. (1990). Child Labor and the Industrial Revolution, Bloomington: Indiana University Press.
- Naseem, I., Rashid, K., & Zaman, S. (2012). South Asia in the Globalized World. Wulfenia 19 (11). Available at SSRN: http://ssrn.com/abstract=2184400.
- Nawaz, M., Nasir, M., Ali, A., & Chaudhry, W. (2011). Trade Liberalization and Child Labor: A Case Study of Pakistan. *Transition Studies Review*, 18 (2), 361-376.
- Nepal, A., & Nepal, M. (2012). Is Child Labour A Substitute for Adult Labour? The Relationship between Child Labour and Adult Illness in Nepal. *International Labour Review*, 151(1-2), 109-121.
- Neumann, K., Verburg, P. H., Stehfest, E., & Müller, C. (2010). The Yield Gap of Global Grain Production: A Spatial Analysis. *Agricultural Systems*, 103 (5), 316-326.
- Neumayer, E., & De Soysa, I. (2005). Trade Openness, Foreign Direct Investment and Child Labor. *World Development*, 33 (1), 43-63.
- Neumayer, E., & Soysa, I. d. (2006). Globalization and the Right to Free Association and Collective Bargaining: An Empirical Analysis. *World Development*, 34 (1), 31-49.
- Nguyen, T. V., & Pham, L. T. (2011). Scientific Output and Its Relationship to Knowledge Economy: An analysis of ASEAN countries. *Scientometrics*, 89 (1), 107-117.
- Nieboer, H. J. (2011). *Slavery as an Industrial System: Ethnological Researches*. Cambridge: Cambridge University Press.
- Nieuwenhuys, O. (1996). The Paradox of Child Labor and Anthropology. *Annual Review of Anthropology*, 25, 237-251.
- Noland, M. (2013). Comment on "ASEAN's New Frontiers: Integrating the Newest Members into the ASEAN Economic Community". *Asian Economic Policy Review*, 8 (1), 42-43.
- Notten, G., de Neubourg, C., Makosso, B., Mpoue, A. B. (2012). *A Multidimensional Profile of Child Poverty in Congo Brazzaville*, in A. Minujín and S. Nandy (eds.), Global Child Poverty and Well-being – Measurement, Concepts, Policy and Action. The Policy Press.

- Ofreneo, R. (2012). Philippine Garments in the Post-MFA Era: Seeking Protection via Free Trade? Institutions and Economies. *International Journal of Institutions and Economies*, 4 (3), 83-102.
- Oh, J., & Rahman Sardar, R. (2013). Gravity Matters: International Trade of Bangladesh. Review of Urban & Regional Development Studies, 25 (1), 34-46.
- Oke, A., Walumbwa, F., Yan, T., Idiagbon-Oke, M., & Ojode, L. A. (2014). Linking economic Status with Technology Adoption in Three Emerging Economies of Sub-Saharan Africa. *Journal of Manufacturing Technology Management*, 25 (1), 49-68.
- Otañez, M., & Glantz, S. A. (2011). Social Responsibility in Tobacco Production? Tobacco Companies' Use of Green Supply Chains to Obscure the Real Costs of Tobacco Farming. *Tobacco Control*, 20 (6), 403-411.
- Park, D., & Shin, K. (2012). The Service Sector in Asia: Is It an Engine of Growth? Working Paper, 12-21, Peterson Institute for International Economics.
- Pasteels, J. M. (2013). "Review of best practice methodologies for imputing and harmonising data in crosscountry datasets." Available at: natlex.ilo.ch/wcmsp5/groups/public/---dgreports/--stat/documents/genericdocument/wcms_389375.pdf.
- Paz, L. S. (2014). Trade Liberalization and the Inter-Industry Wage Premia: The Missing The Role of Productivity. *Applied Economics*, 46 (4), 408-419.
- Péridy, N. (2005). The Trade Effects of the Euro-Mediterranean Partnership: What Are The Lessons for Asean Countries? *Journal of Asian Economics*, 16 (1), 125-139.
- Perrin, B., Pepper, J., Latham, M., Gorstein, J., Bhaskaram, P., Khanum, S., . . . Phungula, M. (2014). Combating Child Labour: Sample Good Practices Guidelines. Annex 4. Food and Nutrition Bulletin, 24 (4), 160-162.
- Phadungkiati, L., & Connell, J. (2014). Social Networks as Livelihood Strategies for Small-scale Traders on the Thai–Lao Border. *Australian Geographer*, 45 (3), 375-391.
- Phillips, N., Bhaskaran, R., Nathan, D., & Upendranadh, C. (2014). The Social Foundations of Global Production Networks: Towards a Global Political Economy of Child Labour. *Third World Quarterly*, 35 (3), 428-446.
- Price, J. I., & Bohara, A. K. (2013). Maternal Health Care Amid Political Unrest: The Effect of Armed Conflict on Antenatal Care Utilization in Nepal. *Health Policy and Planning*, 28 (3), 309-319.
- Qiu, H., Yang, J., Huang, J., & Chen, R. (2007). Impact of China-ASEAN Free Trade Area on China's International Agricultural Trade and Its Regional Development. *China & World Economy*, 15 (5), 77-90.
- Rahim, M. M. (2013) Legal Regulation of Corporate Social Responsibility: A Meta-Regulation Approach of Law for Raising CSR in a Weak Economy. Heidelberg, Springer.

- Rahman, M. M., & Khanam, R. (2012). Child Labour: The Effects of Globalisation. Journal of Applied Business and Economics, 13(4), 59-71.
- Ramasamy, B., & Yeung, M. (2007). Malaysia–Trade Policy Review 2006. The World Economy, 30 (8), 1193-1208.
- Ranjan, P. (2001). Credit Constraints and the Phenomenon of Child Labor. *Journal of Development Economics*, 64 (1), 81-102.
- Ranjan, P. (2003). Trade Induced Convergence through Human Capital Accumulation In a Credit-Constrained Economies. *Journal of Development Economics*, 72 (1), 139-162.
- Ratner, B. D., Åsgård, B., & Allison, E. H. (2014). Fishing for Justice: Human Rights, Development, and Fisheries Sector Reform. *Global Environmental Change*, 27, 120-130.
- Ravallion, M., & Wodon, Q. (2000). Does Child Labour Displace Schooling? Evidence on Behavioural Responses to an Enrollment Subsidy. *The Economic Journal*, 110 (462), 158-175.
- Ray, R. (2000). Analysis of Child Labor in Peru and Pakistan: A Comparative Study. *Journal of Population Economics*, 13 (1), 3-19.
- Ray, R. (2000). Child labor, Child Schooling, and Their Interaction with Adult Labor: Empirical Evidence for Peru and Pakistan. *The World Bank Economic Review*, 14 (2), 347-367.
- Ray, R., & Chatterjee, B. (2012). Trade and Child Labour: A Review of Theoretical Literature. *Journal of International Economics*, 3 (1), 93-113.
- Rees, R., Riezman, R., & Marchand, B. U. (2012). The Effect of Parental Labor Supply on Child Schooling: Evidence from Trade Liberalization in India. Retrieved from http://www.ualberta.ca/~ural/files/RRUM 082012.pdf.
- Reggio, I. (2011). The Influence of the Mother's Power on Her Child's Labor in Mexico. Journal of development economics, 96(1), 95-105.
- Reyes, C., Asis, R. D., & Mondez, M. B. D. (2014). *Child Poverty in the Philippines* (DISCUSSION PAPER SERIES NO. 2014-33). Philippine Institute for Development Studies.
- Ricardo, David. 1817. *On the Principle of Political Economy and Taxation*. London: John Murray, 1821. Volume 1 in P. Sraffa, ed., with the assistance of M. H. Dobb, The Works and Correspondence of David Ricardo. Cambridge: University of Cambridge Press, 1953.
- Rigg, J. (1998). Rural–Urban Interactions, Agriculture and Wealth: A Southeast Asian Perspective. Progress in Human Geography, 22 (4), 497-522.
- Rigg, J., Salamanca, A., & Parnwell, M. (2012). Joining the Dots of Agrarian Change in Asia: A 25 Year View from Thailand. *World Development*, 40 (7), 1469-1481.

- Roberts, B. (2004). A Gravity Study of the Proposed China-ASEAN Free Trade Area. *The International Trade Journal*, 18 (4), 335-353.
- Rodrik D (2011) The Globalization Paradox: Democracy and the Future of the World Economy. W.W. Norton, New York, W. W. Norton & Co.
- Rogers, C. A. & Swinnerton, K. A. (2002). A Theory of Exploitative Child Labor. Manuscript, U.S. Department of Labor.
- Rogers, C. A., & Swinnerton, K. A. (2008). A Theory of Exploitative Child Labor. *Oxford Economic Papers*, 60 (1), 20-41.
- Romer, D. H., & Frankel, J. A. (1999). Does Trade Cause Growth? *American Economic Review*, 89(3), 379-399.
- Rosati, F. C., & Rossi, M. (2003). Children's Working Hours and School Enrollment: Evidence from Pakistan and Nicaragua. *The World Bank Economic Review*, 17 (2), 283-295.
- Rosati, F.C. and M. Rossi (2001), "Children's Working Hours, School Enrolment and Human Capital Accumulation: Evidence from Pakistan and Nicaragua", Understanding Children's Work, UNICEF Innocenti Research Centre.
- Rule, J. (1985). Essays in Comparative Labour History. Labour History Review, 50(1), 22-31.
- Sachdev, H. J. (2013). Thai Exporter and US Importer Relations: A Pilot Study Using Email Survey. *Journal* for Global Business Advancement, 6 (3), 202-211.
- Salazar, L. C., & Das, S. B. (2007). Bridging the ASEAN Developmental Divide: Challenges and Prospects. *ASEAN Economic Bulletin*, 24 (1), 1-14.
- Sally, R., & Sen, R. (2011). Trade Policies in Southeast Asia in the Wider Asian Perspective. *The World Economy*, 34 (4), 568-601.
- Samaratunge, R., Alam, Q., & Teicher, J. (2008). The New Public Management Reforms in Asia: A Comparison of South and Southeast Asian Countries. *International Review of Administrative Sciences*, 74(1), 25-46.
- Samimi, P., Lim, G. C., & Buang, A. A. (2011). Globalization Measurement: Notes on Common Globalization Indexes. *Journal of Knowledge Management, Economics & Information Technology*, 1(7), 197-216.
- Sankharat, U. (2013). Cambodian Child Migrant Workers in the Rong Kluea Market Area in Thailand. *Asian Social Science*, 9(11), 24-32.
- Santos, D., Scholar, M., Tickamyer, A., & Scheuble, L. (2012). Child Labor: An Analysis of Its Predictors and Policies. *The Penn State McNair Journal*, 18, 97-122.

- Sapir, A. (1995). The Interaction between Labour Standards and International Trade Policy. *The World Economy*, 18(6), 791-803.
- Sawyer, W. C., Sprinkle, R. L., & Tochkov, K. (2010). Patterns and Determinants of Intra-Industry Trade in Asia. *Journal of Asian Economics*, 21(5), 485-493.
- Schaer, M.E. and Stillman, S., 2010. xtoverid: Stata Module to Calculate Tests of Overidentifying Restrictions after xtreg, xtivreg, xtiv2reg and xthtaylor. Available at: http://ideas.repec.org/c/boc/bocode/s456779.html.
- Schaffer, M. E., & Stillman, S. (2011). XTOVERID: Stata Module to Calculate Tests of Overidentifying Restrictions after xtreg, xtivreg, xtivreg2, xthtaylor. *Statistical Software Components*.
- Schultz, T. P. (2009). How Does Family Planning Promote Development? Evidence from a Social Experiment in Matlab, Bangladesh—1977–1996. Yale University, Economic Growth Center, New Haven, Conn.
- Scott, A. J. (2006). The Changing Global Geography of Low-Technology, Labor-Intensive Industry: Clothing, Footwear, And Furniture. *World Development*, 34(9), 1517-1536.
- Sedara, K., Sophal, T., and Acharya, S. (2002) 'Land, Rural Livelihoods and Food Security in Cambodia', Working Paper 24, Phnom Penh: Cambodia Development Resource Institute.
- Sen, A. (2013). Work and S. International Labour Review, 152(1), 81-92.
- Serra, R. (2009). Child fostering in Africa: When Labor and Schooling Motives May Coexist. *Journal of Development Economics*, 88(1), 157-170.
- Serrano, R., & Pinilla, V. (2014). Changes in the Structure of World Trade in the Agri-Food Industry: The Impact of the Home Market Effect and Regional Liberalization From a Long-Term Perspective, 1963– 2010. Agribusiness, 30(2), 165-183.
- Servais, J.-M. (2011), The International Labour Organization, in J. Wouters (ed.), International Encyclopedia of Laws: Intergovernmental Organizations. Deventer: Kluwer, 3rd ed., 2011.
- Shakya, S. R., Kumar, S., & Shrestha, R. M. (2012). Co-benefits of a Carbon Tax in Nepal. *Mitigation and Adaptation Strategies for Global Change*, 17(1), 77-101.
- Sharma, K. (2007). Trade, Growth and Child Labour Practices in South Asia. in Gamini Herath and Kishor Sharma (eds.) Child Labour in South Asia (Aldershot, Hampshire, UK and Burlington, VT, USA: Ashgate), Chapter 5.
- Sharma, S. C., & Chua, S. Y. (2000). ASEAN: Economic Integration and Intra-Regional Trade. *Applied Economics Letters*, 7(3), 165-169.

- Siah, K. L., Choong, C. K., & Yusop, Z. (2014). AFTA and the Intra-Trade Patterns among ASEAN-5 Economies: Trade-Enhancing or Trade-Inhibiting? International *Journal of Economics and Finance*, 1(1), 117-128.
- Siddiqi, F., & Patrinos, H. (1995). Child Labor: Issues, Causes and Interventions. Human Resources and Operations Policy Working Paper 56, World Bank, Washington, DC.
- Siddiqui, R., Siddiqui, R., & Akhtar, M. R. (1998). A Decomposition of Male-Female Earnings Differentials. *The Pakistan Development Review*, 37, 885-898.
- Sim, A.A., Suryadarma, D., and Suryahadi, D. (2012). The Consequence of Child Market Work on the Growth of Human Capital. Jakarta: SMERU.
- Singh, A., & Zammit, A. (2004). Labour Standards and the 'Race to the Bottom': Rethinking Globalization and Workers' Rights from Developmental and Solidaristic Perspectives. *Oxford Review of Economic Policy*, 20(1), 85-104.
- Singh, P. (2005). Child Labour in India. Combat Law Magazine. Available at: http://works.bepress.com/pankaj_singh/4.
- Smith Adam (1776 [1993]), An Inquiry into the Nature and Causes of the Wealth of Nations; A Selected Edition, Sutherland, Kathryn (ed.), Oxford University Press.
- Smith, A. (1904), An Inquiry into the Nature and Causes of the Wealth of Nations, (1776). Reprint, London: Printed for W. Strahan and T. Cadell.
- Smith, C. A. (1984). Does a Commodity Economy Enrich the Few While Ruining the Masses? Differentiation Among Petty Commodity Producers in Guatemala 1. *The Journal of Peasant Studies*, 11(3), 60-95.
- Sotharith, C. (2011). "Industrial Development Policy and Intermediate Goods Trade in Cambodia," in Kagami, M. (ed) Intermediate Goods Trade in East Asia: Economic Deepenign Through FTAs/EPAs (BRC Research Report No. 5). Bangkok: Bangkok Research Center, IDE-JETRO. *Cambodia* - ASEAN -LATIN BUSINESS FORUM 2012.
- Spence, M. (1976). "Product Differentiation and Welfare." American Economic Review Papers and Proceedings, 66(2), 407-414.
- Startz, R. (1989). Monopolistic Competition as a Foundation for Keynesian Macroeconomic Models. *The Quarterly Journal of Economics*, 104(4), 737-752.
- Stern, D. I., & Cleveland, C. J. (2004). Energy and Economic Growth. Encyclopedia of Energy, 2, 35-51.
- Stigler, G. J. (1957). Perfect Competition, Historically Contemplated. *The Journal of Political Economy*, 65(1), 1-17.
- Sultan, P. (2008). Trade, Industry and Economic Growth in Bangladesh. *Journal of Economic Cooperation*, 29(4), 71-92.

- Swinnerton, K. A., & Rogers, C. A. (1999). The Economics of Child Labor: Comment. American Economic Review, 89, 1382-1385.
- Sychareun, V., Faxelid, E., Thomsen, S., Somphet, V., & Popenoe, R. (2011). Customary Adolescent Sexual Practices Among the Akha of Northern Lao PDR: Considerations for Public Health. Culture, Health & Sexuality, 13(2), 249-262.
- Solow, R. M. (1956). A Contribution to the Theory of Economic Growth. *The Quarterly Journal of Economics*, 70, 65-94.
- Tariq, B. and Ab-Rahim, R. (2016). The Environmental Effects of Intra-industry Trade in the SAARC Region. *International Journal of Business and Society*, 17(1), 113-140.
- Terada, T. (2011). Comment on "Politics of Association of Southeast Asian Nations Economic Cooperation". *Asian Economic Policy Review*, 6(1), 41-42.
- Terwiel, B. J. (2013). Early Interactions between South and Southeast Asia: Reflections on Cross-Cultural Exchange. *Asian Studies Review*, 37(1), 117-118.
- Thapa, S. B. (2012). Nepal's Trade Flows: Evidence from Gravity Model. *NRB Economic Review*, 24(1), 16-27.
- The World Bank. (2014). Data. Retrieved September 1, 2014, from Nepal: Retrieved from http://data.worldbank.org/country/nepal.
- Timmer, C. P. (2014). Food Security in Asia and the Pacific: The Rapidly Changing Role of Rice. *Asia & the Pacific Policy Studies*, 1(1), 73-90.
- Tongzon, J. L. (2002). *The Economies of Southeast Asia: Before and after the Crisis. Cheltenham*, UK: Edward Elgar Publishing.
- Topalova, P. (2010). Factor Immobility and Regional Impacts of Trade Liberalization: Evidence on Poverty from India. *American Economic Journal: Applied Economics*, 2(4), 1-41.

Trebilcock, M., Howse, R., & Eliason, A. (2012). The Regulation of International Trade. London: Routledge.

- Tzannatos, Z. (1999). Women and Labor Market Changes in the Global Economy: Growth Helps, Inequalities Hurt And Public Policy Matters. *World Development*, 27(3), 551-569.
- Udry, C., 2006. *Child Labor*, in Banerjee, A.V., Benabou, R., and Mookherjee, D. (Eds.), Understanding Poverty. New York, Oxford University Press, 243–257.
- UNICEF (2013) Definition of Child Labour. http://www.childinfo.org/ labour_1053.htm. Accessed October 2013.
- UNESCAP. (2010) Economic and Social Survey, 2010, UNESCAP, Bangkok.

- UNESCO (2013) Lists of intangible Cultural Heritage and Register of Best Safeguarding Practices. United Nations Educational, Scientific and Cultural Organization. http://www.unesco.org/culture/ich/index.php?lg=en&pg=00011&multinational=3&display1= inscriptionID&display=stats#tabs. Accessed October 2013.
- UNICEF (1986) The State of the World's Children 1986. New York: United Nations.
- UNICEF (2014). State of the World's Children. United Nations Children's Fund (UNICEF).
- UNICEF. (2005). The State of the World's Children 2006: Excluded and Invisible. Unicef.
- UNICEF. (2006). *The State of the World's Children 2006*. Available at http://www.unicef.org/sowc06/pdfs/sowc06_Table9.pdf.
- UNICEF. (2006). The State of the World's Children 2007: Women and Children: The Double Dividend of Gender Equality (Vol. 7). UNICEF.
- UNICEF. (2006). The State of the World's Children 2007: Women and Children: The Double Dividend of Gender Equality (Vol. 7). Unicef.
- UNICEF. (2007). *The State of the World's Children 2007*. Available at http://www.unicef.org/sowc07/docs/sowc07_Table_9.pdf.
- UNICEF. (2008). *The State of the World's Children 2008*. Available at http://www.unicef.org/sowc08/docs/sowc08 Table 9.pdf.
- UNICEF. (2009). *The State of the World's Children 2009*. Available at http://www.unicef.org/sowc09/docs/SOWC09_Table_9.pdf.
- UNICEF. (2011). The State of the World's Children 2011. Available at http://www.unicef.org/adolescence/files/SOWC_2011_Main_Report_EN_02092011.pdf.
- Unicef. (2011). The State of the World's Children 2011-Executive Summary: Adolescence an Age of Opportunity. UNICEF.
- UNICEF. (2012). Monitoring the Situation of Children and Women. Nepal Multiple Indicator Cluster Survey 2010. Mid-and Far Western Regions. Final Report 2012. *Kathmandu: UNICEF*.
- UNICEF. (2012). *The State of the World's Children 2012* Available at http://www.unicef.org/sowc2012/statistics.php.
- UNICEF. (2013). *The State of the World's Children 2013* Available at http://www.unicef.org/sowc2013/files/Table_9_Stat_Tables_SWCR2013_ENGLISH.pdf.
- UNICEF. (2014). *The state of the world's children 2014* available at http://www.unicef.org/sowc2014/numbers/documents/english/EN-FINAL%20Table%209.pdf.

- US Department of Labor (2015). Philippines 2014 Findings on the Worst Forms of Child Labor Significant Advancement. Available from: http://www.dol.gov/ilab/reports/child-labor/philippines.htm [assessed 17 November 2015].
- US Department of Labor (2015). Sri Lanka 2014 Findings on the Worst Forms of Child Labor Moderate Advancement. Available from: http://www.dol.gov/ilab/reports/child-labor/sri_lanka.htm [assessed 17 November 2015].
- Vallance, S. (1999). Performance Appraisal in Singapore, Thailand and the Philippines: A Cultural Perspective. *Australian Journal of Public Administration*, 58(4), 78-95.
- Van Binsbergen, J. H., Fernandez-Villaverde, J., Koijen, R. S., & Rubio-Ramirez, J. (2012). The Term Structure of Interest Rates in a DSGE Model with Recursive Preferences. *Journal of Monetary Economics*, 59, 634–648.
- van de Glind, H. (2010). Child Trafficking: A Modern Form of Slavery1. Child Slavery Now: A Contemporary Reader, Portland, Ore. : Policy Press.
- Vanek, J. (1968). The Factor Proportions Theory: The N-Factor Case. Kyklos, 21(4), 749-756.
- Veeramani, C. (1998) 'Intra-Industry Trade under Economic Liberalisation: The Case of Indian Capital Goods Industries', *Journal of Indian School of Political Economy*, 11, 455-73.
- Ventevogel, P., Jordans, M. J., Eggerman, M., van Mierlo, B., & Panter-Brick, C. (2013). Child Mental Health, Psychosocial Well-Being and Resilience in Afghanistan: A Review and Future Directions Handbook of Resilience in Children of War (pp. 51-79): Springer.
- Vona, S. (2013). Intra-Industry Trade: A Statistical Artefact or a Real Phenomenon? *PSL Quarterly Review*, 43(175), 383-412.
- Wade, R. H. (2003). What Strategies are Viable for Developing Countries Today? The World Trade Organization and the Shrinking of 'Development Space'. *Review of international political economy*, 10(4), 621-644.
- Wahid, F., & Furuholt, B. R. (2012). Understanding the Use of Mobile Phones in the Agricultural Sector in Rural Indonesia: Using the Capability Approach as Lens. *International Journal of Information and Communication Technology*, 4(2), 165-178.
- WB (2013) Improved Sanitation Facilities (% of Population with Access) The World Bank. http://data.worldbank.org/indicator/SH.STA.ACSN/countries?display=map. Accessed October 2013.
- WB (2014) Unemployment, Total (% of total labor force) (modeled ILO estimate) http://data.worldbank.org/indicator/SL.UEM.TOTL.ZS. Accessed October 11 2014.
- WDI (2012). World Development Indicators, World Bank: Washington.
- WDI (2014). World Development Indicators, World Bank: Washington.

- WDI (2014). *World Development Indicators*. Washington, The World Bank, Available at: http:// databank.worldbank.org.
- Webbink, E., Smits, J., & de Jong, E. (2012). Hidden Child Labor: Determinants of Housework and Family Business Work of Children in 16 Developing Countries. *World Development*, 40(3), 631-642.
- Weeraratne, B. (2005). Labour Standards and International Trade: The Case of EU GSP Concessions to Sri Lanka: Institute of Policy Studies. Working Paper No. 8.
- Weiner, M. (1991). The Child and the State in India: Child Labor and Education Policy in Comparative Perspective. Princeton: Princeton University Press.
- Willis, R. J. (1973). A New Approach to the Economic Theory of Fertility Behavior. *The Journal of Political Economy*, 81(2), 14-64.
- Wood, A. (1995). How Trade Hurt Unskilled Workers. The Journal of Economic Perspectives, 9(3), 57-80.

World Bank (2008) World Development Indicators 2008 CD ROM, World Bank, Washington DC.

- World Bank (2009a) Migration and Remittance Data, World Bank, Washington DC, Available at http://econ.worldbank.org/wbsite/external/extdec/extdecprospects/0,,conten tMDK:21352016~pagePK:64165401~piPK:64165026~theSitePK:476883,00.html (accessed on 29 December 2009).
- World Bank (2009b) *WDI Online Word Development Indicators*, World Bank, Washington DC, available at http://ddptxt.worldbank.org/ext/DDPQQ/member.do?method=getMembers&userid=1&queryI d=6 (accessed on 11 October 2010).
- World Bank (2012). World Development Indicators 2012. [Data file] Available at http://data.worldbank.org.
- Yusoff, M. B., & Febrina, I. (2014). Trade Openness, Real Exchange Rate, Gross Domestic Investment and Growth in Indonesia. Margin: *The Journal of Applied Economic Research*, 8(1), 1-13.
- Zaman, K., Shah, I. A., Khan, M. M., & Ahmad, M. (2013). on External Debt: New Evidence from a Panel of SAARC Countries. *Journal of Economic and Social Studies*. 3(2), 131-149.
APPENDIX A

A.1 Monopolistic Competition and the Child Labor

A.1.1 Production

The profit function is the firm's revenue less labor cost and child labor penalties such that:

$$\pi_{i=} p_i \left(1 - \mathring{A}_i\right) x_i - w\alpha - (w\beta + t\varsigma l_i) x_i \qquad \dots (A-1)$$

Where in monopistic competition P = AR = ATC = LRAC

$$p = p_i \left(1 - \mathring{A}_i \right) x_i \qquad \dots (A-2)$$

By putting the values of p_i and ςl_i , in equation 11 (see, Chapter 4), the following profit function is obtained:

$$\pi_i = p_i \left((1 - \mathring{A}_i) x_i \right) \left((1 - \mathring{A}_i) x_i \right) - w\alpha - w\beta x_i - t \left(1 - \mathring{A}_i \right)^{\sigma} x_i \qquad \dots \text{ (A-3)}$$

A.1.2 First Order Conditions

First order conditions for profit maximization with respect to *x* implies:

$$p'(1-\text{\AA})^{2}x + p(1-\text{\AA}) - w\beta - t(1-\text{\AA})^{\sigma} = 0 \dots (A-4)$$

Divide equation by (1 - Å) and rearrange to obtain:

$$p'(1-\text{\AA})x + p - t(1-\text{\AA})^{\sigma-1} - w\beta(1-\text{\AA})^{-1}$$
 ... (A-5)

The first order conditions for profit maximization with respect to Å is:

$$-xp'(1-Å)x - xp + \sigma t(1-Å)^{\sigma-1}x = 0 \qquad \dots (A-6)$$

Divide the equation (A-6) by -x and rearrange, equation (A-6) can be written as:

$$p'(1-A)x + p - \sigma t(1-A)^{\sigma-1} = 0$$
 ... (A-7)

A.1.3 Second Order Conditions

The second order condition for profit maximization with respect to quantity of output is:

$$\frac{\partial^2 \pi}{\partial x^2} = p''(1 - \text{\AA})x(1 - \text{\AA})x + p'(1 - \text{\AA})^2 + p'(1 - \text{\AA}) \dots (A-8)$$

Since $p' = \frac{\partial p}{\partial x} < 0$ and assuming $p'' = \frac{\partial^2 p}{\partial x^2} < 0$, then $\frac{\partial^2 \pi}{\partial x^2} < 0$ and satisfy the second order condition for a maximization. $\frac{\partial p}{\partial x}$, $\frac{\partial^2 p}{\partial x^2}$ and $\frac{\partial^2 \pi}{\partial x^2}$ shows that increase in the production has an

inverse effect on prices and profits of the firms, so there is need to check these effects by using Å which is fraction of output allocated towards labor amelioration.

The second order condition for profit maximization with respect to resources allocated for labor amelioration is:

$$\frac{\partial^2 \pi}{\partial \mathring{A}^2} = -xp''(-x)\left(1 - \mathring{A}\right)x + xp'(1 - \mathring{A})x(-1)x + p'(-x) + (\sigma - 1)\sigma t \left(1 - \mathring{A}\right)^{\sigma - 2}(-1)\dots(A-9)$$

Since $p' = \frac{\partial p}{\partial A} > 0$ and assuming $p'' = \frac{\partial^2 p}{\partial A^2} < 0$, then $\frac{\partial^2 \pi}{\partial A^2} > 0$ and satisfy the second order condition for a maximization.

 $\frac{\partial p}{\partial \dot{A}} > 0$, $\frac{\partial^2 p}{\partial \dot{A}^2} < 0$ and $\frac{\partial^2 \pi}{\partial \dot{A}^2} > 0$, shows that, firstly the fraction of output allocated towards a labor amelioration, increase the product price but after that due to an amelioration in the labor less fraction of output allocated towards labor amelioration, hence it decreases the product price prices and lesser resources allocated towards the labor amelioration.

A.2 The Labor Amelioration

Rearrange, equation (A-5) and equation (A-7) to obtain:

$$\sigma t (1 - \text{\AA})^{\sigma - 1} - t (1 - \text{\AA})^{\sigma - 1} = w\beta (1 - \text{\AA})^{-1} \qquad \dots \text{ (A-10)}$$

Dividing through by $(1 - \text{Å})^{-1}$, collect terms and rearrange to obtain:

$$t(1 - \mathring{A})^{\sigma}(\sigma - 1) = w\beta \qquad \dots \text{ (A-11)}$$
$$(1 - \mathring{A})^{\sigma} = \left(\frac{w\beta}{t(\sigma - 1)}\right)$$
$$(1 - \mathring{A}) = \left(\frac{w\beta}{t(\sigma - 1)}\right)^{\frac{1}{\sigma}}$$
$$\mathring{A} = 1 - \left(\frac{w\beta}{t(\sigma - 1)}\right)^{\frac{1}{\sigma}} \dots \text{ (A-12)}$$

When firms hire the labor at high cost, then, skilled, trained and educated workers are preferred then no need of training (amelioration), on the other hand, if firms take the labor at very low cost then there is a need of amelioration for the workers (Freeman, 2000). Thus, the fraction of output allocated towards labor amelioration.

Then, substituting into equation 9 (see, Chapter 4), the equation for child labor intensity can be written as:

$$cl = (1 - \text{\AA})^{\sigma} \dots (\text{A-13})$$

$$(1 - \text{\AA}) = \left(\frac{w\beta}{t(\sigma-1)}\right)^{\frac{1}{\sigma}}$$
 see equation (A-11)

$$cl = \left(\frac{w\beta}{t(\sigma-1)}\right)^{\frac{1}{\sigma}} \dots (A-14)$$

Equation (A-14) shows that wage (w) and co-efficient of wage are directly related to child labor, but child labor penalties and response in child labor due to the fraction of output allocated towards the consumption are inversely proportion to child labor.

A.3 Table A.1: Relationship of Child Labor and Amelioration with Different Variables				
Variables	Amelioration (Å)	Child Labor (cl)		
	w β -ve	+ve		
	$egin{array}{ccc} t & & \ & + \mathrm{ve} \ & & & \end{array}$	-ve		

Note: -ve indicate the negative relationship with child labor and amelioration. +ve indicate the positive relationship with child labor and amelioration.

A.4 Effects of Child Labor Policy

Changes in the child labor policy can have implication effects on the consumer welfare as well as economic preference. If a country wants a better quality care for vulnerable children, for example, in the form of no child labor is desirable, then, stringent child labor regulation can be imposed to attain it. However, a strict child labor policy affects the firm's production decisions in the consideration of rising compliance costs. Hence, in the developing economy, there exists a trade-off between the achievements of child labor policies and economic objectives. This section of study analyzes the effects of an increase the robustness of child labor policy. More exclusively, the comparative effects of an increase in child labor rate are examined for the following variables: the fraction of output allocated towards amelioration, and child labor intensity.

A.5 Amelioration

Theoretically, an increase in child labor penalty is expected to increase the fraction of output used for the amelioration of labor. Thus, if child labor penalty increase then the marginal benefit of reducing child labor increases.

From equation (A-12), the effect of a change in the child labor penalty to the level of amelioration can be shown as follows:

$$\mathring{A} = 1 - \left(\frac{w\beta}{t(\sigma-1)}\right)^{\frac{1}{\sigma}}$$
$$\frac{\partial\mathring{A}}{\partial t} = \frac{(t^{-2})}{\sigma} \left(\frac{w\beta}{t(\sigma-1)}\right)^{\left(\frac{1}{\sigma}-1\right)} \left(\frac{w\beta}{\sigma-1}\right) > 0$$

The result shows that the direction of the relationship between a positive change in the child labor penalty and fraction of output allocated for amelioration is positive. Hence, an increase in child labor penalty increases the fraction of output allocated for the amelioration of labor. In other words, stricter child labor regulation acts as an incentive for the firm to assign a greater amount of resources for the amelioration of labor from the firm.

The increase in child labor penalty necessitates the firm to increase the allocation of output for labor amelioration and reduce the child labor penalty payments.

A.6 Child Labor Intensity

From equation (A-14), the effect of a change in child labor penalty on child labor intensity is given by the following:

$$cl = \left(\frac{w\beta}{t(\sigma-1)}\right)$$
$$\frac{\partial cl}{\partial t} = -t^{-2}\left(\frac{w\beta}{\sigma-1}\right) < 0$$

An increase in the child labor penalty persuades the firms to allocate more resources towards reducing child labor. This result is consistent with the theoretical expectation that a stricter child labor policy leads to the firm's decision to increase its labor amelioration. Further, since the level of amelioration is increasing, that increases the wage rate or income which leads to a decrease in child labor intensity.

A.7 Policy Induced Child Effects

Solve for *n* firms (see equation 6 chapter 4)

$$\sum_{i=1}^{n} \varsigma l_i = \sum_{i=1}^{n} N x_i \dots (A-15)$$

Putting the value of *N* (see equation 8, Chapter 4)

$$\sum_{i=1}^{n} \varsigma l_i = \sum_{i=1}^{n} x_i (1 - \text{\AA}) q_i^{-1} \cdot x_i$$

Let $x_i (1 - \text{\AA}) q_i^{-1} = \text{\AA}_i$
$$\sum_{i=1}^{n} \varsigma l_i = \sum_{i=1}^{n} \text{\AA}_i \cdot x_i$$

 $n\varsigma l = n \cdot \text{\AA} \cdot x \dots$ (A-16)

Let $n\varsigma l = CL$

$$CL = n \cdot \text{Å} \cdot x \dots (A-17)$$

Take a logarithmic transformation of the equation equation (A-17) and write as:

$$\log CL = \log n + \log \text{\AA} + \log x \dots (A-18)$$

Totally differentiate:

$$\frac{1}{CL}dCL = \frac{1}{n}dn + \frac{1}{CL}dCL + \frac{1}{x}dx \text{ or } \frac{dCL}{CL} = \frac{dn}{n} + \frac{d\mathring{A}}{\mathring{A}} + \frac{dx}{x} \dots (A-19)$$

Then, take the percent change (multiply through by 100%) to obtain:

$$\widehat{CL} = \widehat{n} + \widehat{A} + \widehat{x} \dots (A-20)$$

Thus, equation (A-18) shows that the impact of regulation on child labor can be disintegrated into the selection, technique and scale effects in the following way:

Selection effect	=	ñ
Technique effect	=	Â
Scale effect	=	â

Hence, the impact of the production of differentiated goods generates three types of effects, namely the selection, scale and technique effects.

Equation	Description
$\frac{\partial \mathring{A}}{\partial t} = \frac{(t^{-2})}{\sigma} \left(\frac{w\beta}{t(\sigma-1)} \right)^{\left(\frac{1}{\sigma}-1\right)} \left(\frac{w\beta}{\sigma-1} \right) > 0$	An increase in child labor penalty increases the labor amelioration by firms.
$\frac{\partial cl}{\partial t} = -t^{-2} \left(\frac{w\beta}{\sigma - 1} \right) < 0$	An increase in child labor penalty decreases the child labor intensity. This is policy induced technique effect.
$p' = \frac{\partial p}{\partial \dot{\Lambda}} > 0 \text{ and } p'' = \frac{\partial^2 p}{\partial \dot{\Lambda}^2} < 0$	An increase in child labor penalty increase the labor amelioration it raises the price level initially, then fall in the price level.
$p' = \frac{\partial p}{\partial x} < 0 \text{ and } p'' = \frac{\partial^2 p}{\partial x^2} < 0$ then $\frac{\partial^2 \pi}{\partial x^2} < 0$	An increase in production increases the supply of goods and services. It decreases the price of and profit margins.
$\widehat{CL} = \widehat{n} + \widehat{A} + \widehat{x}.$	An increase in child labor penalty may increase or decrease the total level child labor in economy depending on the sum of policy induced scale, technique and selection effects.

A.8 Table A.2: Policy Induced Child Labor Effects

In autarky case, this model shows that the effect of child policy can be decomposed into policy induced selection, scale and technique effects. Further, this model shows that stricter child labor policy, i.e. The child labor penalty increases the labor amelioration and decrease the child labor. The aggregate impact of change in child labor policy is sum of policy induced selection, scale and technique effects.

APPENDIX B

B.1 Imputations Methods

The nature of the missing data points, leads this study, towards three complementary approaches, as described by Pasteels (2013).

In the first case, time series has short gaps between existing data points (gaps are not more than three successive years). The missing periods are interpolated using a logarithmic growth function.

In the second case, an alternative (second best available) source of data for a given country is utilized for the missing periods.

$$\hat{a}_{(t+1)} = a_{(t)} * \frac{b_{(t+1)}}{b_{(t)}}$$

This information is employed to fill missing data points in the preferred time series (first best variable a), and the growth rate for second best source (variable b) is used. Note that this imputation approach is used for only ASEAN countries.